

Investigating the benefits of Participatory Scenario Planning for tackling social-ecological problems

PhD International Development

School of Agriculture, Policy and Development

Samuel Poskitt

November 2017

Declaration

Declaration: I confirm that this is my own work and the use of all material from other sources has been properly and fully acknowledged.

Signed:

Samuel Poskitt

Abstract

Participatory scenario planning (PSP) is a method in which diverse groups of participants imagine alternative narratives of plausible future events, conditions and trajectories. Researchers and practitioners commonly regard PSP as a useful method for tackling social-ecological problems (SEPs) by incorporating the knowledges of different stakeholders in dialogue to help build a holistic understanding of them. However, within PSP practice in the field of social-ecological resilience the literature is imprecise on the benefits it may have for those whose knowledge it incorporates. Seminal critiques of participation indicate that the outcomes of participatory processes typically reflect the interests of those who initiate them. This doctoral research therefore investigated the benefits PSP may have for those whose knowledge it incorporates. My analysis focused on learning, given the emphasis in the literature on using PSP to facilitate dialogue between participants with different knowledges.

I collected data from: a critical review of 23 cases of PSP described in the peer-reviewed literature, semi-structured interviews with PSP practitioners, and two case studies of PSP processes. I analysed this data to explore: 1) the expected and reported benefits of PSP; 2) how and under what conditions learning occurs; and 3) how and why learning varies. To understand learning in PSP, I developed an innovative conceptual framework, built on a seminal learning theory, the 'Zone of Proximal Development'. This postulates that learning occurs through interactions between different participants, which enable them to push beyond their usual range of thinking.

My analysis indicates that learning is the most commonly expected and reported benefit associated with PSP. I reason that PSP stimulates learning by prompting focused discussions, which expose participants to different assumptions about the future. PSP thus encourages creative, focused thinking, that can help participants to push beyond their usual range of thinking. I found that learning varies according to participants' relative prior expertise in specific topics, social and economic backgrounds, previous experience (if any) of PSP, and the design of specific PSP processes. I reason that learning varies because of the relative extent to which different participants consider the knowledge encountered in PSP to be relevant to their interests, as well as developed through a fair and unbiased process. This perception is strongly influenced both by facilitators and the objectives of wider projects in which specific PSP processes are embedded. I conclude that, although PSP can be a useful method for learning, there is a need for more robust governance to ensure critical reflection and evaluation of the use of PSP in this context.

Acknowledgements

I am very grateful to have had this opportunity to conduct an in-depth study into a topic of such abiding interest. This PhD would have been impossible without the fully-funded, joint scholarship I received from the James Hutton Institute, and the Economic and Social Research Council's South East Doctoral Training Centre. I would like to thank my two supervisors, Dr Andrew Ainslie, and Dr Kerry Waylen, who have been nothing short of models for professionalism, commitment, and academic excellence. Additionally, I am thankful for the advice and support provided by Dr Kirsty Blackstock, Dr Grady Walker and Dr Henny Osbahr. My sincerest thanks go to all the participants, including practitioners, case study participants, and facilitators, who contributed their insights to this research. Special thanks go to the organisers of two participatory scenario planning processes, in Tanzania and South Africa, for welcoming me, and allowing me to study their processes as case studies. Finally, I would like to thank my family and friends for their undiminishing advice and encouragement. I could not have done this without you.

For Toby the dog, whose love of the world inspired me to study it.

Contents

Chapter 1 - Introduction

1.1 Introduction 1	1
1.2 An introduction to scenario planning1	13
1.2.1 Methods for thinking about the future1	13
1.2.2 Origins of scenario planning1	.4
1.2.3 What is scenario planning?1	15
1.2.4 Types of scenario planning1	6
1.3 Scenario Planning for tackling complex problems in SES 2	20
1.4 Research Aim and Questions 2	22
Research Aim 2	22
Research Questions 2	22
1.5 Thesis Outline 2	22

Chapter 2 Literature review and conceptual framework

2.1 Introduction
2.2 PSP, participation and social-ecological resilience27
2.3 Perceived benefits of PSP 29
2.4 How can learning theories contribute to understanding how learning occurs in PSP? 32
2.4.1 What is learning?
2.4.2 Traditions of Learning Theory
Behaviourist Learning Theory
Cognitivist Learning Theory
2.4.3 Boundary Objects
2.4.4 Scaffolding
2.5 What theories can help understand who learns what and why?

. 40
. 41
. 42
. 42
. 44
. 46
. 46
. 47
. 48
. 49
. 49
- - -

Chapter 3 - Research Methodology

3.1 Introduction	51
3.2 Research Approach 5	51
3.3 Research Design 5	52
3.4 Research Methods 5	54
3.4.1 Case Review	54
3.4.2 Practitioner Interviews5	57
3.4.3 Case Studies	59
Food Security Futures	51
Positive Futures for Southern Africa 6	53
Pre-workshop questionnaires6	65
Observation of PSP Workshops6	65
Case Study Participant Interviews	57
3.5 Analysis	59
Reliability and Validity7	71

3.6 Positionality and issues faced	71
3.7 Research Ethics	75
3.8 Conclusion	75

Chapter 4 - What are the expected and reported benefits of participatory scenario planning processes?

4.1 Introduction
4.2 "It helps to break it down, step-by-step." – Structuring future possibilities
4.2.1 Expected Benefits78
4.2.2 Reported benefits 81
4.2.3 Limitations of PSP for structuring future possibilities
4.3 "I learnt a lot because my understanding was completely different." - Learning 83
4.3.1 Expected benefits
4.3.2 Reported Benefits 88
4.3.3 "The thing I really would like is a stronger link to an underlying theory." – Limited
understanding of how learning occurs93
4.4 "You can actually try those adaptations out, without having to wait until that future
unfolds." - Developing Strategies for tackling complex problems
4.4.1 Expected benefits
4.4.2 Reported Benefits 101
4.4.3 "I don't think the effect has lasted" – Limited evidence for the implementation of
responses, developed in PSP105
4.5 "I wasn't able to concretely follow up on it." - Limited provision for evaluation of PSP
4.5.1 Limited consideration of the pros and cons of PSP
4.5.2 Limited consideration of alternative methods 112
4.6 Conclusion 117

Chapter 5 - How and under what conditions does learning occur in participatory scenario planning processes?

5.1 Introduction
5.2 "The learning potential lies in interactions." – Learning occurs Interactions between
different participants 120
5.3 "Giving people a structure to push beyond where their thinking would normally take
them." – PSP prompts discussion
5.4 Scaffolding136
5.5 Conducive conditions for learning in PSP141
5.5.1 "Everyone felt very sharply distinct" - Diversity of participants is conducive to
learning
5.5.2 "We hit a dead stop." – Challenges of including diverse participants
5.5.3 "You need a very good facilitator to come in and make sure that everyone has a
voice." - Importance of facilitators for managing interactions 147
5.5.4 "Scenario does not exist in Swahili." – Some participants are less able/willing to
engage in PSP 150
5.5.5 "I want to be operated on by a surgeon." – The importance of relevant expertise
5.5.6 Space, time and resources156
5.6 Conclusion

Chapter 6 - How, and why, does learning vary in participatory scenario planning processes?

6.1 Introduction	160
6.2 Who learned what?	161
6.2.1 "It was a good exercise for re-organising thoughts that I've had for many yea	ırs.' -
Refining existing understanding in topics of expertise	165
6.2.2 "I learned a lot around the AI stuff." – Learning new content about specific,	
unfamiliar topics	168

Learning varied according to the unfamiliar topics participants encountered 169
Learning varied according to the design of the workshops
6.2.3 "It might apply new sets of relations between humans and ecosystems." -
Identifying opportunities to tackle social-ecological problems
Variation aligns with participants' existing fields of expertise
Learning varied, based on the topics for discussion, set by facilitators
Learning varies, based on participants' social and economic background
Learning varied according to the spatial levels at which participants work
6.2.4 "Scenario planning is a panacea!" - Learning about new ways of thinking about
the future 177
6.3 Why did learning vary? 181
6.3.1 Learning varied because of the relevance of information for different participants
6.3.2 Learning varied because of power imbalances
Facilitation and power 191
6.4 Conclusion

Chapter 7 - Conclusion

7.1. T	Thesis Overview	200
7.2 Re	esearch Findings	201
7.2	2.1 What are the expected and reported benefits of participatory scena	rio planning?
		201
7.2	2.2 How, and under what conditions, does learning occur in Participator	ry Scenario
Pla	anning?	202
7.2	2.3 How and why does learning vary in participatory scenario planning	202
7.3 Or	Original Contributions	203
7.4 Im	nplications for practice and future research	206
7.5 Cc	onclusion	210

Bibliography	205
Appendix 1	214
Appendix 2	217
Appendix 3	220

Chapter 1 – Introduction

1.1 Introduction

The recent, influential review of participatory scenario planning (PSP), by Oteros-Rozas et al. (2015), indicates that in the past 25 years PSP has become increasingly popular as a method for addressing complex problems in social-ecological systems (SES), primarily by bringing diverse stakeholders into dialogue. Complex, uncertain and destructive problems have emerged because of unsustainable relationships between society and the environment, and require new ways of thinking and doing if they are to be tackled effectively. However, the actual benefits PSP may have for tackling complex problems in SES are poorly understood. In particular, although Oteros-Rozas et al. (2015) indicate PSP is for encouraging learning, there is limited understanding of what such learning entails, how it occurs, and how and why it varies between different stakeholders in PSP processes.

As Steffen et al. (2015) observe, industrialised societies have typically pursued rapid socioeconomic growth at the expense of environmental sustainability. Human activities are therefore threatening the earth's ability to produce the services on which society depends (Boyd and Folke, 2012). Indeed, as warned by Rockstrom et al. (2009), global environmental change is occurring at an unprecedented pace and is likely to have catastrophic impacts on a global level, as the Earth's ability to support humanity is increasingly threatened. Crutzen and Stoermer (2000) suggest humanity has become such an influential force that the Earth has entered a new geological epoch, known as the 'Anthropocene'. In line with this recognition of humanity's extensive influence on the environment, Folke et al. (2010) argue that society and environmental systems should be understood as coupled social-ecological systems (SES).

Unsustainable and inequitable conditions in SES have created complex and uncertain problems, such as climate change, biodiversity loss, and food insecurity, which Wilkinson and Eidinow (2008) refer to as 'wicked problems'. Wicked problems, first conceptualised by Rittel and Webber (1973), are described by Balint et al. (2011) as large-scale, long-term problems, that combine multiple, compounding threats and uncertainties, with distinct, conflicting public values. I refer to these problems as social-ecological problems (SEPs) to distinguish my focus on problems in social-ecological systems.

If SEPs continue unchecked, they are likely to have severe, negative impacts for both humanity and natural environments, now and in the future. As such, these problems urgently

need to be addressed, but their complexity, compounded by uncertainty about the future, makes this particularly difficult. This has led to calls for new ways of tackling SEPs, which involve the participation of stakeholders with different types of knowledge. For example, Balint et al. (2011) argue that the complexity of SEPs, and the divergent ways of understanding them, call for participatory processes that facilitate dialogue between different stakeholders. Similarly, Wilkinson and Eidinow (2008) reason that to effectively tackle SEPs, approaches are required that take a holistic perspective to understanding SES, incorporate diverse knowledges and link knowledge with practical action at multiple spatial levels. This reflects the theory of change put forward by Lewin (1997), which posits that behaviour within a system is contingent on the general institutions in which groups and individuals are embedded. Thus, Lewin argues that building a holistic understanding of the broad context in which change is needed, is a prerequisite for developing and implementing specific actions towards that change.

'Futures thinking' tools, such as 'horizon scanning,' (Sutherland and Woodroof, 2009) 'Delphi technique,' (Mukherjee et al., 2015) and 'scenario planning,' (Wilkinson and Eidinow, 2008) are often posited as methods that can help facilitate dialogue between different stakeholders, by exploring how the future could turn out, with the aim of tackling SEPs. Participatory scenario planning (PSP), in particular, has received significant attention among researchers and practitioners, as a tool for helping to deal with SEPs. In their recent review of PSP, Oteros-Rozas et al. (2015) indicate that, in the authors' extensive collective experience, the rationale behind PSP is often to 'integrate different types of knowledge,' (p.33). They describe it as a process in which stakeholders collaborate on the development of alternative, plausible accounts of potential future events, conditions and trajectories. PSP is thus advocated as an effective method for incorporating the diverse knowledges of different stakeholders, thereby facilitating dialogue that can inform decision-making and action to create more sustainable conditions in SES.

However, as Patterson et al. (2017) contend, ideas about how sustainable futures should turn out are socially constructed, and therefore contested and political. As one prominent critic of participation, Kothari (2001) observes, participatory processes commonly reproduce existing social structures, which means they often privilege the knowledges of some actors over others. Cornwall (2008) thus states that the outcomes of participatory processes typically reflect the interests of those who initiate them. This has important implications for the use of PSP to help address SEPs. Patterson et al. (2017) highlight that questions regarding how sustainable futures might look, and who gets to decide this, are commonly overlooked in participatory processes

pertaining to help tackle SEPs. In this Doctoral thesis, I therefore set out to investigate the benefits PSP may have, for those whose knowledge it incorporates.

1.2 An introduction to scenario planning

1.2.1 Methods for thinking about the future

Scenario planning is one of a range of methods that can be used to aid structured thinking about the future, although it has become especially prominent in the past 25 years. It sits alongside other 'futures thinking' approaches such as: 'backcasting,' 'Delphi technique,' 'horizon scanning,' and 'visioning,' which are outlined in Table 1.1, below.

Table 1.1 Description of methods for structured thinking about the future that havebeen used to help tackle SEPs.

Futures thinking method	Description	Purported benefits
Backcasting (Robinson, 1982) Delphi technique (Mukherjee et al., 2015)	Developing steps towards achieving desirable future conditions, by working backwards from the future to the present. Collection of experts' judgements on how a particular issue may develop in the future, using a structured questionnaire. Statistical analysis is conducted on the responses. Participants are then asked to review their responses. After several rounds of reviewing, the median of the	Developing plans for how desirable futures can be achieved. Collecting expert knowledge for use in assessing complex issues. Assumed to reduce influence of social pressures, because participants do not meet face-to- face.
Horizon Scanning (Sutherland and Woodroof, 2009)	responses is used to forecast the future development of the issue in question. Systematically looking for threats and opportunities regarding how a particular issue may develop in the future. Involves: scoping the issue, gathering information, identifying signals and observing trends, developing responses to how these trends will develop in the future.	Identifying foreseeable threats and opportunities. Selecting which threats and opportunities are most worthwhile studying and preparing for.
Scenario Planning (Van der Heijden, 1996)	Developing plausible narratives of future events, conditions and trajectories.	'Embracing' future uncertainty. Developing a more holistic understanding of complex systems and the relationships within them.
Visioning (van der Helm, 2009)	Developing ideological expressions of how desirable futures should look. Usually developed to contrast with undesirable elements of the present.	Developing visions of desirable futures is believed to help stimulate action towards achieving them.

Source: author construct

Scenario planning is not the only one of these tools to have been applied to tackling SEPs. There is evidence for the efficacy of backcasting (Robinson et al., 2011), Delphi technique (Mukherjee et al., 2015) horizon scanning (Stanley et al., 2015), and visioning (Sheppard et al., 2011). Additionally, there are cases in which combinations of scenario planning and other futures methods have been used, including: scenario planning with backcasting (Palacios-Agundez et al., 2013), scenario planning alongside visioning (Sheppard et al., 2011), as well as scenario planning in combination with Delphi technique and backcasting (Badjeck and Diop, 2011). However, as will be seen in Section 1.2.5, PSP has become particularly prominent. As such, it is the use of scenario planning that is the focus of this research.

1.2.2 Origins of scenario planning

Scenario planning (SP) was initially developed by the US military as a tool for planning potential strategies in the Cold War (Bradfield et al., 2005; Schoemaker, 1993). It was later adapted for use in corporate strategic planning, initially and most successfully by the Royal Dutch/Shell oil company, to help anticipate and respond to possible future threats and opportunities (Van der Heijden, 1996; Wack, 1985). Royal Dutch/Shell famously responded more successfully to the 1973 'Arab Oil Crisis' than their competitors, because they had already anticipated a similar eventuality using SP and considered how they would respond. As a result of this success, Bradfield et al. (2005) observe that the number of major corporations using SP doubled within just two years, and the approach became prolific in the business sector by the early 1980s.

In the world of corporate strategic planning, SP has purportedly been a highly beneficial method. According to the prominent SP scholar, Van der Heijden (1996), it is useful for helping organisations to react quickly and effectively to change. He suggests SP helps achieve this through helping organisations to develop a broader awareness and understanding of their environment, the causal relationships within it, and how well they fit into it. Another authoritative figure, Schoemaker (1991), explains SP is useful because it explores both the trends and uncertainties of an issue in a holistic and systematic way, rather than just predicting the most likely future. Likewise, Wack (1985) explains that SP helps to deal with uncertainty by breaking down the vast and complex array of future conditions into coherent narratives of alternative plausible futures. In a more recent review of SP, Amer et al. (2013) explain that this holistic and systematic approach helps organisations to appreciate drivers that may be considered insignificant, but

subsequently become highly influential. However, the apparent success of PSP in the corporate world, does not necessarily mean it has the same benefits in environmental contexts.

1.2.3 What is scenario planning?

In their review of SP techniques, Bradfield et al. (2005) point out that there is little consensus regarding definitions, characteristics and methodologies of doing SP. Indeed, a variety of definitions exist regarding what a 'scenario' is. In his influential paper on the conceptual foundations of SP, Schoemaker (1993 p.195) defines scenarios as 'focused descriptions of fundamentally different futures, presented in a coherent script-like, or narrative fashion'. Another prominent SP scholar, Van der Heijden (1996 p.195) defines scenarios as 'internally consistent and challenging narrative descriptions of possible futures'. Like Schoemaker, he views scenarios as narratives, but emphasises internal consistency and plausibility. Another influential voice in SP discourse, Schwartz (1998 p.4) defines a scenario as: 'a tool for ordering one's perceptions about alternative future environments in which one's decisions might be played out'. He thus emphasises the role of scenarios for decision-making.

Although these, and other, definitions each present scenarios in different ways, two commonalities can be found between them. Firstly, they all emphasise that SP considers *multiple*, alternative futures, rather than predicting a single, definitive future. Schoemaker phrases this as 'fundamentally different futures'; van der Heijden as plural 'futures'; and Schwartz as 'alternative future environments'. Secondly, they all highlight the importance of plausibility in scenarios. Schoemaker describes this in terms of coherence, whilst van der Heijden describes it in terms of internal consistency. Schwartz's emphasis on decision-making also implies a need for plausibility, so that the robustness of future decisions may be tested, realistically. In the context of this research, I therefore consider scenarios to be narratives of alternative, plausible, future conditions. However, this definition is quite broad and does not fully explain what scenario *planning* is. Defining SP is made more complicated by the array of different approaches that exist for practicing it. In the next section I consider four typologies of SP and then define what type of SP is of most interest for this research.

1.2.4 Types of scenario planning

Several scholars have developed typologies of different approaches. I examine four influential examples below, in the chronological order in which they were published.

- i) Types of SP suggested by Van Notten et al. The typology developed by van Notten et al. (2003) is based on three broad themes that influence the design of SP processes. Firstly, they delineate SP according to the project objectives, which range from exploring what could possibly happen in the future, to testing the potential future implications of specific actions and policies. Secondly, they delineate according to the collection and use of data in SP, which ranges from qualitative data and participatory data collection to quantitative data and deskbased data collection. Finally, they distinguish between complex scenarios, featuring a wide range of interacting variables, and simple scenarios that focus on a niche subject.
- ii) Types of SP suggested by Bradfield et al. Bradfield et al. (2005) frame their typology around three different schools of thought, in SP, that have developed at different times and in different places. First, they describe 'Intuitive Logics,' the original SP techniques developed by the US military and Royal Dutch/Shell. This involves constructing scenarios by exploring how drivers developed in the past, how they might develop in the future and how they might shape future events, conditions and trajectories. Second, they outline Probabilistic Modified Trends, which aim to account for the problem that extrapolating trends into the future does not account for what might happen if those trends are modified by unforeseen drivers and events. Finally, they describe *La Prospective*, which focuses on using computer-based mathematical models to inform public policy and shape the future in ways that are supposedly beneficial for society.
- Types of SP suggested by Borjeson et al. Like van Notten et al., Borjeson et al.
 (2006) base their typology around the theme of the project objectives, but they frame this in terms of the questions that underpin SP processes. Firstly, they suggest a category of 'predictive' SP, based on the question: 'what will happen?' These processes tend to use quantitative data and focus on the most likely future. Secondly, they propose SP may be 'exploratory,' based on the question of 'what could happen?' These processes commonly explore several, alternative futures. Finally, they suggest there are 'normative' SP processes, based on the question:

'how can a specific, desirable end-point be reached?' These can consider how current conditions may be preserved, or they may focus on how to make them more desirable.

iv) Types of SP suggested by Wilkinson and Eidinow Wilkinson and Eidinow (2008) focus specifically on SP in social-ecological contexts. They categorise approaches to SP into those focused on problems and those focused on actors. They suggest problem-focused SP processes assume that the future is knowable and that a more accurate understanding of the future is necessary to make better decisions. As such, they project historical trends into alternative futures. They describe actor-focused processes as dealing with the different understandings and relationships actors have regarding the environment. These processes subsequently involve the participation of multiple different actors to deliberate on a common issue. Wilkinson and Eidinow go on to argue that neither of the above categories pay sufficient attention to the epistemologies underpinning SP processes, for tackling SEPs. They therefore propose a new, integrated approach, which they call 'Reflexive Interventionist Multiple Actor' SP. They explain that this approach emphasises the inclusion of diverse epistemologies in SP and aims to change the way people understand how the future is created, rather than just responding to potential future events.

The literature on scenario planning for tackling SEPs typically shows limited engagement with these typologies. There is therefore much scope for critical reflection on how they could be utilised by practitioners of scenario planning. Such debates are beyond the scope of this thesis, but it is important, nonetheless, to clarify what types of SP are of most interest in this research. I therefore use these typologies to outline which types of PSP appear to be used for addressing SEPs, below.

Oteros-Rozas et al. (2015) explain that scenario planning for tackling SEPs is commonly used in a participatory way, with the aim of incorporating diverse knowledges. The participatory approaches described by van Notten et al. (2003), and the 'Reflexive Interventionist Multiple Actor' approach set out by Wilkinson and Eidinow (2008) are therefore of particular interest.

Star et al. (2016) indicate that scenario planning processes in the context of SEPs are frequently distinguished according to whether they take an exploratory approach or a normative one, according to the typology of Borjeson et al. (2006). As Star et al. (2016) point out, PSP

processes in this context increasingly incorporate both exploratory and normative elements. However, they are distinctive approaches. Exploratory, or plausibility-based approaches explore plausible future events, conditions and trajectories in an open-minded way, and analyse them to assess the consequences of different possibilities (Star et al., 2016; Wilkinson and Kupers, 2014). For example, Brown et al. (2016) take an explorative approach to explore how the future is anticipated from the perspectives of different stakeholders, and then test the robustness of different strategies for adaptation. Normative approaches, on the other hand, explore participants' preferred futures and identify actions that need to be taken to realise them (Star et al., 2016). For example, Palomo et al. (2011) developed normative scenarios to identify and develop steps towards desirable outcomes for participants. As Peterson et al. (2003) observe, scenario planning is regarded as most useful in situations of substantial complexity and uncertainty, such as SEPs, in which relying on predictions of the most likely future can lead to a false sense of certainty that is blind to variability and surprise. As such, I do not consider predictive scenarios in this research.

Johnson et al. (2012) indicate that the development of qualitative narratives is the aspect of scenario planning that enables the integration of different knowledges. However, as Oteros-Rozas et al. (2015) point out, scenario planning often incorporates qualitative and quantitative information. My main focus is, therefore, on processes that have qualitative narratives at their core, but I also include processes that incorporate quantitative data.

In the literature on scenario planning for tackling SEPs, there is little reference to the different 'schools of thought' described by Bradfield et al. (2005), which makes it difficult to select a particular 'school of thought' to focus on. However, most examples appear to develop scenarios based on how different drivers might develop in the future. For example, Palacios-Agundez et al. (2013) explore the development of drivers in ecosystem management, while Tschakert et al. (2014) consider how drivers influencing community-level adaptation to climate change, could develop. This most closely reflects the 'Intuitive Logics' approach.

To summarise, although a range of approaches to doing scenario planning exist, I focus on the types of processes that I identified in the literature on scenario planning for tackling SEPs. This appears to include approaches that are explorative, as well as those that are normative. This also includes participatory approaches that involve the construction of qualitative narratives, and that reflect the 'Intuitive Logics' approach. Hereafter, I refer to these types of SP as 'participatory scenario planning' (PSP). I consider PSP to be interactive processes in which diverse groups of stakeholders develop multiple, plausible, qualitative narratives of future events, conditions and

trajectories, with a focus on tackling SEPs. I acknowledge that, whilst my analysis is comprehensive, it does not account for absolutely every iteration of scenario planning that could be used to address SEPs. Figure 1.1, illustrates what a 'typical' PSP process consists of, taken from Bennett et al. (2016b), who provide a detailed description of the method they used for PSP. However, it is vital to note that, as Bennett et al. acknowledge, PSP is frequently modified, to fit individual contexts. Hence, this should be taken as a representative, but not definitive, example of PSP.



Figure 1.1 – Illustration of a 'typical' example of a PSP process

Source: Bennett et al. (2016b p.1777)

The figure shows that PSP usually involves: identifying the social-ecological problem to be tackled and the purpose of PSP; exploring social-ecological systems and drivers of change; developing scenarios of plausible future events, conditions and trajectories; and using the scenarios to assess responses to the futures they describe.

1.3 Scenario Planning for tackling complex problems in SES

Reviews of PSP indicate that it has increasingly been applied to thinking about SEPs and has become widespread over the past 25-30 years. However, there is a lack of clear information regarding when it started to become popular, and exactly how widespread it has become. Reviews of PSP provide largely vague accounts of when it started to be used in tackling complex problems in SES. For example, Oteros-Rozas et al. (2015) suggest scenario planning has increasingly been applied to different contexts in environmental research over the past 25 years, but provide no indication of how much it has increased, or how this increase was triggered. Alcamo and Henrichs (2009) describe how a multitude of environmental scenarios processes have emerged in the past 30 years but provide no indication of exactly how widespread these processes are, or why they have emerged, specifically, in the past 30 years.

In their description of the historical context of PSP, Rounsevell and Metzger (2010) suggest the application of PSP techniques to complex problems in SES emerged with growing concerns for the state of the global environment in the 1970s. Similarly, Wilkinson and Eidinow (2008) suggest scenarios processes have been applied to environmental contexts since the late 1970s, as a result of increasing environmental concerns like water stress, ecosystem degradation, and air pollution. Many authors, including Bohensky et al. (2009), Malinga et al. (2013), and Plieninger et al. (2013), cite the prominent Millennium Ecosystem Assessment (MA) as their inspiration for using scenario planning. In its assessment report, the MA itself makes reference to previous global environmental scenario planning processes (Carpenter et al., 2005). They cite the models used in the *Limits to Growth* by Meadows and Club of Rome. (1972) as the first example of scenario planning being applied to environmental concerns.

As Ramirez and Wilkinson (2016) observe, practice and scholarship on PSP are increasingly fragmented along differences in approach, success criteria, methods and techniques. This research speaks primarily to PSP practice in the academic field of social-ecological resilience, much of which stems from the MA, in which PSP is typically used by researchers with a view to explore the complex dynamics of SES, influence decision-making, and foster learning (Oteros-Rozas et al., 2015). Literature on the use of PSP in this context demonstrates that it has been applied to a wide range of SEPs, including: sustainable energy production (Rivard and Reay, 2012), ecosystem service management (Malinga et al., 2013), climate change adaptation (Tschakert et al., 2014), food and nutrition security (Vervoort et al., 2013), conservation (Waylen et al., 2015), and land-use management (Swetnam et al., 2011). PSP has been applied at a range of spatial levels, including global (Carpenter et al., 2006; Kok et al., 2016), regional (Vervoort et al., 2013),

national (Bohensky et al., 2006), and local (Mistry et al., 2014), and in locations across the world, including Western Europe (Reed et al., 2013), Eastern Europe (Hanspach et al., 2014), South America (Brown et al., 2016), South East Asia (Bohensky et al., 2009), Australia (Pearson et al., 2010), Sub-Saharan Africa (Chaudhury et al., 2012), and the Arctic Circle (Wesche and Armitage, 2014).

In this field, reviews of the benefits PSP may have for tackling SEPs are limited, despite its apparent widespread use. This indicates a relative immaturity in PSP practice in the field of socialecological resilience. However, reflection on the benefits of PSP in this field appear to be receiving increased emphasis. Indeed, Waylen et al. (2015) call for greater reflection on the benefits PSP may have, as well as its strengths and weaknesses, whilst the review by Oteros-Rozas et al. (2015) claims to be the first to synthesise information on the benefits of PSP from across different cases, in which PSP was applied to SEPs. They find that PSP encourages learning about possible future developments in SES and enables a 'shared understanding' through sharing different knowledge. However, it is notable that all the authors involved in the paper were themselves involved in the cases they wrote about. As such, they arguably had an interest in portraying PSP as beneficial. In the paper, the authors do not explicitly reflect on whose knowledges and interests are reflected in the shared understanding they report, or indeed on the extent to which different stakeholders actually buy into it. Equally, the authors do not detail exactly what learning involves, or how it occurs in PSP. Admittedly, the authors may have found it difficult to include such detail in a review paper on the general benefits of PSP.

As indicated in Section 1.1, PSP processes could replicate existing social structures, thereby privileging the knowledges of some actors over others (Kothari, 2001). The outcomes may thus be defined by those who initiate them, not by the participants themselves (Chambers, 1997). It is therefore important for research to assess the benefits PSP may have, for those whose knowledge it is claimed to incorporate. In this study, I contribute to understanding the benefits of PSP, from an independent, critical perspective. I do so by addressing the research aim and questions, set out below.

1.4 Research Aim and Questions

Research Aim

 The aim of this research is to assess the benefits participatory scenario planning processes may have for tackling social-ecological problems.

Research Questions

- 1. What are the expected and reported benefits of participatory scenario planning?
- 2. How and under what conditions does learning occur through participatory scenario planning?
- 3. How and why does learning vary in participatory scenario planning?

1.5 Thesis Outline

In this thesis, I contribute to understanding the purported benefits of PSP in the field of social-ecological resilience. I focus specifically on learning, given the emphasis placed on using PSP to incorporate the knowledges of different stakeholders in dialogue about tackling SEPs. I approach the research from a critical perspective on participation. This is informed by Kothari (2001), who contends that participatory processes commonly reproduce existing social inequalities, such that different knowledges are privileged over others. My research is therefore motivated by a need to understand the benefits of PSP for those whose knowledge it is claimed to incorporate. The thesis is structured as follows:

In Chapter 2, I link the use of PSP to wider literature on social-ecological resilience. I highlight an assumption that bringing together stakeholders with different worldviews and fields of expertise can help encourage more equitable and sustainable conditions in SES. I then draw upon participation literature, which emphasises the importance of investigating the purported benefits of PSP. I develop a conceptual framework for understanding how learning occurs, and how and why it varies in PSP. This is underpinned specifically by the work of the cognitivist learning theorist, Vygotsky (1978). He asserts that people learn through interactions with others,

which enables them to 'enter' their 'zones of proximal development,' or to think 'outside the box.' I use the concept of 'boundary objects,' (Star and Griesemer, 1989) to explain how PSP can prompt discussion between different people, and the concept of 'scaffolding,' (Wood et al., 1976) to explain how facilitators of PSP can encourage learning, by helping participants to engage in PSP processes. To understand how and why learning varies, I draw on the credibility, salience and legitimacy framework, set out by Cash et al. (2003). This explains that people are most likely to learn when they encounter knowledge they consider to be based on valid evidence, relevant to their requirements and produced through a fair, inclusive and unbiased process. I link this to literature on power and adult learning, specifically Brookfield (2005), which helps to explain how power imbalances between participants, and between participants and facilitators can influence who learns what.

In Chapter 3, I set out the research methodology I used to explore learning in PSP. The focus on interactions between different participants, emphasised the importance of accounting for how different participants make sense of the knowledge they encounter in PSP. My methodology was, therefore, underpinned by a constructivist, qualitative approach that enabled me to explore how learning is shaped by the meanings and interpretations participants attribute to knowledge encountered in PSP. Data were collected from three sources: Firstly, I conducted a case review, analysing 30 cases reported on in the peer-reviewed literature, in which PSP is used to help tackle SEPs. Secondly, I conducted semi-structured interviews with practitioners (researchers, project coordinators, professional facilitators) of PSP. Thirdly, I undertook two case studies of specific PSP processes – 'Positive Futures for Southern Africa,' (PFSA) and 'Food Security Futures,' (FSF). I analysed data from these three sources and linked my analysis back to the conceptual framework. In doing so, I developed an understanding of the expected and reported benefits of PSP, how, and under what conditions learning occurs, and how and why it varies for different participants.

In Chapter 4, I describe the expected and reported benefits of PSP, as well as its limitations. I find that the use of PSP is typically underpinned by an assumption it can break down, or structure future possibilities into alternative narratives. This is understood to help participants explore the complexity and uncertainty of SEPs. However, this is not frequently reported as a benefit, because it is linked to, and sometimes acts as a precursor to other benefits. Learning is commonly expected as a benefit of PSP and is often, reportedly, achieved through bringing different participants together to deliberate on the future. However, I find that practitioners typically have limited theoretically-grounded understanding of how learning occurs and appear to

give limited attention to how and why learning varies for different participants. I also find that proponents of PSP commonly expect and report that it helps develop responses to SEPs and test them in alternative future conditions. Again, there is sparse evidence of these responses being implemented. This could be because of limited connection of PSP with long-term processes of policy-making and action. I conclude the chapter by highlighting a lack of provision for systematic evaluation of PSP processes, especially regarding its strengths and weaknesses compared to other methods.

In Chapter 5, I present my findings that explain how, and under what conditions, learning occurs in PSP. I show that learning occurs through interactions between participants with different worldviews, fields of expertise, and social and economic backgrounds. PSP can act as a 'boundary object' (Star and Griesemer, 1989) that prompts discussion between participants, and thus exposes them to different knowledges. PSP can thus stimulate creative, focused thinking about the future of SEPs, which encourages participants to push beyond their usual range of thinking In this way, PSP enables participants to enter their 'zones of proximal development,' as described by Vygotsky (1978). Facilitators can encourage participants to engage in creative, structured thinking, by explaining, and guiding them through, specific activities in PSP processes, and through using questions to prompt discussion. Conducive conditions for learning to occur in PSP include careful selection of participants who have relevant expertise and are willing to listen to, and deliberate on, the views of others, and skilled facilitation to ensure interactions between participants are constructive.

In Chapter 6, I show how learning varies between different participants, and explain the reasons for these variations. I find four distinct themes of learning in PSP: i) refining existing understanding on a topic of expertise, ii) learning new content about specific, unfamiliar topics, iii) identifying opportunities for tackling SEPs, and iv) learning about the value of PSP for thinking about the future. I find that learning can vary, according to participant's prior expertise, and ignorance, in specific topics, their social and economic backgrounds, their previous experience of PSP, the spatial levels at which they focus, and the design of specific PSP processes. I then show that learning varies according to the extent to which participants find the knowledge they encounter to be relevant to their interests, and produced through fair, unbiased processes. I also find that facilitators play a significant role in shaping learning by defining what is considered acceptable for discussion. Facilitators themselves are influenced by the objectives of wider projects, in which specific PSP processes are embedded.

In Chapter 7, I conclude the thesis by raising the question, if learning is shaped by facilitators, and the objectives of wider projects, then who and indeed what is PSP really for? I argue that greater critical reflection is needed, by practitioners, on the societal impact of PSP. I assert more robust governance is needed to ensure best practice and systematic evaluation of PSP processes in this context. Future research could thus usefully focus on if and how learning helps participants to take more informed decisions and actions to tackle SEPs. There is also a need for critical reflection on the role of facilitators, and for adequate training of those wishing to facilitate PSP. Greater understanding is also required, regarding the relative strengths and weaknesses of PSP, as compared to other futures-oriented, and narrative-based methods, for tackling SEPs.

Chapter 2 – Literature Review and Conceptual Framework

2.1 Introduction

In Chapter 1, I introduced participatory scenario planning (PSP) as interactive processes in which diverse groups of stakeholders develop multiple, plausible narratives of future events, conditions and trajectories. I showed that PSP has increasingly been used with the expectation, by researchers and practitioners, that it can help tackle social-ecological problems (SEPs) by bringing diverse stakeholders into dialogue to build a holistic understanding of them. In the first part of this chapter, I explore literature that reviews and explains the purported benefits of PSP for tackling SEPs. I show that PSP scholars consider it to be particularly useful as a tool for taking account of the knowledges of different stakeholders, and thus encouraging learning. However, research that assesses the benefits of PSP remains limited, especially regarding how, and under what conditions learning occurs, and how and why learning varies between different participants.

In the second part of this chapter, I explore how theory on learning can help explain how learning occurs in PSP. I reason that the cognitivist theory, the 'Zone of Proximal Development,' (Vygotsky, 1978) is useful for explaining how learning occurs through interactions between participants with different fields of expertise. I build on this by arguing that the process of PSP acts as a 'boundary object,' (Star and Griesemer, 1989) which prompts discussion between different participants. I also propose that the concept of 'scaffolding' (Wood et al., 1976) may be used to explain how facilitators can help participants to learn, through assisting them to engage in PSP processes. I then draw on the credibility, salience and legitimacy framework, set out by Cash et al. (2003) to explain that learning could vary based on the extent that participants find the knowledge they encounter to be credible, salient and legitimate. I use literature on learning and power (Brookfield, 2005) to show how paying attention to power imbalances between participants, and between facilitators and participants can help explain variations in participants' learning. Finally, I draw these theories together to propose a conceptual framework that can be used to study learning in PSP.

2.2 PSP, participation and social-ecological resilience

As indicated in Chapter 1, PSP is increasingly used with a focus on addressing the challenges posed by prevailing unsustainable and highly uncertain conditions in social-ecological systems (SES). However, there has thus far been limited assessment of how and why it is useful in this context. In their review of PSP, Oteros-Rozas et al. (2015) link the rationale for PSP to broader participation discourse, regarding perceived normative and pragmatic benefits of stakeholder participation in deliberating on the management of SES. They cite Stringer et al. (2006) who state that participation can encourage collaborative and democratic management of SES by facilitating the sharing of knowledge between different stakeholders. Stringer et al. thus explain that participation can enable learning and the development of 'shared understandings' of SES between participants. Similarly, Butler et al. (2015) argue that because complex problems span multiple spatial levels, and stakeholders at different levels have different knowledges and interests, the different stakeholders should be involved in planning adaptations to those problems. They find that planning processes involving the participation of stakeholders from multiple levels, encouraged knowledge exchange and learning between them.

As indicated by the experienced PSP practitioners Henrichs et al. (2010) there exist several modes of participation, and the one that is used depends on the purpose of individual PSP processes. Henrichs et al. explain that for processes aimed at research and scientific exploration, the role of participation is to include scientific experts from different disciplines, as well as stakeholders with relevant expertise, to draw on their insights in the field of study. In a process that aims to educate stakeholders, the role of participation is to include a wide range of different actors to enable discussions incorporating diverse perspectives, as well as to empower the voices of marginalised stakeholders. In processes aimed at supporting decision-making, participation is used to allow stakeholders to drive the process, with the aim of improving its relevance and legitimacy for key decision-makers.

The use of PSP to incorporate diverse knowledges reflects wider notions in contemporary literature on social-ecological resilience, which commonly assume that dialogue and collaboration between different stakeholders can help achieve 'transformations for sustainability,' or 'fundamental changes in structural, functional, relational, and cognitive aspects of socioecological systems, that lead to new patterns of interactions and outcomes,' (Patterson et al., 2017 p.2). Indeed the prominent resilience scholars, Boyd and Folke (2012) assert that such transformations require actors from across different levels and sectors to pool their knowledges and experiences of SES. This sort of thinking has resulted in a widespread assumption that

processes of knowledge exchange, or the development, sharing and use of knowledge, in different ways, for different contexts, are necessary for creating more sustainable conditions in SES (Fazey et al., 2013).

Recent literature on PSP therefore presents it as a method for integrating the knowledges of different stakeholders to help tackle SEPs. Indeed, Oteros-Rozas et al. (2015) indicate that in their experience, the rationale behind PSP is often to: 'integrate different types of knowledge,' (p.33). Similarly, Kok et al. (2016) state that it is a useful tool for taking account of the knowledges and perspectives of different stakeholders, whilst Bennett et al. (2016a) claim that it can help to consider diverse perspectives on how just and sustainable futures might look. It is evident then, that PSP is commonly viewed as a tool that can enable stakeholder participation, with the aim of tackling SEPs.

This reflects the theory of change proffered by the influential social psychologist, Lewin (1997). Lewin argues that societal problems are contingent on the general laws and institutions of the system in which they are embedded. He therefore posits that it is vital to understand the system as a whole, before specific decisions and interventions can be made to change it. Ideas about transformations to sustainability appear to reflect this in their assumption that including diverse knowledges in dialogue through PSP can help build a more holistic perspective of SEPs, and thus help identify opportunities to tackle them. I therefore find it useful to consider the benefits of PSP in relation to Lewin's theory of change.

However, as pointed out by Patterson et al. (2017), ideas about sustainability are socially constructed, and are therefore highly contested and political. The effects of change will be different for different actors and will thus produce winners and losers. They subsequently argue that insufficient consideration has been given to the power imbalances involved in tackling SEPs. To understand the benefits of PSP, it is therefore important to consider how power imbalances shape the benefits of PSP, and how they can reproduce the conditions that create winners and losers. It is therefore useful to consider some of the seminal critiques of participation that exist in the literature.

In one of the most influential critiques of participation, Kothari (2001) contends that, although participation often aims to include the knowledges of marginalised groups in the production of knowledge, knowledge is diverse, power-laden and socially constructed. Kothari therefore argues that processes purporting to include diverse knowledges are heavily infused with power dynamics, which can shape their outcomes. As I set out above, PSP is typically used with

the intention of incorporating diverse knowledges. The implication of Kothari's critique is that different participants, coupled with the knowledges they contribute, are privileged over others. This means the outcomes of PSP for different actors are likely to reflect the power imbalances between them.

The influential participation scholars, Cornwall (2008) and Chambers (1997) both highlight the important power imbalances between the actors who initiate and facilitate participatory processes, and those who participate in them. Cornwall asserts that participation has the potential to encourage collective action by participants that can foster social change. Similarly, Chambers supposes that participation can move research from a process of information extraction, to one that empowers participants, as well as creating useful outcomes for them. This is reflected in the assumptions in the resilience literature, that stakeholder participation can produce useful outcomes for tackling SEPs, through knowledge exchange and learning. However, Cornwall also contends that participatory processes are typically owned by whoever initiates them, whilst Chambers acknowledges that the facilitators of participatory processes ultimately have power over their outcomes. This means the actors who initiate participatory processes have significant power over shaping the discussions that occur within them, and the outcomes they produce.

It is evident then, that PSP is considered useful for taking account of the knowledges of different stakeholders, with the aim of informing approaches to tackling SEPs. However, the participation literature indicates that the benefits of PSP are shaped by the power of those who initiate and facilitate PSP processes, as well as by the intricate power imbalances between participants. It is important, therefore, to understand what the benefits are for those whose knowledges are purportedly included in PSP processes. I explore some of the perceived benefits of PSP in Section 2.3, below.

2.3 Perceived benefits of PSP

To develop an understanding of what exponents of PSP consider it to be useful for, I look first to the literature on the origins of scenario planning in corporate strategic planning. In his highly influential work, one of the founders of scenario planning, Wack (1985), describes how it is useful for expanding people's thinking. He explains that people construct their views of reality through the assumptions they hold about the world. Wack states that scenario planning helps to structure uncertainty by breaking down future possibilities into coherent narratives. In this way, he explains, scenario planning enables people to articulate and reflect on their assumptions about

the world, consider alternative perspectives, and thus develop a broader understanding of whatever system they are operating within.

Similarly, two other influential early scenario planning scholars, Schoemaker (1993) and Van der Heijden (1996), both indicate scenario planning can help people to expand their understanding of the world by structuring highly uncertain futures into manageable narratives. Specifically, van der Heijden states that it can help people to understand complex causal relationships between different drivers. In a more recent review of scenario planning, the Futures Studies scholars, Amer et al. (2013), also indicate that it can help people to think about complex systems in a more strategic way, by breaking down the complex array of uncertain possibilities into manageable storylines about the future.

The scenario planning specialists Ramirez and Wilkinson (2016) build on this earlier work to describe how scenario planning can be useful for enabling 'reframing' and 'reperception.' They describe reframing as a process of exploring alternative future contexts, or frames, which leads to an exchange of different perspectives, and thus the creation of new knowledge and shared perspectives, as well as consideration of different options for action. They describe reperception as the identification of new courses of action to be taken for achieving change. This links closely to Lewin's (1997) theory of change. Reframing can help groups and individuals to develop a more holistic understanding of the general system in which they are embedded, whilst reperception enables them to make decisions and act to deal with a specific situation.

This literature from the worlds of corporate strategic planning and Futures Studies is grounded in a different context to social-ecological resilience, but it provides a useful foundation for understanding what PSP may be useful for. As shown above, scenario planning is considered a useful tool for breaking down uncertain and unwieldy future possibilities into manageable sets of plausible narratives. This process is believed to enable learning, in terms of developing a broader understanding of complex systems. However, Wack, Schoemaker and van der Heijden refer explicitly to corporate organisations, and their managers as the beneficiaries of scenario planning. Ramirez and Wilkinson (2016) refer to a specific approach, referred to as the Oxford Scenario Planning Approach (OSPA). The approaches, beneficiaries, and perceived benefits of PSP in socialecological contexts may be rather different.

Several attempts to explore the benefits and drawbacks, of PSP in SES exist, many of which were inspired by the influential Millennium Ecosystem Assessment (MA), mentioned in Chapter 1. For example, the academics who were involved in using PSP in the MA, Biggs et al.

(2007), and Kok et al. (2007) state that PSP is useful for exploring the future development of SES, identifying actions that may be taken to deal with future developments, as well as encouraging knowledge sharing and communication between different participants, particularly across different spatial levels. However, both papers indicate that it is difficult for PSP to remain relevant and legitimate to participants at different levels. Biggs et al. emphasise that PSP processes can easily become hijacked by the interests of stakeholders at a given spatial level, which can side line the interests of others. This emphasises the importance of understanding the benefits of PSP for different participants.

In the past decade, there has been a growing recognition in PSP practice in the field of social-ecological resilience that PSP is most useful as a tool for learning about social-ecological problems (Brown et al., 2016; Johnson et al., 2012; Oteros-Rozas et al., 2015; Varum and Melo, 2010). As described in Section 2.2, there exist different rationales for using PSP. Henrichs et al. (2010) describe these as: research and scientific exploration, education, and supporting decisionmaking. Henrichs et al. explain that, although each of these rationales has a different mode of participation, the participation of stakeholders in any form of PSP process can result in learning for those involved. Assumptions about learning are often linked to the idea that PSP can incorporate the knowledges of different stakeholders in deliberations on how to tackle SEPs. In their often-cited paper on PSP, Johnson et al. (2012) describe it as a 'vehicle for learning,' (p.10) because it enables people to learn through engaging with different perspectives on an issue. They indicate that PSP enables this by incorporating diverse knowledges into the narratives of future events, conditions and trajectories that it creates. The authors go on to assess whether learning occurred in a particular PSP process. They find that participants learned about others' perspectives, developed new social networks, and identified opportunities for tackling socialecological problems. However, although Johnson et al. provide evidence that learning occurs in PSP, they do not provide a detailed analysis of how, or for whom learning occurs. They imply that learning occurs through the incorporation of diverse knowledges, but they do not explain how this can result in learning. Equally, they do not draw upon learning theories that could help to explain how learning occurs in PSP.

The importance of incorporating diverse knowledges for encouraging learning is reflected more recently by Oteros-Rozas et al. (2015). In their paper, they synthesise the outcomes of 23 cases in which the authors used PSP, to assess what its apparent benefits were. They find that it enhanced the engagement of participants with different knowledges, and that through it, participants learned about the impacts of global level changes on the local level and could better

understand the complexity of SES. They indicate that PSP is most likely to result in these learning outcomes when it is linked to objectives that are shared by both researchers and participants. However, as shown in the literature on participation, in Section 2.2, the actors who initiate and facilitate participatory processes usually maintain control over their outcomes. Indeed, Oteros-Rozas et al. acknowledge that their PSP processes were shaped by value judgements about sustainability that were at the core of researchers' and funders' motivations. It is important to question, therefore, the extent to which the objectives of the researchers were shared by the beneficiaries, and how this influenced the learning that occurred.

In the accounts of Johnson et al. (2012) and Oteros-Rozas et al. (2015), there is an implicit assumption that learning occurs through encountering different knowledges during PSP processes. However, they do not detail the specific attributes of PSP that can lead to learning. Furthermore, they show that learning may be shaped by the values of those who initiate and facilitate PSP. However, these dynamics have been given limited attention in PSP literature. In this research, I address these gaps in understanding, by exploring how learning occurs, and how and why it varies for different participants, in PSP. To do this, I draw on pedagogical theories, as well as literature on learning and power. I use these to develop a conceptual framework, presented in Section 2.6, to help study how learning occurs, and how and why it varies.

2.4 How can learning theories contribute to understanding how learning occurs in PSP?

In this section, I explore theories that can help to understand how learning occurs in PSP. I begin by exploring classical, cognitivist perspectives. In particular, I propose that the Zone of Proximal Development (ZPD), described by Vygotsky (1978), is a useful concept for explaining how learning occurs in PSP. Vygotsky explains that learning occurs through interactions between people. However, I go into more detail than just explaining learning as the result of interactions and explain how interactions are encouraged through the creation of 'boundary objects' (Star and Griesemer, 1989) in the process of creating and analysing scenarios, and through 'scaffolding,' (Wood et al., 1976) through which facilitators can assist participants to engage effectively in PSP processes. I explore these concepts further, below, but first it is important to define what I understand learning to be.

2.4.1 What is learning?

Unsurprisingly, definitions of learning vary according to different learning theorists. It is noted by De Houwer et al. (2013) that learning scholars are rarely explicit about how exactly they define learning. They observe there is a common assumption in learning literature, that learning is simply any change in an organism as a result of experience. Indeed one broad definition of learning, provided by Parker (2005), states that learning, at its most basic, is simply change on the part of the learner, as a result of external stimuli. However, De Houwer et al. (2013) contend that this definition of learning is too broad and learning should be defined more specifically.

A more detailed definition of learning is provided by the prominent learning theorist, Knud Illeris. He defines learning as: 'any process that in living organisms leads to permanent capacity change and which is not solely due to biological maturation or ageing' (Illeris, 2009 p.7). In this definition, Illeris also presents learning as change, as a result of external stimuli. However, he goes on to emphasise the importance of the internal cognitive processes through which people interpret their experiences. He therefore presents learning as more than just a simple reaction to external stimuli and indicates that the learner must make meaning out of them for learning to occur.

Another influential learning theorist, Peter Jarvis, presents a similar definition, and provides more detail on what making meaning out of external stimuli involves. He describes learning as a process, in which a person interprets an experience, then integrates it into their knowledge through practical, cognitive, and/or emotional transformation (Jarvis, 2005). Jarvis thus emphasises the importance of learners interpreting their experiences of external stimuli. He also emphasises that learners' interpretations are shaped by their social context. As I discuss in Section 2.5, the social context in which learning occurs is an important consideration for understanding who learns what, and why.

It is evident then, that two important themes exist in how learning is defined: change, and interpreting experience. In this research, I consider learning to be a change in a person's cognitive or physical capacity, that results from that person interpreting their experiences of external stimuli. With this definition in mind, I turn to explore two traditions of learning theory, and how they may contribute to an understanding of learning in PSP.

2.4.2 Traditions of Learning Theory

In this section, I begin by considering how behaviourist and cognitivist traditions of learning theory may help to understand learning in PSP. I argue that the behaviourist perspective is useful in so far as it highlights how changes in behaviour serve as indicators of learning. However, I find that the cognitivist perspective is more useful for explaining how learning occurs.

Behaviourist Learning Theory

Early protagonists of behaviourism, such as Skinner (1972), rejected assumptions that human behaviour was the result of character traits, emotions and states of mind. Instead, Skinner argues that human behaviour is shaped purely by environmental stimuli. The implication is that human learning occurs through the reinforcement or punishment of particular behaviours by these stimuli. People thus learn to behave in a way that avoids punishment, which in turn is rewarded by environmental stimuli. In the same vein, Skinner assumes that human learning is demonstrated through overt skills and behaviour. A study of learning, informed by behaviourism, would therefore need to identify and observe changes in the skills and behaviour demonstrated by learners.

However, Jarvis et al. (2003) contend that the behaviourist perspective leaves insufficient room for creativity and freedom on the part the learner. They contend that behaviourism implies that learners merely learn to conform to behaviours that avoid punishment and gain reward. I therefore reason that behaviourism does not adequately account for the potential role of hypothetical thinking about the future, in PSP. Nevertheless, Merriam and Bierema (2013) observe that behaviourist assumptions continue to underpin many approaches to learning to this day. Roessger (2012) attributes the continued prominence of behaviourist assumptions to the easily-measurable nature of changes in behaviour, which are taken as evidence for learning. As such, changes in behaviour could be useful indicators of learning in PSP.

Cognitivist Learning Theory

Cognitivist perspectives on learning move away from behaviourism by focusing on the relationship between learning and the development of cognitive processing capacity. The pioneer of cognitivist perspectives on learning was Jean Piaget. Piaget (1929) developed a five-stage framework of cognitive development and associated learning from infancy to age fifteen. He proposes that children begin with basic sensory and motor skills then, ultimately, develop formal operational skills, which make them capable of abstract conceptualisation and applying abstract concepts to practice. Piaget assumes that the process of learning about a particular subject follows the process of cognitive development. As such, a learner may only develop abstract

concepts about a subject once s/he has progressed through the four preceding stages of development.

However, as Jarvis et al. (2003) contend, Piaget only theorised the stages of cognitive development up until age 15, yet development and learning continue well beyond childhood. On a different tack, Vygotsky (1978) critiques Piaget's theory on the basis that it assumes learning may only progress once a particular level of cognitive development has been attained. Instead, Vygotsky argues that learning occurs first and then leads to cognitive development. Vygotsky subsequently proposes his own influential learning theory, the 'Zone of Proximal Development.' In this, he distinguishes between *actual* cognitive development, or the level of capacity a person has for independent problem-solving, and the *potential* problem-solving capacity a person has when assisted by others. He refers to this potential capacity as the Zone of Proximal Development (ZPD). This is illustrated in Figure 2.1, below.

Figure 2.1 Illustration of the Zone of Proximal Development (ZPD) and Zone of Current Development (ZCD).



Source: Harland (2003 p.265)

Vygotsky assumes that an individual's capacity for learning increases when they receive assistance from others, who are more capable in the subject that is being learned about. When such assistance is provided, he refers to this as 'entering' the ZPD. This is illustrated in Figure 2.1, in which the ZCD, or 'Zone of Current Development,' represents what a person can learn through deliberating on a problem independently. The ZPD, or 'Zone of Proximal Development,' represents the potential learning that could occur through collaboration with more capable others. When the potential learning occurs successfully, the ZCD expands to meet the outer edge of what was the ZPD, whilst the ZPD expands further. In a more recent description of the ZPD,

Wals and Dillon (2013) explain the ZPD as the potential learning that can occur through interactions with other people. They indicate that such interactions can help people to understand things they would have been unable to, without being encouraged or challenged by another.

Vygotsky's Zone of Proximal Development has played a significant role in influencing how learning is understood and facilitated. For example, the use of the ZPD underpins Engestrom's (1987) expansive learning theory, and Wood's (1976) concept of 'scaffolding,' both of which are established approaches, in their own right, to understanding learning. However, Chaiklin (2003) warns that Vygotsky specifically intended for the ZPD to explain cognitive development in children, which he describes as the progression of an individual's capacity to cope with their social and environmental context. He therefore urges that careful consideration should be given when using it to understand learning in other contexts.

Vygotsky specifically states that learning occurs through interactions between learners and people who are more capable in a given field, or who have a more advanced level of cognitive development. In PSP, the participants and facilitators are typically adults, rather than children, and therefore have similar levels of cognitive development. The relevance of the ZPD for explaining learning in PSP could therefore be questioned. However, if PSP leads to participants encountering different knowledges, as indicated in Section 2.3, they are likely to encounter others who are more capable in different fields of expertise. If interactions with participants from different fields of expertise lead to learning that helps them tackle social-ecological problems, this equates to development of their capacity to deal with their social and environmental context.

For example, a smallholder farmer is likely to have more expertise about localised rainfall patterns in a village, than a climate scientist. Conversely, the climate scientist is likely to have more expertise in global atmospheric processes than a smallholder farmer. If both the farmer and climate scientist were asked to apply their knowledge to exploring the possible effects of increased climate variability, each participant would benefit from interactions with the other in learning about certain aspects of the problem. The smallholder farmer might learn about global atmospheric processes, and the climate scientist would learn about local rainfall patterns. I therefore reason that the ZPD helps to explain how learning occurs in PSP, and use it as the baseline for my conceptual framework, which I describe in more detail in Section 2.6.
2.4.3 Boundary Objects

As described above, the Zone of Proximal Development (ZPD) provides a useful way of understanding learning in PSP, as the result of interactions between participants from different fields of expertise. However, in this research, I aim to develop a detailed understand of the specific attributes of PSP that enable participants to 'enter' their ZPDs. One useful way to explain this is through the concept of 'boundary objects,' (Star and Griesemer, 1989).

The concept of boundary objects fits within wider discourse about 'boundary work.' Boundary work, as described by Guston (2001), involves processes that negotiate the exchange of knowledge between science and policy, and/or between different social worlds. Guston describes how such processes of negotiation are often carried out by 'boundary organisations,' that manage the tensions between research and practice by negotiating the exchange of knowledge, whilst maintaining accountability to both sides. White et al. (2010) describe how boundary organisations typically create spaces for the participation of different stakeholders and are facilitated by professionals who serve a mediating role between them. As White et al. go on to describe, boundary organisations also provide opportunities for the creation of boundary objects.

Boundary objects were first conceptualised by Star and Griesemer (1989) as material, or abstract objects, that occupy several interacting, intellectual worlds and remain relevant and outwardly acceptable to all of them. They explain that boundary objects thus facilitate effective communication between diverse actors, which helps them to cooperate despite their disciplinary, and other, differences. In this way, White et al. (2010) reason, boundary objects enable the negotiation and exchange of knowledge between different groups, whilst allowing groups to uphold their own, unique identities. An illustrative example would be if a farmer and a hydrologist are deliberating on the management of a river basin. To the farmer, the river basin is a source of water and nutrients for crop growth and, ultimately, income. Meanwhile, to the hydrologist it is a subject of scientific interest and a system in need of effective management, in which the farmer is a user. The creation of a boundary object, by the two actors, such as a land-use map, could enable them to exchange their knowledge about managing the river basin, whilst allowing the farmer to continue viewing it as a source of income, and the hydrologist to continue viewing it as a subject of scientific interest.

As Star (2010) clarifies, a 'boundary' in this context is not the physical edge of something. Instead, she states that it is a physical or conceptual space that is shared by actors from different social worlds. Star also emphasises that an 'object' is not necessarily a material thing, but can be a concept that people work towards and with. In correspondence with Chaudhury et al. (2012), I

propose that the process of imagining plausible futures in PSP, fits particularly well with this description of a boundary object. The inclusion of different knowledges in PSP means the scenarios are spaces that are shared by participants from different social worlds. Scenarios are also conceptual objects that participants work towards and with. Participants work towards the creation of scenarios, and then work with the scenarios to develop and test responses to social-ecological problems.

2.4.4 Scaffolding

Earlier in this section, I presented Vygotsky's Zone of Proximal Development as an especially useful way of understanding how learning occurs through interactions between participants with different fields of expertise. I also proposed that viewing scenarios as boundary objects helps understand how PSP encourages these interactions. However, when Vygotsky (1978) proposed the ZPD, he argued that that learners require *assistance*, not just *interaction*, for them to enter the ZPD and for learning to occur. Wood et al. (1976) suggests such assistance may come in the form of 'scaffolding,' which he describes as someone with more expertise than the learner, helping them to complete tasks they would not have been able to complete alone.

In Wood's conceptualisation of scaffolding, he describes it as a process in which experts control the elements of a task that are beyond a learner's existing capacity. Learners are thus gradually introduced to those elements and supported to complete them. For example, a driving instructor begins by introducing a learner to the basic skills of operating a car – starting the engine, using the pedals, moving the car forwards and backwards – before gradually introducing them to more advanced skills – making three-point turns, negotiating roundabouts and overtaking. The instructor supports the learner by giving instructions and demonstrations, and asking questions to prompt the learner's thinking.

A more recent description is provided by Van der Pol et al. (2010) in their review of scaffolding literature. They describe scaffolding as a three-stage process involving: 'contingency,' 'fading' and 'transfer of responsibility'. They define 'contingency' as tailoring the support, provided by an expert, to a learner's existing ability. For example, the driving instructor designs each lesson according to the learner's existing ability to drive a car. The first lesson focuses on operating the pedals, the second on negotiating roundabouts, and so on. 'Fading' is described as decreasing the level of assistance, as the learner becomes more competent. To begin with, the driving instructor will give frequent instructions and ask frequent questions, but as the learner

becomes more competent at negotiating roundabouts, this support will decrease. 'Transfer of responsibility' is described as transferring the responsibility for learning from the expert to the non-expert. In the example of the driving instructor, eventually they provide no guidance at all, for negotiating roundabouts, and the learner is expected to continue developing their skills independently.

However, Van der Pol et al. (2010) also critique this view of scaffolding because it assumes that what is learned is predefined by the expert. For example, the driving instructor defines that the learner will develop a particular set of skills and understanding that will enable them to safely drive a car. Instead, Van der Pol et al. argue that learners should be viewed as active participants, rather than recipients of knowledge, and scaffolding should be viewed as an interactive process, in which learners and experts create new knowledge together. This reflects the argument made by learning scholars, Fernández et al. (2001). Fernandez et al. point out that Wood views scaffolding as occurring between experts and non-experts, but they contend it can occur in peer-to-peer interactions as well. Specifically, they indicate that peer-to-peer scaffolding occurs when people engage in what they call 'exploratory talk.' They describe this as a process of people proposing new ideas, and then giving and receiving critical, but constructive feedback from others. This resembles the process of a group of writers writing a play together, if one writer suggests an idea, and the others give critical and constructive feedback on it, they collectively develop the events, characters and storylines that make up the play. In the same way, exploratory talk involves learners engaging critically and constructively with others' ideas. This enables them to develop new understandings, and in the process to drive the learning process forward.

In PSP, both expert-learner and peer-to-peer scaffolding are valuable for understanding how learning occurs. It is typically the case that one or more facilitators guide the participants through PSP processes. However, the role of the facilitator is something that appears to have received scant attention in PSP literature. Facilitators can be said to provide scaffolding that helps participants engage in, as well as learn through PSP. They provide 'contingency' through designing stages of PSP processes that gradually introduce participants to thinking about the future in a structured way, 'fading' through decreasing the amount of instructions they provide as participants become familiar with the process, and 'transfer of responsibility' as they leave participants to let their discussions run their course.

Scaffolding could also occur between different participants in PSP. As indicated in Section 2.2, PSP is commonly used to incorporate different knowledges. Different participants therefore have expertise in different aspects of a social-ecological problem. Equally, participants engage in

PSP as active participants in an interactive process. As such, they are more likely to receive peerto-peer scaffolding from each other, as active participants, than to receive scaffolding from an expert. The concept of scaffolding is therefore useful for understanding how learning occurs through PSP.

In sum, I reason that learning in PSP can occur through interactions between participants with different fields of expertise, as explained by Vygotsky's Zone of Proximal Development. I add to this by combining the ZPD with the concept of boundary objects to explain how the process of developing and analysing scenarios creates triggers for discussion, which can encourage participants to 'enter' their ZPDs. I also draw upon the concept of scaffolding to help explain how facilitators provide assistance that enables participants to engage in PSP processes. In the next section, I explore literature that can help explain how and why learning varies between different participants in PSP.

2.5 What theories can help understand who learns what and why?

As I described in Section 2.2, the inclusion of different stakeholders in PSP does not mean they are included on an equal footing. Participants may not learn to the same extent, or at all. Different participants could also learn about different subjects. It is therefore important to understand how and why learning varies between participants in PSP. In this section, I consider how literature on adult learning, knowledge co-production, and power can help to explain how and why learning varies.

2.5.1 Criteria for encouraging learning

In their seminal book on adult learning, Knowles et al. (2005) present a list of criteria that need to be met for adults to engage in learning. First, they indicate that adults need to know why learning is valuable, in terms of improving their performance as workers and members of society, as well as improving their quality of life. Second, Knowles et al. argue that adult learners tend to be self-directed and self-motivated, which can hinder their willingness to learn, if facilitators of learning processes fail to respect this. Third, they specify that the extent to which adult learners are willing to engage in learning is strongly influenced by their diverse prior experiences. Fourth, they explain that adults are most likely to engage in learning when they can see how it will help them in their everyday lives. Hence, they emphasise that learning processes for adults are most likely to be effective when they are linked to real-life situations.

The above criteria presented by Knowles et al. indicate that adults are most likely to learn when they perceive new knowledge to be relevant, and when they consider learning processes to be sincere and appropriate to their needs and experiences. This corresponds closely with the credibility, salience and legitimacy (CSL) framework, set out by Cash et al. (2003). Cash et al. explain that the likelihood of people accepting new knowledge depends on the extent to which they consider it to be based on valid evidence, relevant to their requirements, and produced through a fair, inclusive and unbiased process. It therefore seems reasonable that PSP participants need to find the knowledge they encounter in their interactions with others to be credible, salient and legitimate for them to enter their ZPD, and thus, to learn.

However, as Hegger et al. (2012) contend, it is vital to acknowledge that different participants have different criteria and thresholds for what they consider CSL. This creates obvious issues, since what is considered credible, salient or legitimate by one actor may be considered invalid, irrelevant or unfair by others. Indeed, Cash et al. (2003) also acknowledge that CSL cannot always be achieved simultaneously for everyone. Moreover, they reason that enhancing one may have adverse effects on another. If the CSL of knowledge encountered in PSP is compromised for one participant, s/he may be less likely to enter their ZPD, and thus less likely to learn. To understand how and why learning varies, it is therefore important to consider the factors that influence what participants consider CSL. I therefore explore which factors might be important to consider, below.

2.5.2 Learning is shaped by social context

The most prominent explanation of how social context influences learning is presented by Bandura (1977) in his seminal work *Social Learning Theory*. Whereas earlier accounts described a unidirectional process, in which individuals learned through encountering social stimuli, Bandura argues that the influence of social context on learning is a reciprocal process. He suggests that the social context shapes what is learned by the individual, and the individual, in turn, shapes the social context, through what s/he has learned. Bandura supposes that learners select, organise and transform the stimuli they receive, in ways that are influenced by their social, economic and political characteristics. This indicates that what participants consider CSL and, thus, what they learn, can be shaped by their social context.

This resonates with the concept of 'knowledge coproduction,' developed in the field of *Science and Technology Studies* by Jasanoff (1996). Jasanoff argues that knowledge and social

organisation are coproduced. This means that the ways people come to know the world, and the ways people come to live in it, are closely entwined, such that a change in one will result in a change in the other. She indicates that, when studying the ways people develop knowledge, it is important to consider how they are shaped by the organisation of society.

The above works of Bandura and Jasanoff emphasise that what is learned by different participants in PSP is shaped by the social context in which it occurs. Considering the social, economic and political characteristics of individual participants, and the relationships between them, can thus help explain variations in what different participants learn. An important aspect of social organisation is the power imbalances that exist between people. I therefore turn to explore how power can influence who learns what.

2.5.3 Learning is shaped by power imbalances

As indicated above, variations in what is learned by different participants can be explained by the extent to which they consider the knowledges they encounter to be CSL. This may be shaped by the social context, in which specific PSP processes take place, including participants' social, economic and political characteristics, and the relationships between them. As I indicated in Section 2.2, different participants may not be included in PSP on an equal footing. It is therefore especially important to consider the power imbalances that exist in the relationships between different participants.

What is power?

Before moving on to consider the insights that studying power imbalances can provide to understanding learning in PSP, it is important to clarify what I mean by power. Literature on theories of power demonstrates it is conceptualised in multiple ways. Indeed, one prominent theorist of power, Lukes (2005) describes power as intrinsically contested. Avelino and Rotmans (2011) therefore argue that, rather than seeking an all-encompassing definition of power, a more important task should be to find a conceptualisation of power that works, in the context that it is used.

Early theories of power, developed by theorists such as Hobbes, Dahl, and Bachrach and Baratz focus on power as sovereignty, held predominantly by the state (Clegg, 1989). These theories were later built on by Lukes, who conceptualises power as an actor's capacity to ensure others consent to domination (Lukes, 2005). He thus indicates that power is exercised through shaping other's beliefs and desires, such that they consider domination to be positive. Power was

later reconceptualised by poststructuralist theorists, including Parsons and Foucault, who critiqued earlier views for their focus on power as prohibitive - in other words, the assumption that empowerment of one person, always leads to the disempowerment of another. Instead, Parsons (1967) argues that power is the capacity of actors to create order, through the use of social norms. The influential philosopher, Foucault (1975), built on this by arguing that power is always present and inescapable, but is not always prohibitive. Rather, he supposes that power is complex and exists alongside resistance. He indicates that power is manifest in actors' knowledge and practice and can be used to achieve strategic effects through disciplining non-conformity.

In this research, what is of most interest is not so much the power of the state in maintaining social order (although this may be implicated in PSP processes), but in the power relations between different participants, and facilitators in PSP. Equally, viewing power as prohibitive and inherently negative is likely to be inaccurate. As I show in Chapter 6, the facilitators of PSP processes have substantial power over what is learned in them, but this does not necessarily mean their use of this power is ill-intentioned. Since PSP is commonly used to take account of the diverse knowledges of different stakeholders, Foucault's view that power is complex and embedded in knowledge is particularly salient, as is the view of Parsons, that power is manifest in the use of social norms to create social order.

Avelino and Rotmans (2011) draw on Parsons and Foucault to build a definition of power they believe is useful for researching processes of multi-stakeholder dialogue for sustainability. They reason that such a definition must be flexible, to allow power to be discussed across different disciplines and epistemologies. Furthermore, they contend that, when thinking about sustainability, conceptualisations of power need to account for the possibility of long-term changes over time, and for diverse power relations between actors. As such, they broadly define power as: "the capacity of actors to mobilise resources to achieve a certain goal," (p.798). They also take a broad definition of 'resources,' as assets, including people, knowledge and money, that help actors to achieve certain goals.

Avelino and Rotmans go on to propose three types of power relations between actors. First, they explain that actors can have power over others. In other words, they can mobilise other people as resources. Second, they reason that some actors can have more power than others. In this case, they have a greater capacity to mobilise resources. Third, they state that different actors can have different kinds of power. For example, some actors have the capacity to create or discover new resources, others have the capacity to uphold the existing distribution of resources, whilst others have the capacity to transform it. They propose that analysing the power relations

between actors, can help explore how innovations that encourage sustainability can interact with existing paradigms. Moreover, they reason that studying power can help understand how actors' social positions influence their relations with others, as well as their ability to make a difference. I use Avelino and Rotmans' conceptualisation of power, and their framework of power relations to engage with the notion of power in this research.

How does power influence learning in PSP?

PSP fits particularly well with the concept of 'spaces of participation,' described by Cornwall (2004). Cornwall describes spaces of participation as spaces that bring together a heterogeneous set of actors, representing different knowledges and interests, to facilitate transformation on a given issue. PSP can be described as a particular type of space of participation, in that it creates spaces where heterogeneous sets of participants, are brought together to construct and deliberate on sets of alternative futures, with the aim of responding to SEPs. Cornwall draws on Lefebvre (1991) and Foucault (1975), to argue that spaces of participation are not neutral, but influenced by how and by whom they are initiated, the expectations and experiences that participants attribute to them, and the social context in which they take place. Cornwall thus supposes that spaces of participation can replicate existing institutions, and associated power imbalances, in wider social and political contexts.

This links to the critical perspective on participation, put forward by Kothari (2001). Kothari contends that participatory processes can, and all too often do, reproduce existing inequalities and serve the interests of actors who are already powerful. Similarly, Kapoor (2005) describes how the power embedded in social contexts can act as a self-surveillance mechanism, whereby participants feel afraid of acting and speaking in ways that contradict the roles expected of them by society or a particular social context.

This is reflected by the adult learning scholar, Brookfield (2005) who provides a detailed analysis of how power can influence learning. Of note is his discussion of power in collaborative, group learning processes. Like Kapoor, he argues that people commonly act in such a way as to conform to what the majority considers to be beneficial. In group learning processes, he indicates this fear of alienation can lead to conformity of participants' ideologies, or 'groupthink.' Brookfield specifies that when groupthink occurs, it typically reflects conformity to the agenda of the strongest voices in the group. Dissenting voices thus become viewed as obstructive to the agenda of the learning process and are subsequently marginalised. This has significant implications for studying learning in PSP. If the voices of dissenting participants are marginalised, the knowledge that participants encounter in their interactions is most likely to be that of the

most powerful voices. This could result in marginalised participants considering the knowledge they encounter to be less CSL, which could reduce their enthusiasm for the process. Moreover, it could limit their ability to enter the ZPD and thus to learn.

Another way that power could influence learning is through the power imbalances between the facilitators of PSP processes and the participants. As I stated in Section 2.2, the influential participation scholar, Cornwall (2008) contends that participatory processes are typically owned and defined by those who initiate them. Similarly, other participation scholars indicate that the outcomes of participatory processes are ultimately defined by the facilitators (Chambers, 1997; Kapoor, 2005). The learning scholar, Tett (2016) explains that facilitators of group learning processes have substantial power over what is learned through them. She describes how facilitators can advocate or oppose the contributions of participants, simply through acknowledging them, asking participants to expand on ideas, and instructing them to write things down. In this way, Tett explains, facilitators effectively signal and ultimately define what is considered acceptable for discussion, and what is not.

However, Cornwall (2004) asserts that spaces of participation can also be opportunities for subversion. She emphasises that participants can always use subtle techniques, like pretending not to hear, or understand, as well as deliberately using the normative roles they are expected to play (for example, feigning compliance), to control the direction of discussions in spaces of participation. It is therefore important to be aware of the complex ways participants can use and subvert power, to make their voices heard, and to contribute their knowledge to PSP processes.

This emphasises the importance of studying the influence of scaffolding, as described in Section 2.4. The above literature indicates that power is manifest in the scaffolding provided by facilitators to participants, and by participants to each other. These interactions shape what is considered an appropriate contribution to discussions, and what is not, which defines whose voices are privileged and whose voices are marginalised. This, in turn, influences the extent to which participants find the knowledge they encounter credible, salient and legitimate, the extent to which they enter their ZPD, and thus, both what they contribute and learn.

2.6 Conceptual Framework for understanding learning in PSP

In this section, I propose a framework for how learning occurs in PSP, centred principally around the concept of the Zone of Proximal Development. I add that PSP stimulates these interactions by creating boundary objects through the process of PSP. I also propose that facilitators provide scaffolding that enables participants to engage effectively in PSP. I conclude this discussion by illustrating how the power imbalances in PSP can help explain how and why learning varies between different participants.

2.6.1 The Zone of Proximal Development

To begin with, I propose that PSP enables participants to 'enter' their zones of proximal development (ZPD), through encouraging interactions between participants, with different fields of expertise. This can result in participants learning from the different knowledges they encounter. For instance, in the example of the smallholder farmer and the climate scientist, described in Section 2.4, the farmer learns from the scientist's knowledge of global atmospheric processes; conversely, the scientist learns from the farmer's knowledge of local rainfall patterns. I illustrate how PSP enables participants to enter their ZPD, and thus learn, in Figure 2.2, below.





Source: author construct

2.6.2 Boundary objects

I reason that these interactions are enabled by processes of developing scenarios in PSP. Different participants contribute their expertise and assumptions about the future, to the development of future events, conditions and trajectories that constitute the scenarios. The process of creating scenarios therefore acts as a type of 'boundary object' that encourages the exchange of knowledge between different participants. This is illustrated in Figure 2.3, below.

Figure 2.3: PSP acts as a boundary object, by prompting discussion between different participants, through the process of creating scenarios.



2.6.3 Scaffolding

I reason that participants are enabled to engage in PSP processes through 'scaffolding,' or assistance, provided by facilitators to participants, and from one participant to another. This assistance acts like the rungs of a ladder, that help participants to engage effectively in PSP processes. It encourages them to contribute their knowledge to the development of scenarios, and to learn from the knowledges of others. I illustrate this in Figure 2.4, below.

Figure 2.4: Assistance provided, by facilitators to participants, and from one participant to another, helps participants to engage effectively in PSP.



2.6.4 Learning is shaped by power imbalances

In Section 2.5, I explored how considering the social context, and associated power imbalances of participants in PSP, could help to explain variations in what is learned by different participants. I reason that 'groupthink' can lead to the promotion of the strongest voices, that conform with the agenda set by those who initiate and facilitate PSP processes. This can lead to dissenting voices becoming marginalised because they are viewed as disruptive or irrelevant to this agenda. This can limit the credibility, salience and legitimacy of the different knowledges participants encounter, which limits their ability to enter their ZPD and learn from the knowledges of others. This is illustrated in Figure 2.5, below.



Figure 2.5: Power imbalances shape who learns what and why

2.7 Conclusion

In this chapter, I began by highlighting the assumption that PSP can include the knowledges of different stakeholders in dialogue about tackling SEPs. I linked this to wider notions that the participation of different stakeholders is needed to help tackle SEPs. However, the socially constructed nature of what sustainable futures should look like, means there will inevitably be winners and losers. It is therefore important to understand the benefits of PSP for those whose knowledges it includes.

I then reviewed literature on scenario planning from corporate strategic planning, Futures Studies, and tackling complex social-ecological problems to explore why PSP is considered beneficial. It appears that PSP is considered most useful as a tool for learning about complex social-ecological problems, but there is limited detailed, and theoretically-informed analysis regarding how learning occurs in these contexts. Equally, there is scarce information and understanding regarding how and why learning varies between different participants.

I therefore used literature on cognitivist learning theories, social learning, power and participation, to develop a conceptual framework that can help build an understanding of learning in PSP. In this framework, I draw on Vygotsky's Zone of Proximal Development (ZPD) to explain that learning occurs through interactions between participants from different fields of expertise. I go further than simply saying that learning occurs through interactions, in that I explain how PSP stimulates these interactions, by encouraging participants to apply their knowledges to the development of future events, conditions and trajectories. In this way, PSP processes act as boundary objects. I also draw on the concept of scaffolding to explain how participants are enabled to engage effectively in PSP processes through assistance provided by facilitators and between participants. Finally, I reason that learning in PSP varies because of power imbalances between participants, and between facilitators and participants. These power imbalances shape the extent to which participants encounter new knowledge that is credible, salient and legitimate, and thus the extent to which they enter their (ZPD) and learn. The individual components of this framework are not, in themselves, ground-breaking, but I have brought them all together and applied them in an innovative way, to the context of PSP. However, to explore how well it explains learning in PSP, it is important to apply it to my empirical study. In the next chapter, I therefore use this conceptual framework to develop a research methodology for this research.

Chapter 3 – Research Methodology

3.1 Introduction

As indicated in Chapters 1 and 2, researchers and practitioners assume that participatory scenario planning (PSP) is useful for including the knowledges of diverse stakeholders in dialogue about responding to social-ecological problems. However, different knowledges are likely to be privileged over others, and the outcomes of PSP processes may be defined by those who initiate them. As such, there is a need for a better understanding of the benefits PSP could have for those whose knowledges it includes. I define PSP as interactive processes, in which diverse groups of people develop alternative narratives of plausible future events, conditions and trajectories. Literature on the benefits of PSP indicates that it could be most useful as a tool for learning, which I define as a change in understanding, as a result of some external stimuli. However, there is limited information regarding how learning occurs, or how and why it varies between different participants. In Chapter 2, I presented an innovative conceptual framework to help explain learning in PSP. In this chapter, I describe the methodology I used to apply this framework in my empirical study of learning in PSP.

In accordance with my conceptual framework, my methodology focused on exploring the interactions that occurred between different participants in PSP, the processes that encouraged them, and the social contexts that shaped them. I therefore took a qualitative approach, underpinned by a social-constructivist epistemology, that sought to understand participants' subjective experiences of PSP processes, and the learning that occurred through them. I expand on this in Section 3.2, below.

3.2 Research Approach

In my conceptual framework, I reasoned that learning occurs through interactions between participants from different fields of expertise, which exposes them to new knowledge. I also emphasised that learning is shaped by the social context in which it occurs. To study learning in PSP, it was therefore important to account for how learning was constructed, through participants' interpretations of the knowledges they encountered in PSP. Lincoln and Guba (2016) indicate that a constructivist methodology is suitable for exploring how people make sense of phenomena, including how they attribute meanings to them. They describe constructivism as a way of exploring the nature of knowledge, as well as the organising principles, and values, of

physical reality and society. They state that a constructivist perspective assumes that people's understanding of reality is not objective but constructed by the meanings and interpretations they attribute to it. Similarly, Bryman (2008) describes constructivism as a valuable way of understanding the roles played by social actors in the creation of knowledge about reality. Critics of constructivism often contend that, if taken to its extreme, viewing reality as entirely socially constructed and subjective is unhelpful. However, in this study, I considered it a useful approach for analysing subjectivity in the ways that learning occurs, and how it varies for different participants. I therefore adopted a constructivist approach in this research.

As described by Bryman, the implication of taking a constructivist perspective is that the ways people understand the world are viewed as social constructs. I therefore acknowledged that the phenomena I observed in the research were mediated by the experience of others, as well as my own frame of understanding. According to Lincoln and Guba, understanding socially constructed phenomena requires consideration of people's narratives of them. I therefore concentrated on exploring participant's narratives of PSP processes, including their expectations, their experiences of PSP processes, their interactions with others, and changes in their understanding of social-ecological problems.

Winchester and Rofe (2010) advise that understanding the processes through which social phenomena are constructed, is typically best approached using qualitative research methods. Quantitative methods were deemed inappropriate for the purposes of this research, because of the importance of capturing the details and subjectivity of individuals' experiences of PSP. I therefore used qualitative methods to explore learning in PSP. I describe how I applied this approach to a research design in Section 3.3, below.

3.3 Research Design

To capture a well-rounded account of learning in PSP, my research was designed around three different sources of data, gathered over 16 months, between August 2015 and December 2016. These were: i) a review of PSP cases in the academic literature, ii) interviews with practitioners of PSP, and iii) two empirical case studies of PSP processes. I describe each of these, and the rationale for selecting them, in Section 3.4, but first, I designate how each of these sources helped to gather information to address each of the research questions.

The case review focused primarily on the first research question, regarding the expected and reported benefits of PSP. It thus provided the basis of my research and directed my focus towards learning as the most commonly anticipated and reported benefit of PSP. The cases in the

review also provided some information that helped to investigate how learning occurred, as well as how and why it varied. The practitioner interviews contributed to all three research questions. They built on the case review by exploring practitioners' views on the expected benefits of PSP, and then investigating how learning had occurred, and how, in their experiences, it had varied. Similarly, the case studies contributed to all three research questions, but provided especially rich data regarding the details of how learning occurred, and how and why it varied in two specific PSP workshops. The connections between the different sources of data are illustrated in Table 3.1, below.

Research Questions	Information sought	Sources addressing Research Questions (in order of priority)
What are the expected and reported benefits of participatory scenario planning?	Rationale for using PSP; Motivations for facilitating and/or participating in PSP; worldviews, social-economic backgrounds, disciplinary perspectives of facilitators and participants; Objectives of the project in which	Case review of PSP processes reported on in academic literature; Semi-structured interviews with practitioners of PSP; Survey of participant expectations in case studies; Observation of PSP workshops in case
How does learning occur in participatory scenario planning?	PSP is used Learning indicated by changes in understanding; Elements of PSP that led to learning; Interactions between participants from different fields of expertise;	studies Semi-structured interviews with participants in case studies; Observation of PSP workshops in case studies; Semi-structured interviews with facilitators in case studies;
	Role of facilitation in enabling engagement with PSP	Semi-structured interviews with practitioners of PSP; Case review of PSP processes reported on in academic literature
How and why does learning vary between different participants in participatory scenario planning processes?	Content of learning by different participants; Content and nature of interactions in and around PSP workshops; Social context in which PSP occurs; Composition of participants in PSP	Semi-structured interviews with participants in case studies; Observation of PSP workshops in case studies; Semi-structured interviews with practitioners of PSP; Case review of PSP processes reported on in academic literature

3.4 Research Methods

3.4.1 Case Review

As indicated in Chapter 1, PSP is frequently used by researchers, as part of wider research projects that aim to inform responses to SEPs. There exists, therefore, a substantial body of peerreviewed, academic literature reporting on individual cases in which PSP has been used. However, reviews that assess the expected and reported benefits of PSP across these cases are scarce. I therefore interrogated and analysed the peer-reviewed, academic literature that reported on cases in which PSP was used specifically to help tackle SEPs. This predominantly helped to address my first research question, regarding the expected and reported benefits of PSP, and to highlight weaknesses in the existing evidence base for the benefits of PSP. The case review also yielded insights into how learning occurred and how and why it varied. The data from this review subsequently formed the basis on which the rest of the research was built.

The scarcity of previous reviews on the purported benefits of PSP, meant there was not a recognised method for me to follow. I therefore developed a suitable method for use in this research. My method differs from that taken in the subsequent review of 23 cases of PSP by Oteros-Rozas et al. (2015). Oteros-Rozas et al. review PSP from the perspective of practitioners who were, themselves, involved in the cases they interrogate. I conducted my review from the perspective of an independent outsider. As such, I was not influenced by an interest in portraying PSP as beneficial or successful in achieving its purported outcomes.

I began by defining criteria for the inclusion of cases, which I then used to identify and select relevant texts. As Haddaway et al. (2015) suggest, I ensured my selection of cases was rigorous by selecting literature from multiple databases, ensuring I was consistent in my selection and analysis of literature, and including a critical appraisal of the literature. I conducted a literature search using the databases: 'Web of Knowledge', 'Google Scholar' and the University of Reading's 'Summon' discovery service. Initially, I used the broad search term: 'scenario planning case studies.' This proved inefficient, however, as it yielded many results describing scenario planning in contexts like corporate strategic planning, business, technology and health, that were not relevant to this research. I therefore decided to narrow the search terms, to focus the results on cases in which PSP was used to help confront social-ecological problems. I therefore refined the search terms to: 'scenario planning environmental management'.

Once I had selected an initial set of cases that met these search criteria, I used a snowball sampling technique to further populate the review. This involved browsing the titles and abstracts of papers cited in each of the cases, and of papers listed on 'Web of Science.' I selected titles that used words and phrases related to the search terms, rather than only including titles that matched the search terms exactly. For example, a title along the lines of 'identifying strategies for poverty reduction under climate change using future scenarios,' would be included, because the terms 'poverty reduction,' climate change,' and 'future scenarios,' are related to 'social-ecological systems,' 'sustainable development,' and 'scenario planning.' Articles were excluded if they did not show any relevance to the search terms. For example, a title along the lines of 'using scenario planning to map the long-term marketing strategy of a major corporation,' would be excluded, because although 'scenario planning' is mentioned, it is not in the context of confronting social-ecological problems.

By taking this approach, an initial total of 53 cases were included in the review. After some critical reflection, these were narrowed to 44, with nine of the cases excluded for several reasons. In some cases, examples of PSP were only mentioned in passing, as part of more general discussions on confronting social-ecological problems and did not provide enough detail for an analysis of their expected and reported benefits. For example, Twyman et al. (2011) refer to several examples of PSP in their discussion of managing 'dryland agro-ecological systems,' but do not provide a detailed description of any of them. In other cases, examples of PSP were mentioned only briefly, as part of wider projects. Additionally, several cases were described in multiple papers by the same groups of authors. In these instances, the information from the papers was synthesised to create just one case.

I subsequently conducted a critical appraisal of the papers to improve the transparency of the sample selection for the case review. I created a set of criteria that were focused on ensuring information-rich cases. Hence, I sought to include papers that provided clear descriptions of the objectives, rationale, benefits and challenges of PSP. I considered a description to be clear, if it provided explicit, detailed information on these aspects of PSP. Specifically, my criteria were:

- 1. Clear objective(s) for the project in which PSP was used.
- 2. Clear rationale for the use of PSP in the project.
- 3. Clear description of the role played by PSP in meeting the objectives.
- 4. Clear description of the purported benefits of PSP.
- 5. Clear description of the method used for conducting PSP.
- 6. Reporting any challenges that were encountered in the use of PSP.

Criteria 1 – 4 were considered vital for the study and, as such, any case that did not meet these four criteria was excluded from the review. Cases that did not meet Criteria 5 and 6 were kept in the review, but I made a note of their deficiencies. Following the critical appraisal, the sample of reviewed cases was narrowed to a set of 30 cases. These cases generally contained rich information on the PSP processes they described. This enabled a detailed analysis of what I considered to be a robust sample, of detailed and convincing literature on the expected and reported benefits of PSP.

However, I acknowledge that, because the final set of cases were selected based on their richness of information, there was an inherent bias towards well-reported cases, and they are not necessarily representative of all PSP processes. In the interest of fairness, it is also important to acknowledge that information in the excluded cases may have been absent because of editorial constraints, as explained by Baxter and Eyles (1997). The cases I excluded may have just been superficially reported, rather than poor examples of PSP. In the practitioner interviews, I therefore sampled practitioners from one of the cases that was excluded from the review, as well as those that were included, to give them an opportunity to provide more detailed insights. I discuss this further in Section 3.4.2, below.

The final sample of 30 cases included the use of PSP in a wide range of different topics including: energy, ecosystem services, community adaptation to climate change, food security, fisheries, forestry and biodiversity conservation. All the cases in the review date from the year 2000 onwards, since this was the period in which they were most frequently identified. Academic literature was the only form of literature included in the review, as there was insufficient time to conduct a thorough analysis of grey, and other, forms of literature in this research. I acknowledge that, although substantial, the review could not practically include every single PSP process that has been used to help confront SEPs. I also recognise that the papers in the case review were written by academics, who had been directly involved in the cases they reported on. It was therefore important to elicit detailed primary data on learning, and to analyse it from the perspective of an independent outsider.

3.4.2 Practitioner Interviews

In accordance with my conceptual framework, it was particularly important to find out about the interactions between participants from different fields of expertise, their engagement in PSP processes, and how participants were assisted to engage in them by facilitators and other participants. It was equally important to find out about the relationships and power imbalances between different participants and facilitators involved in PSP. Conducting in-depth interviews with people, can be a powerful way of understanding such detailed social interactions (Yeo et al., 2016). I therefore conducted semi-structured interviews with practitioners of PSP, since as indicated in Chapter 2, the people who initiate and facilitate participatory processes, typically have substantial control over what goes on in them and what the outcomes are. I considered a practitioner to be anyone with experience of coordinating and/or facilitating PSP processes, that aimed to help tackle SEPs. This included a mix of academic researchers, who had used PSP as part of research and knowledge exchange projects, as well as professional facilitators. I found that the distinction between researchers and facilitators was often blurred, in that academic researchers often acted as facilitators in processes they coordinated, and facilitators often acted as researchers.

I adopted a purposive sampling approach to select practitioners of PSP in social-ecological contexts. I began by contacting the authors of papers that were included in the case review. I also contacted authors of cases that were excluded from the review, to give them an opportunity to explain the cases they described, in more detail. This was so I could find out more about the expected and reported benefits of the PSP processes, in which they were involved, and their understanding of how learning occurs, and who learns what. I also selected some prominent figures in the field of PSP, by identifying the authors of highly-cited papers on social-ecological PSP, and through suggestions from my doctoral supervisors. Additionally, I selected two facilitators from the Food Security Futures case study, and one from the Positive Futures for Southern Africa case, who all had wide-ranging experience of PSP outside their respective case study. Finally, I selected one practitioner who I met at a futures-oriented conference, as she had substantial experience of facilitating PSP, and other futures-thinking tools, in social-ecological contexts. I contacted each of the practitioners by email and asked if they were willing to engage in a conversation about specific cases they had worked on, and their thoughts and experiences on PSP more generally.

In total, I interviewed 16 practitioners. Of these, 14 practitioners were academic researchers in fields including ecology and ecosystem services (n=5), geography (n=2), sustainable

energy (n=1), interdisciplinarity (n=1), sustainable development (n=1), food systems (n=3), climate change adaptation (n=1). The remaining two practitioners were both professional facilitators. Nine practitioners were from the cases in the case review, one was from a case that was excluded from the review, three were prominent figures in PSP discourse, and three were from my case studies. Only one of the practitioners I contacted actively turned down the opportunity to be interviewed, because of ill health, but nine others simply did not respond when I emailed them.

Most of the practitioner interviews took place by Skype or telephone, because the practitioners were located around the world. However, four of the interviews took place face-to-face, as the practitioners were available locally and suggested we meet in person. The interviews were designed to last around one hour, and most lasted between 50 and 75 minutes. To avoid taking too much of practitioners' time, I checked how much time they had available at the start of each interview and then again after about 45-50 minutes. All the practitioners I interviewed appeared happy to speak to me, were open to being questioned, and provided detailed and eloquent responses. The practitioner interviews thus yielded rich information about the rationale for using PSP, the benefits it was believed to have produced, and ideas about how learning occurred, what learning entailed and who learned what from whom.

The practitioner interviews were semi-structured to achieve both breadth of coverage and depth of information regarding the key topics of interest in the research (Yeo et al., 2016). As Yeo et al. suggest, to achieve this breadth and depth, I kept interview questions open-ended to encourage in-depth discussions, as well as avoiding leading participants to answer in certain ways. I also used responsive follow-up, or 'probing' questions to expand on, as well as explain interviewees' initial responses. Prior to commencing the practitioner interviews, I prepared onepage lists of discussion topics. These were designed to ensure that I covered the breadth of topics, and the depth of information required to address my research questions. I also sent the list of discussion topics to practitioners before their interview, so they could approach it with a clear understanding of my research and the subjects I wished to discuss. Additionally, I prepared more detailed interview guides, for myself, which contained a list of open-ended questions and discussion prompts designed to facilitate in-depth discussion of the topics needed to address my research questions. However, as Bryman (2008) advises, the interview guides were not definitive, and I remained flexible and responsive to following-up on leads, interesting points, and inconsistent responses. In practice, therefore, the precise questions and order of discussion topics sometimes diverged from the interview guide. An example of this interview guide can be found in Appendix 1.

I asked practitioners about a variety of topics. Firstly, I asked them to describe their experiences of PSP, including why they thought it was beneficial, or not, in the contexts in which it was used. This was to find out about what they considered the expected and reported benefits of PSP to be. Secondly, in Chapter 2, I reasoned that the agendas set by the people who initiate and facilitate PSP processes have a substantial influence on what is learned by different participants. I therefore asked about practitioners' professional and disciplinary backgrounds, and their reasons for using PSP, to help understand how they set their agendas, and what philosophies underpinned them. Thirdly, I asked them to describe cases in which they thought learning had occurred, focusing on how they explained it, and what had been learned by whom. Finally, to explore challenges and uncertainties in PSP, and ways it may be improved, I asked practitioners about the questions they thought needed to be addressed, and any changes they would like to see made to PSP practice.

3.4.3 Case Studies

The case review and the practitioner interviews provided useful information, but they still relied on post-rationalised accounts of PSP processes, by the people who were involved in them. It was therefore important to observe, first-hand, the interactions that took place in specific PSP processes, how they were encouraged, and how assistance provided by facilitators, and between participants, enabled learning to occur. It was also vital to elicit the experiences of participants in PSP, rather than relying solely on the accounts of practitioners.

Lewis (2003) suggests that where such in-depth, contextual information is required, case study research is a useful approach. Bryman (2008) describes a case study as an in-depth analysis of an apparently unique case. I therefore conducted two case studies of specific PSP workshops. I selected these case studies, based on the criteria that they: 1) met my definition of PSP as the development of alternative narratives of plausible future events, conditions and trajectories; 2) were participatory and included a range of different stakeholders; 3) encouraged knowledge exchange between different participants; and 4) focused on tackling SEPs. I found several PSP processes, globally, that met these criteria and were ongoing during the period of study. I selected two of these, which were the easiest to access, because of an existing contact held by one of my doctoral supervisors. Both workshops were part of wider projects that aimed to help tackle socialecological problems. The first was a PSP workshop in Tanzania. This was part of the wider 'Food Security Futures' (FSF) project, which explored threats and opportunities for achieving food security under climate change. The second was a workshop in South Africa. This was part of

'Positive Futures for Southern Africa' (PFSA), an initiative that aimed to develop hopeful and innovative, but realistic ways of thinking about future relationships between human and environmental systems.

I chose to use two case studies so that a comparison could be made between them. Lewis (2003) advises that where multiple case studies are used, some degree of difference between them is inevitable. It was therefore important to decide upon and manage a level of consistency between them, to ensure they were comparable. PSP processes vary widely, in terms of their objectives, the way they are integrated into wider research and practical agendas, and the people who initiate, facilitate and participate in them. This makes it difficult to make any generalisable claims about learning in PSP. As such, I carefully selected the two case studies, such that there was a balance of similarities and differences between them. These similarities and differences are delineated in Table 3.2, below.

	Food Security Futures	Positive Futures for Southern Africa	
Similarities	Both complied with my definition of PSP in Section 3.1. Although their approaches		
	differed, they still followed essentially the same logic – using present signals, trends		
	and drivers to develop storylines of alternative futures.		
	Participatory process		
	Included a range of different worldviews, social-economic backgrounds, a disciplinary perspectives.		
	Actively encouraged knowledge exchange between different participants		
	Facilitators ranged in expertise and experience		
Differences	Exploratory approach to assess the	Normative approach to explore what	
	implications of plausible futures on	participants' preferred futures would look	
	food and nutrition security in	like and how they might be realised.	
	Tanzania.		
	'Matrix' Approach, including	'Manoa' approach with addition of 'Futures	
	'backcasting'	Wheels' and '3 Horizons' methods	
	Scenarios developed from existing	Scenarios developed from 'weak signals' of	
	drivers of change	the future in the present, in the form of 'seeds'	
	National-level focus	Regional-level focus	
	Aimed at informing policy	Aimed at learning about how 'positive	
		futures might look, and how they may be reached	
	Seemingly haphazard organisation	Organised strategically, well in advance	
	Downscaled CCAFS 'East Africa' scenarios	Development of completely new scenarios	
	Very little funding	Substantial funding	
	Participants commuted to venue on both days	Participants stayed at the venue for the 3.5- day duration of the workshop	

Table 3.2 Similarities and differences between the Food Security Futures and Positive

Futures for Southern Africa case studies

problem – food and nutrition Anthropocene more broad security	cal problems in the dly
---	----------------------------

Source: author construct

Food Security Futures

The first case study was a PSP workshop held as part of the 'Food Security Futures (FSF)' project, which explores threats and opportunities for food and nutrition security (FNS) at a European and global level. The workshop took place in Tanzania, as part of Food Security Futures' global-level interests. It focused on challenges and opportunities for ensuring FNS for school children in Dar es Salaam. The FSF workshop took an exploratory approach, based on a motivation to explore plausible scenarios and assess the consequences of different future possibilities for FNS in Tanzania. The organisers intended to have a direct influence on policy, through reviewing how Tanzania's second Five Year Development Plan (FYDP) might meet the future challenges of FNS under different scenarios. They therefore brought together 20 participants from a mixture of government institutions (n=10), NGOs (n = 2), education (n = 4), academia (n = 3), and the private sector (n = 1), all of whom were Tanzanian and worked and lived in Dar es Salaam. The schedule of the workshop, including descriptions of the specific activities within it are described in Table 3.3 below.



Table 3.3: Food Security Futures PSP workshop process

Activities	Description
(Day 1) Introduction	Facilitators gave a presentation explaining workshop objectives and PSP. They subsequently explained each activity before it was undertaken. Participants were split into three groups, each exploring a different theme: i) issues directly affecting FNS, ii) capacity-building for FNS, and iii) cross-cutting themes. Participants selected their own groups but were all asked to mix the groups up halfway through Day 1.
Visioning	Participants constructed a vision of what they would ideally like the future to look like. These visions were based on specific policy objectives from the Tanzanian government's new FNS Policy. Each participant wrote their individual visions on post-it notes then presented their ideas to the group. The groups then discussed and decided on a collective group vision.
Backcasting	Participants developed concrete steps, backwards, from the point of achieving the vision in the future to the very first step that needed to be taken in the present to work towards it. The backcasts were presented as timelines of post-it notes on a horizontal piece of flip-chart paper.
Transcription of backcasts by facilitators	The visions and steps developed in the backcasts were typed-up, as they appeared on the post-it notes. The notes were then elaborated to develop coherent, detailed narratives.
Downscaling CCAFS 'East Africa' scenarios	The groups were each allocated one 'East Africa' scenario, which had been previously developed by the research program, CCAFS. Participants read their scenario and imagined what events, conditions and trajectories would be like in Tanzania in that scenario. Individuals wrote their ideas on post-it notes then presented them to their group for discussion. Each group then collectively developed a narrative of how Tanzania would look in their scenario.
(Day 2) Modelling comparison	The lead facilitator presented pre-developed quantitative models that corresponded to each of the 'East Africa' scenarios. Each group then compared their downscaled scenario to the model.
Causal mapping	The groups each decided on the three most important drivers in each scenario and then identified five other drivers that had the most influence on them. All the drivers were subsequently linked by drawing single-direction arrows, marked as positive or negative, between the post-it notes, according to the influence each driver had on another.
Scenario-based review of backcasts	Each group received all three plans developed in the backcasts and imagined they were trying to implement them in their group's scenario. The scenarios acted as parameters, within which the participants had to work out how to overcome challenges and successfully implement the plans.
Recommendations	The groups discussed and decided on what recommendations they would make for implementing each plan under their scenario. These recommendations were presented by each group in a plenary session at the end of the workshop. There was insufficient time for participants to give feedback on the recommendations produced by other groups. Instead, participants were asked to comment on the recommendations when they had been typed-up and distributed by the facilitators, but this did not happen until three months later.



Positive Futures for Southern Africa

The second case study was a workshop in South Africa, organised as part of the 'Positive Futures for Southern Africa' (PFSA) project. PFSA part of is a global project, premised on the idea that human changes to the Earth are unsustainable, and have inadvertently induced a new geological epoch – the Anthropocene. However, there is a limited sense of how desirable, plausible alternatives might look. The aim of PFSA is therefore to develop narratives exploring how alternative, socially just and environmentally sustainable, or 'positive futures' might look, and how social-ecological systems may be transformed to create 'positive futures.'

According to the organisers of the workshop, the PFSA project is underpinned by a philosophy of change, that assumes transformations in SES often begin in 'shadow networks,' 'niches,' or 'transformation arenas,' which are peripheral to conventional spaces of decision-making. Hence PFSA assumes it is important to support the dispersal of transformations out of these peripheral spaces and into mainstream discourse and decision-making. As such, PFSA has been identifying and collecting a database of small-scale initiatives, or 'seeds' that could be described as supporting transformations to positive futures.

The PSP workshop in South Africa used these seeds to develop alternative narratives describing how a positive future might look for southern Africa, and how the seeds might be supported to bring these positive futures into being. In doing so, the PFSA workshop took a more normative approach than the FSF case study, in that it explored what participants' preferred futures would look like, and how they might be realised. However, although the scenarios were normative, they were built on logical reasoning to ensure their plausibility. The workshop brought together a diverse range of 23 participants, including a mixture of academics, practitioners and artists. It was facilitated by four academics and one professional facilitator. It took place over 3.5 days in a hotel in Cape Town. The schedule of activities in the workshop are outlined in Table 3.4, below.



Table 3.4: Positive Futures for Southern Africa workshop process

Activities	Description	
(Introductory afternoon) Introductory presentation	Facilitators gave a presentation introducing the participants to the concept of the Anthropocene, and the objectives of the workshop.	
'PRACTIS' (Platform for Research in Art, Culture, & Theory in Society) session	This involved using art to stimulate discussions about complexity, subjectivity, futures, and transformations to develop a 'shared language' for discussions in the rest of the workshop. The PRACTIS session was facilitated by one of the participants, who developed PRACTIS as a tool for stimulating thought about art, culture theory and society	
(Day 1) Introduction to process	Participants were introduced to the method. The facilitators provided a general overview of the process, before explaining the first activity in detail, as they did for each subsequent activity. The participants were then split into three groups of six, and one group of five. The four groups each went into one of four 'break-out' rooms, where they engaged in the activities. Each group was allocated three diverse 'seeds,' including two seeds based in Southern Africa, and one global-level seed related to technology, which they used to build their scenarios.	
'Futures Wheels'	The groups imagined what the seeds would look like as 'mature conditions,' or mainstream societal norms, in the future. They described these by writing 'mature conditions' on post-it notes. The participants then arranged the post-it notes in concentric circles, or 'wheels'. The central circle represented the direct 'primary' impacts the 'mature' seeds would have. The second circle represented the 'secondary' impacts, or the impacts of the primary impacts. The outer circle represented the tertiary impacts, or the impacts of the secondary impacts. For example, if I spilled tea on my laptop, the primary impact would be the laptop breaking, the secondary impact would be that I had to buy a new laptop, and the tertiary impact would be that I would have less money to spend on other things.	
Cross-impact matrices	Each group explored how the seeds would affect each other. For example, one group explored how encrypted currency would affect a small, self-sufficient community. These connections were plotted on a table and connected by drawing lines between them.	
Scenario skeletons	Each group created a general overview of what their scenario would look like, including general themes for the events, conditions, and trajectories that would constitute it. Each group then presented their 'scenario skeleton' to the entire workshop using a headline statement, three imagined statistics and an artistic expression.	
(Day 2) 'Deep dive' into scenarios	Each group built on their scenario skeleton by adding detail to conditions, events and trajectories.	
'3 Horizons'	Groups were asked to imagine three trajectories, or 'horizons'. 'Horizon 1' represented the dominant way things are in the present. 'Horizon 3' represented the way things are in the future scenarios, which could become dominant in the future. 'Horizon 2' represented the way things will be during the transition from Horizon 1 to Horizon 3. The groups imagined conditions, events and trajectories that defined the decline of Horizon 1, the rise of Horizon 3 and the transition between them.	
Scenario presentations	Each group prepared and then gave a presentation that illustrated their scenario to the entire workshop.	
Plenary discussion	After each group had presented their scenario, the participants and facilitators reconvened in plenary to discuss commonalities between the scenarios, what insights the scenarios provided into how a good Anthropocene might look, and what steps could be taken to create one.	



Pre-workshop questionnaires

As I showed in Chapter 2, the literature on power and participation highlighted the importance of considering the social context in which PSP occurs for understanding how and why learning varied for different participants. In the case studies, it was therefore important to find out about the participants, their worldviews, occupations, social and economic backgrounds, and their expectations and motivations for attending the workshops. To elicit this information, I asked participants in both case studies to complete a short questionnaire prior to the workshop. This consisted of five open-ended questions, concerning: participants' occupations, motivations for attending the workshop, and the benefits they expected it to have. The full questionnaire is included in Appendix 2. The questions were designed to be open-ended, to accommodate the variety of expectations participants may have had, rather than predefining the choice of answers available to participants.

In the FSF case study, I was unable to ask participants to fill in the questionnaire until they arrived on the first day of the workshop. This was because logistical constraints meant that most participants were invited at very short notice. However, because participants had to commute to the venue, and heavy traffic meant a lot of them arrived late, most participants did not have time to complete the survey. As such, the data collected from the survey was minimal. In the PFSA case, I distributed the survey five days prior to the workshop. This was so that participants had sufficient time to complete the survey, but also so it was close enough to the start of the workshop that it was on their minds and they would be more inclined to reflect on their expectations and motivations for attending it. In total, 10 of the 23 PFSA participants completed and returned the survey. Additionally, one participant indicated a preference for discussing his expectations and motivations in person. I therefore interviewed him at the start of the workshop, using the questions from the survey. The questionnaires thus provided useful contextual information, regarding who the participants were and what their expectations of the process were.

Observation of PSP Workshops

In my conceptual framework, I stated that learning in PSP occurs through interactions between participants from different fields of expertise. I conceptualised processes of creating scenarios as boundary objects, that facilitate the exchange of knowledge between different participants. I also reasoned that participants are enabled to engage in these processes through

assistance from facilitators, as well as other participants. It was therefore important to observe the interactions that occurred in the case studies, how they were encouraged by the process of creating scenarios, and how assistance from facilitators and other participants enabled participants to engage in them. It was also necessary to observe the interactions between different participants, especially how the power imbalances between them, and the 'scaffolding,' provided by facilitators, and by participants to each other, shaped what was learned.

McNaughton Nicholls et al. (2016) postulate that ethnographic observation is a useful method for gaining insights into social interactions, that can complement and triangulate the insights provided by verbal accounts. In the case studies, I therefore conducted an ethnographic observation of the workshops as they took place. As McNaughton Nicholls et al. state, observing anything and everything is both impossible and undesirable. They indicate that careful decisions thus need to be made, to focus on specific aspects of an activity. I therefore compiled observation guides before each of the workshops, to help focus on the aspects of PSP that were identified as important in the conceptual framework. These included the extent and type of interactions that occurred during the workshops, the role that developing and analysing the scenarios played in encouraging and shaping interactions, and how participants were encouraged to engage effectively in PSP, through assistance by facilitators and other participants. Detailed copies of the observation guides can be found in Appendix 3. The observation guides were not definitive, however, and, as advised by McNaughton Nicholls et al., I continually reflected on which phenomena I was focusing on, and why I found them interesting.

My observations began with communicating with the workshop organisers and facilitators (in both cases the organisers doubled-up as the facilitators), via email and Skype. This continued through face-to-face meetings prior to each workshop, and debriefing with them during and after the workshops. During these meetings I noted and asked questions about their preparations for the workshops, their aims and objectives, their method of doing PSP, and the participants they selected. After each workshop, I also asked them to reflect on how they facilitated the workshop and what they thought the outcomes had been.

At the start of both workshops the facilitators gave me chance to introduce myself, my research, and my intention to observe the workshop. Thereafter, I moved around the workshop space, watching, listening and speaking to participants and facilitators. I made detailed, written notes on the method, facilitation and interactions in the workshops. In both cases, it was impossible to observe every single participant and every single group constantly. I therefore had to split my time between different groups of participants. I moved from group to group and sat

with each group for between 10 and 20 minutes, listening to and watching their discussions. I tried to be as discrete as possible to avoid diverting participants' attention away from the workshop activities. During breaks in the discussions, I also asked participants to tell me a little about what they were discussing, at what stage they were in the process, and what they were finding easy/hard, interesting/boring etc. I also spoke to as many participants as possible during breaks, mealtimes, and informal settings outside the workshop space, and made notes afterwards.

Case Study Participant Interviews

To calibrate my observations with the experiences of participants, it was paramount to elicit their own accounts of the interactions that unfolded in the workshops, what they learned and how they accounted for this learning. As described in Section 3.4.2, in-depth interviews can be a powerful way of understanding detailed social interactions (Yeo et al., 2016). I therefore conducted semi-structured interviews with participants from the two case studies.

I sampled interviewees purposively to explore the accounts of participants with different worldviews, social and economic backgrounds, and disciplinary perspectives. To select interviewees, I initially contacted all the participants by email, and asked them to meet for conversation to reflect on the workshops. I then interviewed the participants who responded to my initial email, before reviewing which worldviews, social and economic backgrounds, and disciplinary perspectives I had captured, and which were still needed. I subsequently sent a second email to participants whose perspectives were still needed, to encourage them to meet me.

In the FSF case study I interviewed 13 of the 22 participants including 4 teachers, 2 academics (in development studies and economics), 2 junior government officials, 3 senior government officials, and 1 businessperson. Out of the 13 interviewees, 6 were male and 7 female, and the age range spanned from young to middle-aged adults. In the Positive Futures for Southern Africa case, I interviewed 13 of the 23 participants. Of these, 5 were practitioners, 6 were academics, 1 worked for an intergovernmental project and 1 was an artist. Of these, 8 interviewees were female and 5 male, and their ages ranged from young to middle-aged adults.

These interviews focused on several aspects of PSP. The first thing was to ascertain whether learning had occurred through the workshops. As stated in Chapter 2, I define learning as a change in understanding as a result of some external stimuli. I elicited this information directly

by asking participants if their understanding of social-ecological problems had changed through the workshops. Furthermore, most participants alluded to learning, unprompted, when I asked about other aspects of the workshops, including what they found most interesting and challenging, and about their interactions with others. The second focus was to find out about interactions between different participants, the role (if any) that these interactions played in enabling learning, and the activities that stimulated these interactions. I therefore asked participants about which aspects of the workshops, and which specific activities they attributed learning to.

The third focus was to explore how and why learning varied between different participants. I therefore asked participants to describe what they had learned through the workshops, what they had found particularly interesting, and why, how comfortable they had felt engaging in PSP, and interacting with other participants, how well the workshop had met their expectations, and anything they would have liked to have discussed in the workshop, but were not able to. I also explored variations in what was learned, by comparing the learning described by different participants, and considering whose voices and agendas this reflected. To explore this in the interviews, I asked about who participants interacted with most during the workshop and the discussions that they learned from. The information from these interviews was subsequently compared with my observations of participants' interactions in the workshops.

As I indicated in the conceptual framework, the literature on scaffolding and participation highlighted the important role of facilitation, for enabling and shaping learning. This meant it was also essential to interview the facilitators of the workshops, to explore how they assisted participants to engage in PSP, as well as what they thought participants had learned. Specifically, I asked them to describe specific instances where they had assisted participants to engage in PSP, how they had done so, and how well they thought it worked. I also asked about whether they were aware of theories and concepts like scaffolding. In the same way as in the practitioner interviews, I compiled a list of discussion topics that were sent to all the interviewees before the interviews, and a more detailed interview guide to help me focus the discussions on relevant topics. A copy of this can be found in Appendix 4.

3.5 Analysis

In accordance with the constructivist epistemology taken in this research, it was important to take an emic approach, that built knowledge from the perspectives of my research subjects. I therefore conducted my analysis in an inductive way, using subjects' observations and interpretations of their experiences to build knowledge from the bottom-up (Ormston et al., 2016). Admittedly, in the analysis of interview and observational data, I used a loose topic structure, to help focus on the broad aspects of PSP that were deemed important in the conceptual framework: interactions, creation of boundary objects, scaffolding, and power imbalances. In this sense, my analysis had a deductive element, but I allowed knowledge to emerge from the accounts of my participants regarding the details of interactions, boundary objects, scaffolding, and power imbalances, and remained open to considering issues that were not highlighted by the conceptual framework.

To build an understanding of learning in PSP from participants' observations and interpretations of their experiences, it was necessary to take an approach to analysis that focused on eliciting and interpreting meaning in their accounts of PSP. I therefore used a thematic analysis, which Spencer et al. (2016) describe as exploring and interpreting themes and patterns in what the data says. This involved reading, thoroughly, through the data and identifying themes in what was reported by the authors of papers in the case review, and what was said and done by the participants in my practitioner interviews and case studies. These themes were progressively combined and recombined to create higher-order themes to help address the research questions.

In the analysis for the case review, I began by thoroughly reading and re-reading the papers that described each of the cases. I thus compiled a detailed set of notes on each case. I organised these notes, using a template I devised, that structured them under the subheadings of 'objectives,' 'expected benefits,' 'methodological approach,' 'reported benefits' and 'challenges faced'. I then arranged the notes in a table to compare the cases with each other. Specifically, I compared their objectives, expected benefits, methodological approaches, reported benefits and challenges. I read and re-read the notes in the table multiple times and thus identified themes in the data, based on the frequency with which different objectives, expected and reported benefits, methodologies, and challenges appeared and how much prominence they were given in each respective paper.

The data from the semi-structured interviews were recorded on a voice recorder to enable me to concentrate on interacting with the participants and providing prompts and followup questions, rather than taking detailed notes of everything the interviewees said. I did take

some brief notes, however, to help keep track of what participants were saying and make it easier to think of follow-up questions. These were used to write brief 'interview summaries' of each interview, immediately after they had taken place. I transcribed the audio recordings as soon as possible after each interview had taken place, so the interview was still fresh in my mind. To save time, five of the practitioner interviews were transcribed by a professional transcriber, shortly after each interview. I subsequently read through these transcripts and calibrated them with the notes I took in the interviews. In the PFSA case study, the responses to the qualitative survey were all received via email and typed-up. In the FSF case, I typed up the few questionnaires that were filled in. The data from the workshop observations were recorded as written notes in research diaries for each case study. These notes were subsequently typed up as soon as possible after the workshops, while the observations were still fresh in my mind.

Once the interview, survey and observational data had been transcribed, I uploaded them into the qualitative analysis software, 'NVivo 11' (QSR, 2015). In NVivo, I initially read through the transcripts, and the interview summaries, multiple times, to refresh my memory on what the main topics of conversation were. Then I read through each transcript in more detail and created 'nodes' - or extracts from the data that provided information, relevant to the research questions. Initially these nodes were quite descriptive, and just showed what happened in participants' experiences of PSP, whether learning had occurred, and what the content of learning was. I then read through the nodes and categorised them into themes emerging from the data. For example, in the PFSA case study, I identified a theme in that participants described how they had been able to think 'outside the box.' All the nodes that provided evidence for this theme were thus organised in NVivo under the heading 'thinking outside the box'.

To create more analytical themes, I read through each of the themes and identified relationships between them. Specifically, this involved creating 'node matrices' in NVivo that compared different themes to identify where the same extracts of data were present in multiple different themes. I also created notes in Word documents that detailed how the different themes connected and influenced each other. For example, as will be explained in Chapter 5, I identified that in the Positive Futures for Southern Africa case study some participants attributed their ability to push beyond their usual range of thinking to the 'Futures Wheels' activity. I therefore noted that there was a relationship between the Futures Wheels activity and 'thinking outside the box.'

Reliability and Validity

As Lewis et al. (2016) describe, the concept of reliability of research findings is sometimes dismissed by qualitative researchers. Such critiques dismiss reliability because, since qualitative research typically focuses on understanding the complexity and contextualised nature of phenomena, it is naïve and irrelevant to think that findings could be replicated. I agree with Lewis et al. however, who contend that reliability is important in qualitative research to ensure the robustness of research findings outside their study sample.

Lewis et al. state that the reliability of research findings depends on the likelihood that key features in the data could recur in different samples, and that they have been analysed in a rigorous way. In this research, my dataset spanned across a wide range of different instances in which PSP was used, thanks to the inclusion of data from the carefully selected 30 cases in the case review, and the accounts of the 16 practitioners I interviewed. The fact my findings were gleaned from such a range of cases indicates that they would be likely to recur, if another set of PSP processes had been studied. To ensure that I analysed my data rigorously, I created research summaries immediately after each interview and observation, so what was said was still fresh in my mind. I listened carefully to the interview recordings, multiple times, to ensure that what I transcribed was accurate. During the thematic analysis, I read through the data and summarised my findings then read through the data again to check whether my initial interpretations were true to the data. I repeated this process several times. As suggested by Silverman (2014), I validated my findings through triangulation. This involved calibrating the results from my three data sources, the case review, practitioner interviews and case studies, against each other and identifying overlaps and discrepancies between them. I did this by summarising the contribution of each source to each of the research questions in a table. Furthermore, I linked the data to my wider reading of PSP literature, as well as theory on learning and power.

3.6 Positionality and issues faced

In accordance with the constructivist philosophy underpinning this research, it was important to consider not only the subjectivity in how participants constructed their experiences of PSP, but also the influence of my own positionality on participants' accounts of PSP. As suggested by Watson (2011), I prepared for my research by developing an understanding of the social and political contexts, in which the research was conducted, and my position in relation to

them. Specifically, I considered how my position as a young, white, male researcher could have influenced the accounts of my research subjects.

In the case review, my position was that of an academic researcher scrutinising the work of other academics. It was therefore important to ensure I represented the papers and their authors in a way that upheld acceptable standards of rigour. As described in Section 3.4, I maintained a high level of transparency and consistency, to ensure I conducted the review in a fair and robust way.

In the practitioner interviews, the participants were a mixture of academics and professional facilitators, from a range of social and cultural backgrounds. They all shared a vested interest in PSP, as their careers and reputations were reliant on it, to a greater or lesser extent. As a researcher providing an outsider's perspective on PSP, there was a distinct possibility I could indicate that PSP does not always deliver on some of its putative benefits. To protect these interests, practitioners may have shaped their responses to downplay, or avoid the weaknesses and challenges of using PSP.

To mitigate this issue, although Wiles et al. (2006) suggests interviewees could be given the opportunity to read interview transcripts, and remove anything that could cause harm, I considered this option inappropriate for maintaining an independent perspective. Instead, I clearly informed practitioners of the intentions of my research, and that I reserved the right to criticise PSP and its practice. I also made it clear that my research had the potential to suggest ways of improving the benefits of PSP to emphasise I was not just looking for its weaknesses. Given these clarifications, the practitioners I interviewed generally spoke openly and eloquently about both the benefits and drawbacks of using PSP.

The FSF case study in Tanzania, and the PFSA case study in South Africa, were both conducted within a context of historical exploitation by Europeans. As a white, British male I may have therefore been associated with social, political and economic privilege. In both case studies, participants may have also assumed I worked for the project - Food Security Futures or Positive Futures for Southern Africa - that coordinated the workshop. As a result, the research participants may have felt uncomfortable speaking openly to me about their experiences of the workshops. Indeed, in the FSF case study, only two interviewees voiced any strong criticism of the workshop. To ameliorate these issues, I clarified that I did not work for FSF or PFSA. Moreover, I stated that I was studying the workshops, as an independent outsider, for the purposes of my own doctoral research. I stated this during the workshops, in emails to participants, asking to arrange
interviews, at the start of interviews, and sometimes during the interviews, to reassure interviewees that they could speak their mind.

In the FSF case study, participants only spoke English as their second or third language, and I spoke hardly any Swahili. To deal with the language barrier I offered to arrange a translator prior to each interview, but the interviewees all declined this offer. I therefore made sure I spoke slowly and clearly, and gave them chance to seek clarification, to help them understand my questions. For example, in one interview the interviewee explained that he understood written English, better than spoken English. At his request, I therefore wrote out each question after I had asked it verbally. These measures generally worked well, as most interviewees provided detailed and eloquent responses to my questions.

In terms of the observations, as McNaughton Nicholls et al. (2016) point out, the researcher's presence often influences the very thing they are trying to observe. To avoid influencing the discussions, and associated learning, through my own contributions, I did not directly participate in the discussions in either of the workshops. However, in both workshops I joined participants in the breaks and at mealtimes. In the PFSA workshop, I also participated in the introductory 'PRACTIS' session (see Table 3.5). However, participants and facilitators reflected afterwards that this made me an 'affable' and 'non-threatening' presence, which helped them feel more comfortable with sharing their experiences, as well as their learning from the workshop, with me.

In both case studies, the research participants were aware that I spoke to other participants and facilitators, and that their responses would contribute to my thesis and subsequent publications, which could be read by others. There was therefore some danger of reluctance to discuss the interactions and relationships that unfolded with other participants and facilitators. To help participants feel comfortable discussing these issues, I reminded them they did not have to discuss anything they did not want to, that their responses were confidential and that they would not be named in the thesis, or any subsequent publications. Having been given these assurances, most participants did subsequently speak in detail about their interactions with other participants.

In general, I approached the research with humility and attempted to build rapport with the research participants. I asked interviewees to suggest a convenient time and comfortable location for interviews to take place. I gave participants the opportunity to ask me any questions for clarification throughout the research. In face-to-face interviews, I offered to buy participants

food and soft drinks out of respect for them taking the time to speak to me. I also checked how much time they had available to avoid taking up too much of it. Out of thanks to the workshop organisers for helping with my research, in both cases, I met with them at the end of the case study to brief them on my initial findings. The workshop organisers will receive a summary of my thesis, once complete.

As in all research, my research design had limitations, as well as biases. These are detailed in Table 3.5, below.

Limitations of the research	Implications for my	Measures taken to minimise
design	findings	the impact of limitations
Insufficient time to explore whether learning led to changes in decision-making and action.	Lack of understanding regarding the potential medium and long-term benefits of learning in PSP.	Asked case study participants if they believed what they had learned could influence their professional roles, and everyday lives. Asked practitioners if learning, through PSP, had resulted in subsequent changes in decision-making and action.
Definition of learning is based on my own interpretation, rather than that of participants	Case study participants could have had a different understanding of what 'learning' means.	Asked interviewees, specifically, about how their understanding of the SEPs covered in the workshops had changed.
Lack of information from grey literature, as well as PSP practitioners from governmental and non-governmental organisations.	My data is biased towards examples of PSP being used by academics.	Interviewed practitioners who also had experience conducting PSP processes for governmental and non- governmental organisations.
Papers in the case review were selected according to criteria that focused on their information- richness.	My data is biased towards well-reported cases of PSP.	Sought to interview authors of cases that were excluded from the review. Unfortunately, only succeeded in interviewing one such practitioner.
Focused on developing an in- depth understanding of learning in PSP, rather than comparing the benefits of other participatory processes for stimulating dialogue on tackling SEPs.	My research does not contribute to understanding the strengths and weaknesses of PSP as compared to different methods.	In Chapter 7, I consider how my contribution of a more detailed, theoretically- grounded understanding of learning in PSP, can inform future research on learning in other participatory processes.

Table 3.5 – Limitations of my research design, implications for my findings, and the measures I took to minimise these issues.

Source: author construct

3.7 Research Ethics

I carefully considered and attempted to manage the ethical issues in this research. I aimed to not only mitigate potential negative impacts of the research, but to encourage opportunities for a net positive impact. I made every effort to ensure I conducted the research in an ethically sensitive and responsible way that accounted for my moral responsibilities to research subjects, gatekeepers, the wider communities in which the research took place, and the academic profession. My research was conducted with ethical clearance from the Research Ethics Committee at the School of Agriculture, Policy and Development, University of Reading.

Participants in the practitioner interviews and case studies were thus provided with an information sheet, clearly explaining the purpose, intent and process of the research. This was also explained verbally, prior to observations and interviews, to ensure it was understandable to the research participants, as well as to give them chance to ask for clarification. Participants were informed they had the right to request that any of their responses be excluded from recording and analysis, and to withdraw from the research at any point. Participants were subsequently asked to sign a consent form, or to provide recorded, verbal consent if they preferred. I ensured the confidentiality of participants' responses by attributing quotes to pseudonyms, rather than participants' real names. Electronic data was stored on a password–protected laptop, and backed up to the James Hutton Institute server, as well as my external hard drive. Data was stored and managed in accordance with the University of Reading's Data Protection Policy and the UK Data Protection Act.

3.8 Conclusion

In this chapter, I have outlined the methodology I used for studying learning in PSP. The research was underpinned by a constructivist philosophy and took a qualitative approach to explore the intricacies of interactions between different participants in PSP. I collected data from three sources: i) a case review of academic literature, describing cases of PSP, ii) semi-structured interviews with practitioners of PSP, and iii) two case studies of specific PSP processes, which included observations, semi-structured interviews, and qualitative surveys. Data were analysed using a thematic analysis, and triangulated across the different sources of data. I took every effort to ensure the research was conducted in an ethical way, and was sensitive to how both my positionality, as well as that of individual participants, could have influenced their accounts of PSP. The results I present in the following chapters are based on my detailed analysis of data, acquired through this methodology. In the next chapter, I begin the presentation of my results by

showing the expected and reported benefits of PSP, informed primarily by the case review, but also by the practitioner interviews and case studies.

Chapter 4 – What are the expected and reported benefits of Participatory Scenario Planning?

4.1 Introduction

In this chapter, I present my findings that show the expected and reported benefits of using participatory scenario planning (PSP). As defined in Chapter 1, PSP refers to interactive processes in which diverse groups of participants develop alternative narratives of plausible future events, conditions and trajectories. In Chapter 2, I showed that PSP has increasingly been used to include the knowledges of different actors, in dialogue that aims to help tackle socialecological problems (SEPs) and thus create more sustainable conditions in social-ecological systems (SES). However, there is a need to understand the benefits PSP may have for those whose knowledge it includes. In this chapter, I explore the expected and reported benefits of PSP, drawing on insights from the case review, interviews with practitioners of PSP, and my two case studies, as described in Chapter 3.

This chapter is structured as follows: in Section 4.2, I show that practitioners' use of PSP is frequently underpinned by an assumption that structuring future possibilities into sets of alternative scenarios, could make it easier for participants to think about complexity and uncertainty in SES. This is not frequently reported as a benefit of PSP, because it is subsumed under more specific benefits, particularly learning, as well as developing and testing strategies for tackling SEPs. However, I also show that PSP does not always deliver on this benefit, because of a limited capacity to deal with surprise. In Section 4.3, I show that learning is the most frequently expected and reported benefit of PSP. However, I indicate that there is a lack of theoretically-informed understanding regarding how learning occurs, and limited consideration of how and why it varies for different participants. Then in Section 4.4, I demonstrate that PSP is also expected and reported to help develop strategies to tackle SEPs. However, there was a lack of concrete evidence for this leading to any change in practice, at least during or soon after PSP processes that could help tackle SEPs. In Section 4.5, I show that practitioners struggle to articulate the strengths and weaknesses of PSP, especially in comparison to other methods, and have limited opportunities to conduct and publish formal evaluations of the impacts of PSP.

4.2 *"It helps to break it down, step-by-step."* – Structuring future possibilities

4.2.1 Expected Benefits

As stated by Peterson et al. (2003), the world is highly complex, and the future holds infinite possibilities, making it highly uncertain. PSP is, they claim, a potentially useful method for thinking about such complex and uncertain futures. My data shows that the rationale for using PSP was frequently underpinned by just such an assumption. Complexity and uncertainty, although distinct, were often difficult to disentangle in people's accounts of the benefits they expected from PSP. This could be because of an assumption that complexity is a source of uncertainty. As Cilliers et al. (2013) describe, complexity is a characteristic of a system that arises because of dynamic interactions between a large number of its components. They state that it is impossible to understand a complex system in its entirety, because of the high volume of interacting components, and the variable nature of those interactions. This means that knowledge of complex systems is always limited. According to Rotmans and van Asselt (1999) variability of phenomena, as well as limitedness of knowledge about them, contribute to uncertainty. Complexity and uncertainty are therefore distinct yet closely linked phenomena.

My research revealed an assumption, by practitioners of PSP, that 'exploring,' 'clustering,' or 'structuring' future possibilities, through PSP, could help participants to think about complexity and uncertainty. This was most plainly evident in the case review, in which the authors of all 30 cases expected PSP to help participants think about complexity, by structuring future possibilities into a small set of alternative narratives. For example, Tschakert et al. (2014) expected that *'exploring a range of possible futures,'* could help participants think about: *'complex feedbacks across scales and between the social and the natural,'* (p.1052). In the case of Brand et al. (2013) the authors begin their paper by presenting PSP as a method that can: *'take into account the complexity of human-environment systems,'* (p.43). They go on to describe how it can: *'widen perspectives in the light of a variety of possible futures,'* (p.45).

In the case described by Vermeulen et al. (2013), regarding the use of PSP for food security under climate change, the authors do not explicitly mention complexity. Rather, they state that they expected processes, in which: *'uncertainties were structured to produce socio-economic scenarios,'* would help participants to *'explore unknown unknowns,'* and *'challenge assumptions about the future,'* (p.8360). In this instance, although complexity is not mentioned, the authors introduce their paper by describing the issue of achieving food security under climate

change as a complex problem, which requires consideration of future uncertainties. Indeed, they cite Stirling (2010), who emphasises that the complexity of the world demands consideration of uncertainty. This indicates an implicit assumption by the authors that structuring future possibilities could help account for the complexity of achieving food security under climate change, and explore the uncertainties surrounding it.

In the practitioner interviews, eight of the 16 informants explicitly indicated that they thought structuring future possibilities in PSP, could help participants think about complexity and uncertainty. One practitioner, Deborah, had extensive experience of using PSP in global and regional-level ecosystem assessments, including the prolific Millennium Ecosystem Assessment (MA), which made her insights particularly salient. She explained that:

"The planet is so vast, that it requires synthesis of a tremendous amount of information to understand any global trend, or pattern, or process... [in the MA] we felt that the scenarios would help make the global something we could think about more easily." (Deborah, 2015)

This indicates an assumption that PSP could help participants to comprehend the immense complexity of global processes. When I subsequently asked her how she thought PSP could do this, Deborah responded:

"The number of ways the future could unfold is infinite... In a scenarios process, one is forced to cluster the possible futures into a manageable number of stories. [In the MA] we came up with four scenarios that we used to organise ideas about the future." (Deborah, 2015)

It is evident then, that structuring future possibilities into sets of stories was expected to help participants to think about complexity and explore uncertainties.

My observations and interviews, in the two case studies, also indicate that structuring future possibilities can be a key motivation for using PSP. However, this was not always immediately obvious. For example, in the Food Security Futures (FSF) workshop I studied in Tanzania, which was part of a wider research project exploring threats and opportunities for food and nutrition security (FNS), there was evidence of an assumption PSP could help participants comprehend complexity, by structuring different possibilities into alternative scenarios. In the 'Description of Work' that FSF were required to submit to their funders, they detailed their rationale and planned activities. This document states that one of the main objectives of the overall project was to: *'capture the multidimensionality of FNS and the complexity of its drivers,'*

but it does not explicitly explain how PSP is expected to do this. However, in the introduction to the workshop in Tanzania, I observed one of the facilitators, Mike, describe PSP as a *"structured approach,"* to thinking about the future of FNS. He went on to explain that creating scenarios would enable the participants to: *"explore different possibilities."* From this, I infer that structuring different possibilities in PSP was expected to help think about the complexity of drivers in FNS.

In the Positive Futures for Southern Africa (PFSA) case study, the assumption that structuring future possibilities could help think about complexity and uncertainty was not explicitly obvious, but it was implicit in the facilitators' expectations. At the start of the workshop, the facilitators gave an introductory presentation, in which they emphasised the complexity of SES and the resultant uncertainty of the future. One of the facilitators, Jane, stated that *"multiple 'positive futures' could exist simultaneously,"* and then stated an interest in exploring and comparing what a 'positive,' or just and sustainable, Anthropocene meant, for different stakeholders. This demonstrates a recognition of uncertainty regarding what a positive future would look like, because of complexity in what it would mean for different people. I thus infer that structuring future possibilities in PSP was expected to help explore complexity and uncertainty, in what positive futures would be like.

The above evidence reveals an expectation that, by structuring future possibilities into sets of alternative narratives, PSP could make it easier for participants to think about complexity in SES and uncertainty. This assumption is not new. One of the early protagonists of scenario planning in corporate strategic planning, Pierre Wack, presents it as a useful tool for structuring uncertainty (Wack, 1985). Wack explains that it helps to break down the infinite array of future conditions into coherent narratives of alternative, plausible futures. More recently, in their review of scenario planning techniques, Amer et al. (2013) reason that structuring uncertainty in this way can help participants understand different drivers, as well as their potential future trajectories. These assumptions are reflected in the expectations shown in my data, that PSP structures the possible futures of SEPs into sets of alternative narratives, thus making it easier for participants to explore the interconnections between different components of SES. I explore whether these expectations were realised in Section 4.2.2, below.

4.2.2 Reported benefits

The above evidence shows that PSP is commonly used with the expectation that structuring future possibilities can help participants to think about the complexity and future uncertainty of SEPs. In my practitioner interviews, as well as my two case studies, this was only rarely reported as a benefit by my informants. For example, one practitioner, Barry, indicated that structuring future possibilities is inherently beneficial because it helps participants cope with uncertainty:

"People just go away learning how to think about the future... I think that's probably the biggest learning... At the end they are able to address uncertainty." (Barry, 2016)

In the FSF case study, one participant, Fiona, clearly articulated how structuring future possibilities had helped her to think about complexity by highlighting 'steps,' or links between different components of SES. She described how she had found it useful to:

"start with the big thing then break it down step-by-step," because "after you give one aspect, it develops another step to another aspect." (Fiona, 2016)

In the case review, though, none of the authors reported this as a benefit of PSP. I infer that this is because the benefits of structuring future possibilities are subsumed under more specific benefits. As I show in Sections 4.3 and 4.4, the most common benefits associated with PSP are: learning about SES and developing strategies for tackling SEPs. For example, in the case review, the case described by Tschakert et al. (2014) reportedly enhanced: *'recognition of interdependencies between environmental, socioeconomic, and governance processes.'* This indicates that structuring future possibilities helped participants learn about the interactions between different components in SES. In a different example, the practitioner, Nigel, described how structuring future possibilities had enabled researchers and participants to:

> "Run project proposals and policy options through the scenarios to figure out which project option is the most robust." (Nigel, 2015)

Thus, he felt that they had been empowered to make informed decisions in highly complex and uncertain situations. It is evident then, that, although structuring future possibilities was not often reported as a benefit, it was related to, and sometimes acted as a precursor to other reported benefits, particularly learning, and developing and testing strategies.

4.2.3 Limitations of PSP for structuring future possibilities

My findings indicate that PSP is useful, because it structures future possibilities into a manageable set of alternative scenarios, which helps participants to think about complexity and uncertainty, and can lead to learning and developing and testing strategies to help tackle SEPs. However, there was some evidence that PSP did not always deliver on this benefit. For instance, in the case described by Brand et al. (2013), the authors reflected that PSP was unable to adequately account for the complexity of the problems they were considering. They described how they, quite reasonably, limited the number of components they included in the scenarios to 20, but thought this did not fully account for the complexity of SEPs. Similarly, Rivard and Reay (2012) and Vermeulen et al. (2013) agree that the ability of PSP to consider the full complexity SES is limited. For example, Rivard and Reay state that PSP *'may not emphasise every parameter*,' (2012: p.373).

These limitations were also raised by the highly experienced practitioner, Deborah. She asserted that structuring future possibilities in PSP has a limited capacity to deal with surprise, even though, as shown in Section 4.2.1, she thought it could help participants think about complexity and uncertainty. She explained:

"So, the world is full of totally off-the-wall events, that nobody would ever foresee, but end up having substantial influence on how things unfold. But if you put them in a narrative, they seem incoherent, they seem made-up, and so scenarios are held to a standard of coherence that the world is not." (Deborah, 2015)

This was reflected in a discussion with one of the Food Security Futures facilitators, Mike, who contended:

"Often scenarios [processes] claim to incorporate some discontinuity, but really, they do not, because they are just continuations of underlying trends, however weak those trends are." (Mike, 2016)

This indicates a deficiency in the capacity of PSP to help participants think about complexity and uncertainty, because it based on the continuation of current trends.

It is not surprising however, that PSP is unable to incorporate every possible eventuality and every interacting component in SES. As Cilliers et al. (2013) contend, such a feat would be impossible. They state that a person can only ever learn more about a complex system, and can never understand it, in its entirety. Similarly, Rotmans and van Asselt (1999) state that uncertainty

can never be eradicated, because knowledge is always limited. One can only ever deal with uncertainty by learning about its sources. The importance of learning was reflected in my research, in which I found that it was the most frequently expected and reported benefit of PSP. I discuss this further in Section 4.3, below.

4.3 "I learnt a lot... because my understanding was completely different." - Learning

4.3.1 Expected benefits

As described in Chapter 2, PSP is commonly used to integrate the knowledges of different stakeholders, in dialogue on tackling SEPs (Oteros-Rozas et al., 2015). Many proponents of PSP, in my research, thus expected participant's exposure to different knowledges to encourage learning. As defined in Chapter 2, I consider learning to be a change in understanding as a result of some external stimuli. In 21 of the 30 cases in the case review, the use of PSP was underpinned by an expectation that bringing together actors, with different roles, interests, knowledges and worldviews, would encourage learning. As illustrated in Table 4.1, assumptions related to learning included 'understanding complexity in SES,' 'understanding links between ecosystem changes and human wellbeing,' 'understanding the impacts of local-level changes at the national level,' and 'incorporating information on environmental change from Indigenous and scientific knowledge.'



Table 4.1 – Evidence that learning was an expected benefit in 21 of the cases in the case review, based on my interrogation of the expectations

in these cases in the peer-reviewed literature on specific cases of PSP.

Case of PSP in peer-reviewed literature Evidence of learning as an expected benefit of PSP	
Bohensky et al. (2006) - ecosystem services in Southern Africa	Expected to assist participants in understanding complexity in social-ecological systems.
Brand et al. (2013) - ecosystem services in the Swiss Alps.	Expected to develop a better understanding of current problems, as well as to generate more robust social knowledge.
Carpenter et al. (2006) the Millennium Ecosystem Assessment.	The MA scenarios aimed to gain a better understanding of ecosystem services in the future, the effects of changes in human systems on them, and the effects of changes in ecosystem services on human wellbeing.
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem service analysis in Tanzania.	Aimed to build an understanding of the links between ecosystem changes, and human wellbeing.
Malinga et al. (2013) - ecosystem service assessment in South Africa.	Incorporating a range of perspectives, through the participation of a range of stakeholders, to help understand complex changes in ecosystem services.
Mistry et al. (2014) describe a scenario planning process on ecosystem management in Guyana.	Aimed to foster 'social learning' on the implications of local-level developments, including indigenous populations, on the national scale and of national and international developments at the local scale.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	Incorporating the knowledges of diverse stakeholders to analyse changes in ecosystem services, as well as associated human wellbeing.
Plieninger et al. (2013) - ecosystem services provided by cultural landscapes in Germany.	Assumed PSP could help identify and mitigate threats to cultural landscapes.
Henly-Shepard et al. (2015) - improving adaptive capacity to hazards in Hawaii.	Assumed that PSP could facilitate 'social learning,' to encourage participants to anticipate, as well as adapt to potential hazards.
Ravera et al. (2011a); and Ravera et al. (2011b) - adaptation in Nicaragua.	Aimed to bring out different participant's visions of the future, to understand the effects of climate variability and socio-economic changes on livelihoods in the future.
Ravera et al. (2011a); and Reed et al. (2013) - environmental management and adaptation in UK uplands.	Explore different stakeholder's visions and desires for the future, to help them understand connections between different variables and evaluate complex indicators.
Tschakert et al. (2014) – climate change adaptation in Ghana and Tanzania.	Create spaces for 'co-learning,' in which participants could gain insights about the past, present and future, by combining knowledge from everyday experience with climate science. Hence, develop a more anticipatory understanding of SEPs.

Wesche and Armitage (2014) - environmental change in northern Canada.	To incorporate information on environmental change from both indigenous and scientific knowledge, and thus develop an understanding of community vulnerability to environmental change, as well as potential adaptation options.
Van Berkel et al. (2011) - rural development in Portugal.	Facilitate a deeper investigation of rural development options, by enriching understanding of local constraints and assets.
Vermeulen et al. (2013); and Vervoort et al. (2013) – food security under climate change.	Identify uncertainties, challenge existing assumptions about the future.
Badjeck and Diop (2011) - fisheries research in West Africa.	Help develop understanding of how different situations of climate change would affect the lives of the communities.
Bohensky et al. (2009) - options for ecotourism in Papua New Guinea.	Integrate different types of knowledge into alternative futures, to explore how different participants react to change, and develop understanding across spatial scales.
Palomo et al. (2011) - protected area management in Spain.	Build consensus between conflicting stakeholders, by co-developing new knowledge about the ecosystem.
Pert et al. (2010) - biodiversity conservation in Queensland, Australia.	Help understand drivers of change and evaluate the impacts of current trajectories.
Mora et al. (2014) - forestry in France.	Combine anticipatory learning on the impacts of climate change with land use planning.
Wollenberg et al. (2000) – 'adaptive co- management' of community forests.	Facilitate 'social learning' regarding system structures, long-term trends and potential uncertainties, by highlighting participant's assumptions and mental habits, through exposure to potential future trajectories.

Source: author construct



Table 4.1 thus shows a common assumption that, by bringing different stakeholders into dialogue, PSP could result in learning.

In one, specific example, Mistry et al. (2014) attribute the complex nature of SEPs to the different interests at stake: 'part of this complexity arises from different interest groups,' (p.127), before arguing that: 'effective and more equitable management of social–ecological systems, therefore, requires analysis at multiple levels of governance, their inter-connections and the competing values and perspectives across scales,' (p.127). They subsequently present PSP as a useful tool for conducting a: 'cross-scale, multiple perspective assessment of emerging social–ecological challenges,' (p.128). This demonstrates an expectation that PSP could bring together these 'multiple perspectives,' to help learn about the complex values and perspectives of interest groups at different levels of governance.

In the practitioner interviews, 11 of the 16 informants also assumed that bringing different stakeholders into dialogue would enable learning about the complex and uncertain dynamics of SEPs. For example, Mary described a PSP process she conducted on upland conservation, which required input from both natural and social science research. She indicated that PSP was expected to bring these different disciplinary perspectives together:

"The approach that we took to scenario planning involved a lot of natural science research to understand some of the mechanisms behind the carbon dynamics, but we then combined that with social science research to try and understand the broader issues that participants were concerned about and the broader system." (Mary, 2015)

This indicates an expectation that, by combining different disciplinary perspectives, PSP would encourage learning about different aspects of upland ecosystems. Another practitioner, Greg described how he had assumed PSP could help connect different sources of knowledge on environmental change:

"Science is not the only type of knowledge, and indigenous knowledge is very important to incorporate when we're thinking about how to adapt to environmental change... [PSP] just seemed like a very effective tool to be able to incorporate different sources of knowledge." (Greg, 2016)

It appears then that PSP was expected to enable learning about different aspects of environmental change, by bringing together different types of knowledge.

In the FSF case study, the expectations were not entirely clear. In the 'Description of Work,' document, described in Section 4.2, learning is not explicitly put forward as an expected benefit. However, when I interviewed one of the facilitators, Thomas, he indicated that he expected the workshop to yield: *"information for stakeholders that will inform the planning process of the government."* Similarly, during the introduction to the workshop, Thomas explained that the aim of the workshop was to: *"build an understanding,"* of FNS in Tanzania. This indicates an assumption that learning would be a benefit of PSP. It was difficult to establish what the FSF participants expected the benefits to be, as only one of the 13 participants I interviewed, Tristan, had a prior awareness of PSP. Tristan described how he thought PSP would help learning:

"I thought it would add up on what I already [knew] because there is new knowledge every day, so somehow, someway I could learn something new." (Tristan, 2016)

Learning was a particularly strong expectation in the PFSA case study. This was reflected in the aims set by the workshop organisers, which included learning about what a just and sustainable future might look like from the viewpoints of different actors. From the early stages of planning the workshop, the organisers viewed PSP as a tool for learning. In an email exchange, six months before the workshop, they explained that they conceptualised PSP as: *'a tool for generating awareness, with a view to building understanding,'* and as a *'learning exercise.'* When I attended their final planning meeting before the workshop, they described how they expected that bringing together scientists, artists, policy-makers and practitioners would enable the participants to learn from one another about how positive futures might be created. I also observed that they repeatedly emphasised this focus on learning and were keen to record participants' learning.

This was reflected by the expectations of the PFSA participants. In my pre-workshop questionnaire, all 10 of the respondents indicated they expected to learn, through their participation in the workshop. For example, Dillan indicated that he expected to learn about how participants from different fields thought in different ways:

"I am hoping to make new connections and to try and understand how thinking in their fields intersects with mine." (Dillan, 2016)

Similarly, Penelope expected to learn about how different participants thought about the future:

"I hope to learn from diverse approaches to thinking about the future and to be able to work together to envision futures that go beyond the incremental." (Penelope, 2016)

It is evident then, that practitioners and participants commonly expected PSP to encourage learning by bringing different stakeholders together, to deliberate on the futures of SEPs. I investigate whether PSP met these expectations in Section 4.3.2, below.

4.3.2 Reported Benefits

In the case review, learning was reported in 23 of the 30 cases. For example, Mistry et al. (2014) report that PSP helped local community members to improve their understanding of potential future 'processes of change and adaptation,' (p.143). They illustrate this with a quote from one of their participants, which states they could 'see the interconnection of one another's visions,' (p.143). This shows that PSP met the authors' expectations, described in Section 4.3.1, that it would help participants learn about interconnections between the perspectives of different stakeholders. Similarly in their paper, Plieninger et al. (2013) report that PSP successfully enabled scientific knowledge to be combined with local knowledge, which: 'provided rich insights, both for the researchers and the local actors, into local views and perceptions,' (p.50). The authors report that these insights improved the ability of local actors to: 'successfully deal with landscape change and the resulting impacts on human quality of life,' (p.50) and that the participants: 'felt better prepared for future developments,' (p.50). This indicates that, through bringing different actors into dialogue, local actors learned about different knowledges, to cope with landscape changes and deal with the uncertainty of potential future developments. As illustrated in Table 4.2, similar descriptions of learning were widely reported, across the case review.



Table 4.2 - Evidence that learning was reported as a benefit in 23 of the cases in the case review, based on my interrogation of the outcomes of

these cases in peer-reviewed literature, on specific cases of PSP.

Case Study using Scenario Planning Evidence of learning as a reported benefit of PSP	
Rivard and Reay (2012) - exploring the future of Malawi's energy sector.	Helped to identify key drivers and compounding factors in Malawi's future energy needs
Bohensky et al. (2006) - ecosystem services in Southern Africa	Participants were enabled to think outside their usual range of decision-making.
Brand et al. (2013) - ecosystem services in the Swiss Alps.	Helped stakeholders to 'broaden' their perspectives and helped to build consensus between them.
Carpenter et al. (2006) the Millenium Ecosystem Assessment.	Highlighted key analytical challenges, particularly the strong feedbacks between changes in human systems and changes in ecosystem services
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem service analysis in Tanzania.	Provided insights into the impact of ecosystem service change on human wellbeing
Malinga et al. (2013) - ecosystem service assessment in South Africa.	Helped identify trends, uncertainties and threats, as well as opportunities for communities to shape their own future.
Mistry et al. (2014) describe a scenario planning process on ecosystem management in Guyana.	Helped to reveal differences in perceptions of management schemes at different levels.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	Encouraged 'social learning'
Plieninger et al. (2013) - ecosystem services provided by cultural landscapes in Germany.	Comparison of general scientific knowledge and local contextual knowledge yielded in-depth insights into the perspectives held by different participants.
Gidley et al. (2009) - climate change adaptation in Australia.	Highlighted gaps in knowledge and encouraged 'social learning.'
Henly-Shepard et al. (2015) - improving adaptive capacity to hazards in Hawaii.	Facilitated 'social learning.'
Ravera et al. (2011a); and Ravera et al. (2011b) - adaptation in Nicaragua.	Researchers claimed to have developed a more in-depth understanding of drivers and uncertainties at different scales. Learning for participants is not reported.

Tschakert et al. (2014) – climate change adaptation in Ghana and Tanzania.	Co-learning was facilitated between scientists and local stakeholders, about factors beyond their usual field of understanding. Enabled interdependencies to be identified between different aspects of SEPs.
Wesche and Armitage (2014) - environmental change in northern Canada.	'Shared outcomes' were reportedly produced, through the incorporation of different knowledges, including 'anticipatory thinking,' in-depth understanding of the local environment, and awareness of different perspectives on the future.
Sheppard et al. (2011) - engaging local level stakeholders in climate change action and awareness.	Participants reported that their understanding of climate change and its impacts on the local area had substantially increased.
Van Berkel et al. (2011) - rural development in Portugal.	The process helped to raise and evaluate participants' assumptions and contradictions.
Vermeulen et al. (2013); and Vervoort et al. (2013) – food security under climate change.	Participants reported that they could take a more integrated, systems perspective, better understand challenges and develop responses to them, as well as identify new regional linkages and opportunities for collaboration.
Badjeck and Diop (2011) - fisheries research in West Africa.	Participants developed a broad, anticipatory understanding of the fisheries system.
Bohensky et al. (2009) - options for ecotourism in Papua New Guinea.	Participants were enabled to identify drivers they had previously not considered.
Palomo et al. (2011) - protected area management in Spain.	Participants considered themselves better able to understand trade-offs in environmental management.
Pert et al. (2010) - biodiversity conservation in Queensland, Australia.	Objectives regarding learning were, reportedly, met, but there is limited detail on what this actually means.
Mora et al. (2014) - forestry in France.	Enabled stakeholders to consider the non-forestry benefits of the forest.
Sandker et al. (2007) - protecting forests in Indonesia.	Generated ideas, encouraged 'shared understanding'

Source: author construct



Interestingly, although most of the cases in Table 4.2, were the same as those that expected learning (see Table 4.1), for some, learning was not explicitly anticipated. For example, Sheppard et al. (2011) reported that participants developed an increased understanding of the impacts climate change could have on their local area. On the other hand, cases like Ravera et al. (2011a); and Ravera et al. (2011b) indicated they expected learning to occur but did not report it as an outcome. They report what the researchers learned, as a research outcome, but not what, if anything, was learned by participants. It may be, therefore, that these cases did not consider participants' learning was to be a priority, at least for reporting in published works. The case of Sheppard et al. implies that learning was an unexpected side-effect, whilst the case of Ravera et al. either indicates that learning did not occur, for participants, or that there was little mandate for reporting it in their two papers. This emphasises the importance of considering who learns what in PSP processes, but as I show in Section 4.3.3, this appears to receive little attention.

In the practitioner interviews, all 16 of the practitioners I interviewed reported that PSP encouraged learning. Learning included: identifying interdependencies between different actions, variables, drivers and actors, new ways of thinking, and recognising the different roles, interests and worldviews of different actors involved in SEPs. For example, the practitioner, Greg, stated:

"It was a really effective way of bringing in scientific data on climate change, for example, and then also bringing in indigenous knowledge about how the environment is already changing and what the impacts are on people in the community." (Greg, 2016)

This shows that PSP fulfilled his expectation, described in Section 4.3.1, that it would help connect different types of knowledge about environmental change. He went on to indicate that participants in the PSP had learned to:

"Conceptualise the future in a different way, and their role in that future... They learned about scientific information that people in the community may not have been aware of." (Greg, 2016)

Similarly, another practitioner, Gavin, described the benefits of a PSP process that brought scientific researchers into contact with indigenous communities, which enabled the participants to learn about drivers, at global and regional levels. He described:

"[They found out about] what's happening at a global and broad regional scale, in terms of climate change and economic social cultural change, bio security, all sorts of things. And they said that really opened their eyes... It really forced them to think about these big drivers and surprises that could happen that might be just on their doorstep." (Gavin, 2016)

In the FSF case study, 10 of my 13 informants also showed evidence of learning, especially about the interconnections between different components of food systems. For example, when I asked one participant, Tracy, a senior teacher with limited experience of thinking systematically about SEPs, what she had learned from the workshop, she stated she had identified new drivers, as well as interdependencies between them. For instance, she indicated that she had identified a link between technology and food poverty:

"Therefore, we [identified] the points that cause Tanzania to have food poverty, while we have got resources. For instance, there is poor management and we have got low science and technology. The technology is still primitive. It is primitive!" (Tracy, 2016)

Similarly, when I interviewed another participant, Alan, who worked in social welfare, he also indicated he had learned about the links between different components of food systems, specifically between food production and nutritional value:

> "The food composition, the food size, the food quantity, all these should be considered alike to ensure that you have quality, nutritional food." (Alan, 2016)

In the PFSA case study, all the 13 participants I interviewed indicated they had learned through the PSP workshop. A particularly strong example of learning for PFSA participants, was about the role Artificial Intelligence (AI) could play in creating just and sustainable futures. AI was included as a 'seed' in one of the discussion groups. As outlined in Chapter 3, the 'seeds' were small-scale initiatives that could be described as conducive to creating just and sustainable futures. Each of the four discussion groups worked with three seeds, which they used to develop their scenarios. The discussion group who used AI, thus explored how it could be used to create more equitable and sustainable urban land use, and as a tool for environmental governance. When I interviewed them, several members of this group indicated, that they had learned

through thinking about AI. For example, Penelope, who had an interest in AI prior to the workshop, talked about how she had changed the way she thought about it:

"I was really challenged constantly on what does this mean? I learned a lot about people's concerns [about AI], and the other participants learnt a lot about the technology and what AI actually means." (Penelope, 2016)

Similarly, another participant, Elliot indicated that his understanding of AI, as well as how it could be used, had changed because of his participation in the workshop:

"I learnt a lot around the AI stuff, because my understanding was completely different. I didn't realise that AI existed in as much as it does already, and that I'm using it every day." (Elliot, 2016)

The above findings show that the expectation PSP can enable learning was commonly realised through bringing together different stakeholders and enabling them to engage with the knowledges of others. This corresponds with the wider assumptions that PSP is most useful for learning, described in Chapter 2. My results find particular resonance with Johnson et al. (2012) who describe PSP as a 'vehicle for learning,' (p.10) because it incorporates diverse knowledges into the narratives of future events, conditions and trajectories that it creates. My findings also resemble the results of the recent review by Oteros-Rozas et al. (2015), which indicate that through encouraging participants to engage with different knowledges, PSP enables them to learn about the complexity of SEPs, as well as the impacts environmental changes could have on them. This corresponds with the explanation of Ramirez and Wilkinson (2016) that scenario planning can encourage 'reframing,' whereby the exchange of perspectives through exploring alternative future narratives enables the development of new knowledge, as well as a more holistic understanding of wider contexts. I therefore reason that learning is a highly important benefit associated with PSP. However, I also found that practitioners struggled to articulate how learning occurs in PSP, and how and why it varies for different participants. I discuss this further in Section 4.3.3, below.

4.3.3 "The thing I really would like is a stronger link to an underlying theory." – Limited understanding of how learning occurs

My findings show that, although learning is a highly significant benefit of PSP, practitioners showed a lack of theoretically-informed understanding, regarding how learning

occurs. As shown in Section 4.3.2, it appeared that learning was linked to interactions between participants from different professional roles, disciplinary perspectives, socio-economic backgrounds, and worldviews. However, in the case review, not a single paper explicitly provides a theoretically-informed account of how learning occurs. The authors typically assume learning could occur, based on reports by other practitioners, rather than on theoretically-grounded explanations of learning in PSP. For example, Mistry et al. (2014) assume PSP could enable learning, based on a paper by Johnson et al. (2012). However, that paper itself does not offer a theoretically-informed account of how learning in PSP occurs.

In the practitioner interviews, all the informants I asked about their theoretical understanding of learning indicated that they had never had the mandate or the capacity to engage with learning theories. For example, Barry, who had extensive experience conducting PSP processes through a global research organisation, was a strong proponent of PSP as a method for learning. However, when I asked about his understanding of learning theories, he admitted he had: *"none whatsoever."* He explained that this was because his organisation focused on practice-based, rather than academic research:

"We're not an academic organisation, so we don't focus on learning theories and things like that... Of course, we know that people have learnt but we haven't taken a systematic approach to measuring learning." (Barry, 2016)

This reflects the relative immaturity of PSP practice in the field of social-ecological resilience. Although the review paper by Oteros-Rozas et al. (2015) shows an increasing appetite for reflection on PSP in this context, my findings indicate that the mandate to link PSP to theory has thus far been limited.

However, the importance of developing a theoretical understanding of learning was emphasised by one highly-experienced practitioner, Terry, who had many years of experience facilitating PSP, and other futures-thinking methods. He argued that there is a need for a better theoretical understanding of learning in PSP, to assess its usefulness:

"It's not theoretically grounded, Scenarios is a practice-led discipline... The thing that I really would like, is a stronger link to an underlying theory, so that we can know when [PSP] is being done well, when it is being done badly, how we should be doing it." (Terry, 2016)

In Chapter 5 of this thesis, I contribute specifically to developing such a theoretically-grounded understanding of how learning occurs.

I also found there was limited explicit information regarding how and why learning varied between different participants. In the case review, I typically found it difficult to distinguish who learned in the PSP processes that the papers described, and how learning varied between different participants. For example, Tschakert et al. (2014) indicate that learning occurred between *'local actors, external academics and NGO practitioners,'* (p.1053) but do not indicate which group of participants learned what.

That said, six of the 23 cases that reported learning, demonstrated some awareness of variation in learning, or at least the factors that could influence such variation. For example, Wesche and Armitage (2014) observed that, in their PSP with local-level indigenous communities: *'participants were more likely to identify local-level forces and direct effects of change on livelihood dimensions, whereas external forces and indirect impacts were likely underestimated,'* (p.1105). They attributed this to the fact their participants had limited formal education, and were unused to 'hypothetical approaches' of thinking about the future. This indicates that learning may be influenced by participants' level of formal education.

In another example, Palomo et al. (2011) described how, in their case, local-level participants had refused to consider how a particular, regional-level driver would influence their local-level scenarios, because they believed it was not relevant in the local context. The authors describe how this driver, concerning the role of new technologies: *'did not translate to [the local scenario] because the working group believed that it did not apply,'* (p.23). They go on to reflect that it would have been interesting to explore the influence of this driver in the local area, which implies the participants missed out on learning about it. In this instance, what participants learned appeared to be influenced by the fact they did not consider the information they encountered to be relevant.

The issue of how and why learning varies is an important consideration, however. As reasoned in Chapter 2, ideas about sustainability are socially constructed, and are therefore highly contested and political (Patterson et al., 2017). As such, there are inevitably power imbalances in any process professing to incorporate the knowledges of different stakeholders. This means the knowledge of some participants may be privileged over that of others (Kothari, 2001). It is therefore important to understand who benefits from PSP, as well as who does not. I discuss this further in Chapter 6, and indeed, show that learning is often shaped by power imbalances. Next, I consider another purported benefit of PSP, developing strategies to help tackle SEPs.

4.4 "You can actually try those adaptations out, without having to wait until that future unfolds." - Developing Strategies for tackling complex problems

4.4.1 Expected benefits

In their review of the use of PSP in public policy, Volkery and Ribeiro (2009) posit that it can enable stakeholders to develop innovative responses to anticipated future developments. Similarly, Hughes (2013) asserts that PSP can help stakeholders to consider short-term actions and policies in the context of their potential future consequences. These assumptions were reflected in the case review, in which 24 of the 30 cases expected PSP to help develop responses to SEPs. They all expected PSP to have this benefit through enabling participants to imagine and test responses to potential future conditions, challenges, and opportunities. This is demonstrated in Table 4.3, below.



Table 4.3 - Evidence that developing and testing responses to SEPs was an expected benefit in 24 cases, based on my interrogation of the expected

benefits in peer-reviewed literature, on specific cases of PSP.

Case Study using Scenario Planning	Evidence of cases expecting PSP to help develop and test responses to SEPs
Rivard and Reay (2012) - exploring the future of Malawi's energy sector.	Exploring pathways to more efficient energy systems.
Brand et al. (2013) - ecosystem services in the Swiss Alps.	Developing strategies to deal with ongoing global changes, at the regional level.
Carpenter et al. (2006) the Millennium Ecosystem Assessment.	Exploring possible options and trade-offs for future management of ecosystem services.
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem service analysis in Tanzania.	Providing information on potential policy changes and their impacts in the future.
Malinga et al. (2013) - ecosystem service assessment in South Africa.	Identifying which ecosystem services were most important for local communities, and how they may be affected by future change.
Mistry et al. (2014) describe a scenario planning process on ecosystem management in Guyana.	Exploring the implications of different management strategies, at different spatial levels.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	Creating proactive strategies to work towards desirable outcomes, as well as improve overall decision-making.
Plieninger et al. (2013) - ecosystem services provided by cultural landscapes in Germany.	Identifying and mitigating threats to enable stakeholders to work towards renewal of 'cultural landscapes'.
Gidley et al. (2009) - climate change adaptation in Australia.	Assessing possible adaptation options.
Henly-Shepard et al. (2015) - improving adaptive capacity to hazards in Hawaii.	Developing more anticipatory adaptation strategies for community disaster planning.
Ravera et al. (2011a); and Ravera et al. (2011b) - adaptation in Nicaragua.	Exploring possible adaptation options.
Ravera et al. (2011a); and Reed et al. (2013) - environmental management and adaptation in UK uplands.	Preparing for potential threats and opportunities.
Shaw et al. (2009) - local level climate change adaptation.	Increasing capacity for climate change adaptation and mitigation.

Tschakert et al. (2014) – climate change adaptation in Ghana and Tanzania.	Investigating possible responses to change.
Wesche and Armitage (2014) - environmental change in northern Canada.	Exploring potential adaptation options and enable stakeholders to implement their own adaptation strategies.
Sheppard et al. (2011) - engaging local level stakeholders in climate change action and awareness.	Piecing together climate change information about local impacts and policy options. Then feed that into practical planning processes.
Van Berkel et al. (2011) - rural development in Portugal.	Exploring barriers and opportunities for rural development options.
Vermeulen et al. (2013); and Vervoort et al. (2013) – food security under climate change.	Testing potential strategies, policies and research to deal with future changes
Bohensky et al. (2009) - options for ecotourism in Papua New Guinea.	Exploring strategies to deal with a range of alternative situations in the future.
Brown et al. (2001) - marine protected area management in Tobago.	Assessing the impacts of different trade-offs.
Palomo et al. (2011) - protected area management in Spain.	Identifying practical steps to mitigate conflicts between conservation and economic development.
Pert et al. (2010) - biodiversity conservation in Queensland, Australia.	Identifying options for action.
Sandker et al. (2007) - protecting forests in Indonesia.	Examining the impacts of a large-scale conversion of forests to palm oil plantation.
Jessel and Jacobs (2005) – implementation of the European Water Framework Directive.	Exploring all the long-term impacts of implementing the Water Framework Directive.

Source: author construct



In one, specific example on encouraging sustainable ecosystem management, Palacios-Agundez et al. (2013 p.7) expected PSP to enable the: *'construction of proactive strategies to adapt management to possible future events.'* Specifically, they: *'analysed how ecosystem services and human well-being might change in a range of plausible futures.'* Furthermore, they: *'identified management strategies,'* aiming to: *'strengthen the link to policy making and achieve a real implementation of the research results in ecosystem management policies.'* Similarly, in the case of the CCAFS East Africa scenarios, described by Vermeulen et al. (2013 p.8360), PSP was expected to develop, and moreover, test national-level policies for food security in different potential future contexts. The authors described how the PSP process was: *'designed to provide multiple plausible, future contexts for decision makers to use in regional, national, and local planning.'* The authors went on to describe how they organised workshops with decision makers, to: *'develop and test adaptive planning actions across the different scenarios.'*

In the two above cases, the emphasis was on linking the responses identified in PSP to policy, but in some cases the focus was more on encouraging actors to adapt to complex problems at the local level. For example, Tschakert et al. (2014) describe a case in which PSP was expected to identify adaptation responses to climate change, in local communities in Ghana and Tanzania. They state that they expected PSP to: *'enhance participants' capacity to embrace change by exploring a range of possible futures, and jointly weighing possible responses,'* (p.1052).

In the practitioner interviews, seven of my 16 informants indicated they had expected PSP to help develop responses to SEPs, through identifying and testing them in different possible futures. For example, Nigel, who was a relatively inexperienced practitioner, stated it was expected to help inform a national government's negotiations for a 'payment for ecosystem services' scheme:

"They wanted to be receiving money to protect their forests to store carbon, so the land use change, and forest change was a big concern. So [we decided to think] about what might, realistically happen under different scenarios." (Nigel, 2015)

He subsequently explained that the PSP was designed to:

"Help the government put together their arguments to take to Copenhagen, for negotiations, so they could get a good deal out of [payments for ecosystem services]." (Nigel, 2015) Hence, there was a strong expectation that PSP would help develop and test responses, to anticipated ecosystem change, which the government could use to support their negotiations.

In another example, Mary, an interdisciplinary researcher who had conducted PSP processes on upland conservation, expected PSP to both identify adaptation options, and then test them in different plausible futures. She asserted that PSP could help participants: *"adapt to whatever change might be coming."* When I asked her to explain how she thought PSP could help with this, she explained:

"Identifying different futures enables people to think up different adaptive options, that could help them to adapt in those futures... because you can actually try those adaptations out without having to wait until that future unfolds." (Mary, 2015)

The aim of developing and testing responses to SEPs was particularly prominent in the FSF case study. The organisers expected the PSP workshop to help develop and test responses to food and nutrition poverty, with the intention of informing the Tanzanian government's Five-Year Development Plan (FYDP). When I interviewed one of the facilitators, Thomas, he showed an assumption that the workshop would inform the FYDP:

"This process will help the government to have the relevant information, to make a good plan." (Thomas, 2016)

He then reinforced this in the final meeting between the facilitators, the day before the workshop, by stating that the workshop was intended to "*feed into*" the food and nutrition policy for the new FYDP. Similarly, when another facilitator, Mike, introduced the aims of the PSP on the first day of the workshop, he explained the aim was to create scenarios to "*develop and test strategies*," for FNS to be included in the FYDP. However, as will be shown in Section 4.4.3, a failure to include influential government stakeholders in the workshop, meant this benefit was not realised.

In the PFSA case study, the organisers did not explicitly indicate that they expected PSP to help develop and test responses to SEPs. As I showed in Section 4.3, their emphasis was on learning about how positive futures might look for different participants. However, in five of the eight pre-workshop questionnaires that I received responses to, the respondents indicated they expected PSP to help them explore strategies for creating just and sustainable futures. For example, Barbara stated: "It will help galvanize action and help us develop a strategy [to achieve just and sustainable futures]" (Barbara, 2016)

This implies that, although the official expectations of the PFSA organisers did not include developing and testing responses to SEPs, there was an assumption among some participants that this would occur. I investigate how successful PSP was in meeting these expectations, in Section 4.4.2, below.

4.4.2 Reported Benefits

The above findings show that both practitioners and participants, expected PSP to help develop and test strategies, for tackling complex problems in SES. In the case review, 23 of the 24 cases, that expected PSP to help develop and test responses to SEPs, also reported this as a benefit. This is illustrated in Table 4.4.



Table 4.4 - Evidence that 23 cases reported that PSP helped develop and test responses to SEPs, based on my interrogation of the reported benefits in

peer-reviewed literature, on specific cases of PSP.

Case Study using Scenario Planning	Evidence of literature reporting that PSP helped develop and test responses to SEPs
Rivard and Reay (2012) - exploring the future of Malawi's energy sector.	Identified possibilities for pathways to more efficient energy systems.
Brand et al. (2013) - ecosystem services in the Swiss Alps.	Helped anticipate future changes and develop strategies to deal with them.
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem service analysis in Tanzania.	Enabled a map of winners and losers under different scenarios to be produced.
Malinga et al. (2013) - ecosystem service assessment in South Africa.	Helped identify threats and opportunities for communities to shape their own future.
Mistry et al. (2014) describe a scenario planning process on ecosystem management in Guyana.	Enhanced understanding of different plausible trajectories.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	Enabled identification of policy options.
Plieninger et al. (2013) - ecosystem services provided by cultural landscapes in Germany.	Developed new ideas for facilitating and managing ecosystem services at the landscape-level.
Gidley et al. (2009) - climate change adaptation in Australia.	Identified a range of possible adaptation options.
Pearson et al. (2010) - sustainability planning in Australia.	Informed practical planning of development.
Ravera et al. (2011a); and Ravera et al. (2011b) - adaptation in Nicaragua.	Produced adaptation options.
Ravera et al. (2011a); and Reed et al. (2013) - environmental management and adaptation in UK uplands.	Facilitated adaptation to future change.
Shaw et al. (2009) - local level climate change adaptation.	Improved adaptive capacity and action on climate change.
Tschakert et al. (2014) – climate change adaptation in Ghana and Tanzania.	Created a space to explore responses to potentially harmful climate change.
Sheppard et al. (2011) - engaging local level stakeholders in climate change action and awareness.	Led to greater awareness of the response options available.

Van Berkel et al. (2011) - rural development in Portugal.	Produced a comprehensive analysis of the human and environmental barriers and opportunities for rural development options.
Vermeulen et al. (2013); and Vervoort et al. (2013) – food security under climate change.	A wide range of policy options under different scenarios were identified.
Bohensky et al. (2009) - options for ecotourism in Papua New Guinea.	Enabled investment in a range of strategies to deal with different alternatives.
Brown et al. (2001) - marine protected area management in Tobago.	Enabled a move towards decision-making, based on trust and community action.
Palomo et al. (2011) - protected area management in Spain.	Helped participants to identify practical steps to desirable outcomes.
Pert et al. (2010) - biodiversity conservation in Queensland, Australia.	Made the need for action clear
Mora et al. (2014) - forestry in France.	Help adaptation strategies to move away from single, technical solutions, to ones that can facilitate rapid social and institutional change.
Sandker et al. (2007) - protecting forests in Indonesia.	Facilitated consideration of the consequences of different possible actions.
Jessel and Jacobs (2005) – implementation of the European Water Framework Directive.	Helped to identify key challenges and resolve them, by identifying feasible management options.

Source: author construct



In one, particularly telling instance, Palacios-Agundez et al. (2013) state that PSP had a direct impact on policy, by including key policy-makers in the process: *'This local, participatory, scenario-planning process is already having a policy impact thanks to the involvement of public administration technicians and policymakers on the recently renewed strategic policy planning for sustainability,'* (p.20). They subsequently report that the outputs of the PSP would be used to inform guidelines for ecosystem management policies. Similarly, in the CCAFS East Africa scenarios, Vermeulen et al. (2013) report that participants were able to identify: *'adaptations expected to work under any scenario with appropriate adjustment, as well as scenario-specific options,'* (p.8360).

This was reflected in the practitioner interviews, in which all 7 of the practitioners who expected PSP to help identify and test responses to SEPs, reported that it did so. For example, the practitioner, Ronald, stated that participants in his PSP had identified strategies for adaptation to environmental change by reading through the scenarios and: *"picking out the opportunities and challenges."* He illustrated using an example in which the weather had become more variable in the scenario, so participants identified increased cooperation as a response:

"They were saying the weather has become more difficult to deal with, so it's going to be easier for those who are part of co-ops and community associations and things like that." (Ronald, 2015)

Similarly, the experienced practitioner, Mary, explained how the information gleaned from the PSP she was involved in had been used to create:

"A project feasibility tool, [which enabled] farmers around the country to actually look at different scenarios in the context of their farm business." (Mary, 2015)

This tool helps farmers to assess how much profit they could make from carbon storage, through peatland restoration, as well as what costs they might incur by doing so, under different scenarios. She explained this helped farmers to: "*make a decision that is as low-risk as possible, over the next thirty years.*"

Another practitioner, Nigel, who I mentioned in Section 4.4.1, indicated that PSP had enabled stakeholders to identify responses to anticipated ecosystem change, which the

government used, to support its negotiations for a 'payment for ecosystem services' scheme. He stated that the responses were:

"Picked up directly by policy in Tanzania, and certainly helped [the government] in terms of carrying on formulating their policy management." (Nigel, 2015)

Indeed, he claimed that the government negotiators had *"made a splash,"* in their negotiations, which implies they achieved a favourable deal. This highlights that, although PSP met stakeholders' expectations, it was clearly influenced by the government's interest in strengthening its negotiating position. The interests of some stakeholders could thus have been subsumed under the interests of the government. This emphasises the importance of investigating the factors that influence who benefits from PSP, which I discuss in Chapter 6.

The above findings correspond with the assumptions made by Volkery and Ribeiro (2009) and Hughes (2013), that PSP can help stakeholders develop responses to SEPs and test them in alternative possible futures. These findings may be explained by the dual concepts of reframing and reperception, as put forward by Ramirez and Wilkinson (2016). Reframing provides groups and individuals with alternative future contexts, in which to explore challenges and opportunities for change. This can lead to reperception, whereby participants develop new responses to better cope with challenges and maximise opportunities. However, to develop and test these responses is one thing, to implement them, is another. In Section 4.4.3 below, I investigate whether the responses to SEPs, developed in PSP processes were implemented.

4.4.3 "*I don't think the effect has lasted*" – Limited evidence for the implementation of responses, developed in PSP

The above evidence indicates that PSP is useful for developing and testing responses to SEPs. However, I found limited, detailed evidence for these responses being implemented in reallife. In the case review, none of the papers that report PSP helped to develop responses, indicated that they had been implemented. For example, Palacios-Agundez et al. (2013) report that PSP informed the creation of guidelines for ecosystem management policies, as described in Section 4.4.2. However, they do not indicate whether these guidelines were heeded by policy-makers. Similarly, although Vermeulen et al. (2013) report that participants identified adaptation options, they do not indicate whether any of these options were implemented. In fairness, it may be that the above responses were implemented, but only after the papers had been published.

Furthermore, it could be that the futures in which those responses could be necessary have not yet unfolded.

In the practitioner interviews, I therefore asked my informants to reflect on whether any responses, developed in specific PSP processes, had been implemented. Typically, they were only able to guess at what impacts might have happened after PSP took place, with the exceptions of Mary and Nigel, whose examples are described in Section 4.4.2, above. This was aptly illustrated by the practitioner, Greg who assumed that participants' learning in PSP had informed subsequent decision-making, but could not pinpoint any detailed examples of this:

"I don't know concretely, but I think now they have this information, and they've thought about what these adaptation options are. I don't think they suddenly went and implemented them all, but maybe those people are in decision-making positions, and they more readily have access to that information, and they've already thought about it before." (Greg, 2016)

Similarly, Ronald, who reported that PSP had helped participants identify adaptation strategies, as described in Section 4.4.2, could not say if any of these strategies had been implemented:

"It's not like people would remember the scenarios and think 'yeah we read about the bed and breakfasts,' [as an opportunity for adaptation]. I can say that it had immediate positive effects but, I can't say that it's lasted now. It's not like it has completely changed the community, it was just a very positive experience for everybody to be part of." (Ronald, 2015)

It is evident then, that although PSP helped to identify responses to SEPs, practitioners struggled to articulate whether they had been implemented in real-life.

In the FSF case study, there was no evidence that PSP delivered on the assumption that it would inform the FYDP. When I interviewed the facilitators, separately, two months after the workshop, none of them could provide any evidence for the workshop having contributed to the FYDP. However, one of the facilitators, Thomas, suggested that, through sharing the learning from the workshop:

"We will indirectly contribute to policy-making, but we won't have a direct influence." (Thomas, 2016)

The other two facilitators, Mike and Harry also admitted the workshop had not been able to influence the FYDP, but Mike initially suggested the workshop might still influence food and

nutrition policy later "*down the line*." However, a conversation with him 18 months after the workshop indicated that it has had no discernible influence.

In PFSA, although the emphasis was predominantly on learning, as I described in previous sections, five of the 13 interviewees did indicate that they were disappointed by the fact no implementable strategies were developed in the workshop. For example, Rachael, who was a social entrepreneur, and was therefore interested in quickly identifying and implementing actions, was frustrated by a lack of thought, in her group, about what everyday actions they needed to implement in the short-term. She queried the value of her group's scenario, saying:

"What does that mean on a Monday morning? Like next week, Monday morning, what does that mean? There was a tension between balancing what needs to be done now, versus a different future [that] we imagined." (Rachael, 2016)

Similarly, Darren who was an artist, and also focused on practical actions, complained that there was a lack of focus on implementing the ideas from the scenarios. He also described the scenarios as 'haywire,' suggesting they were too avant-garde, to be practically useful:

"I think that's one thing that our conference lacked, was there wasn't any mention of [the scenarios] being applied in any way... So, you can go as haywire as you like, but if it's not implementable, even from tomorrow onwards, and you haven't imagined it in a way that is doable, and then what's the use of that?" (Darren, 2016)

The above evidence demonstrates that, although PSP helps stakeholders identify and test responses to SEPs, there is limited evidence of these responses being implemented in real-life. This was explained by one practitioner, Carol, who had extensive experience linking the outcomes of PSP to policy. She stated that, for responses identified in PSP to be implemented, further work is needed, following PSP workshops, to connect them to real-world policies and actions:

"If you treat that one moment of scenario development and scenario-guided policy testing as the be-all and end-all of the exercise then yeah, there are weaknesses, but if you treat it as a process, so that various people have inputs, and you connect it to processes of policy-development, in the end you have policy impact." (Carol, 2016)

This reflects the argument by Chaudhury et al. (2012), that careful efforts are needed to connect the outcomes of PSP processes with real-life policy and practice. They indicate that the results of PSP need to be perceived as credible, salient and legitimate by stakeholders for them to be implemented. Vervoort et al. (2014) also indicate that the link between PSP processes and real-life policy and practice can be strengthened through long-term engagements with PSP. They reason that PSP is most likely to lead to implementable responses when it is embedded in existing systems of decision-making, rather than when PSP processes are used as one-off interventions. Likewise, Brown et al. (2016) argue that for responses to be implemented in real-life, PSP should be used as part of ongoing community engagements, rather than as an isolated process. It is evident, therefore, that although PSP can help to develop responses to SEPs that are implemented in real-life, concerted efforts are needed, beyond just conducting isolated PSP processes, for implementation to occur.

An additional explanation was put forward by the experienced practitioner, Mary. She observed that responses developed in PSP processes are not frequently implemented, because practitioners do not systematically select participants with the capacity to implement them:

"One of the most important factors that influence the outcome of any participatory process is the composition of the group. So, stakeholder representation is crucially important... I believe this is a step that is very rarely included in scenario analyses." (Mary, 2015)

This was reflected by another practitioner, Vera, who had several decades of experience as a professional facilitator of PSP, as well as other futures thinking tools. When I asked her what she thought was required for PSP to be successful, she replied:

"One of the key conditions is to have a group of people with the motivation and ability to act." (Vera, 2016)

This emphasises the importance of including influential stakeholders in PSP.

The problem of not including stakeholders with sufficient capacity to implement responses was particularly apparent in the FSF case study. The organisers originally invited government stakeholders involved in the development of the FYDP, with the rationale that this would help directly link the outcomes of the workshop to policy. However, most of them were unable to attend, and the eventual participants lacked the capacity to influence decisions regarding the FYDP. As described in Section 4.4.2, the workshop therefore did not produce the
desired benefits in terms of informing government policy. Indeed, as the facilitator, Thomas described:

"To start with, we expected some people who could influence the FYDP, but unfortunately most of them did not come. That was one of the limitations." (Thomas, 2016)

The above evidence demonstrates, that, even when attempts are made to connect PSP processes to real-world policies and actions, a failure to include influential participants can limit opportunities for responses to be implemented in real-life. This resembles the findings of Nieto-Romero et al. (2016) in their assessment of PSP processes in Eastern Europe. They explain that a lack of influential participants is an important barrier that reduces the likelihood of responses identified in PSP being implemented. Similarly, in a paper analysing the challenges of using PSP to influence climate change adaptation in the public sector, Rickards et al. (2014) explain that a failure to include stakeholders with the ability to implement the outcomes of PSP makes it more difficult to secure their 'buy-in'. This subsequently reduces the likelihood that responses, developed in PSP, will be implemented.

The apparent limitations of PSP for supporting real-world responses to SEPs emphasise the importance of considering the strengths and weaknesses of PSP, in different contexts, and comparing it to alternative methods for tackling SEPs. However, my data showed limited explicit discussion of this by practitioners. I investigate this issue further in Section 4.5, below.

4.5 *"I wasn't able to concretely follow up on it."* - Limited provision for evaluation of PSP

4.5.1 Limited consideration of the strengths and weaknesses of PSP

As the data presented in previous sections attests, the information provided by the authors of the papers in my case review is rich and insightful. However, their papers included limited information regarding the strengths and weaknesses of PSP, especially when compared to alternative methods for stimulating dialogue about tackling SEPs. Limitations of PSP were discussed in 25 of the 30 cases in the case review. In Sections 4.2 to 4.4, I described the limitations that were specifically associated with achieving the purported benefits of PSP. In addition to these, I found that two general limitations were identified across the cases: representation of

different stakeholders, and the time-intensiveness of PSP. For example, Brand et al. (2013) describe a process in which PSP was expected to help stakeholders develop a more 'systemic understanding' of ecosystem services, by including different stakeholder groups. However, the authors found that some stakeholder groups were underrepresented, and some participants did not have time to attend the whole process. Similarly, Van Berkel et al. (2011), indicate that the representation of some farming stakeholders was limited because they had other commitments and did not have time to attend the entire process. Swetnam et al. (2011), and Reed et al. (2013) also highlight the problem of PSP being very time-intensive.

The fact that multiple cases acknowledge these limitations shows that they are welldocumented. Discussions regarding the limitations of PSP were typically only given short shrift, however, and typically consisted of anecdotal evidence and reflections by the authors, rather than the results of systematic evaluation processes. In five of the 30 cases, there was no explicit discussion of any limitations of PSP. Perhaps the most surprising of these, because of its prominence, was a paper regarding the use of PSP in the Millennium Ecosystem Assessment (MA). Carpenter et al. (2006) offer an overview of the scenarios process used in the MA, but they do not explicitly report on any limitations. The MA's overall report does include some consideration of limitations in the 'Lessons Learned' section (Carpenter et al., 2005). These include issues of making the global MA scenarios relevant at the local level, and to policy-makers, which reflects the limitations of PSP in contributing to the development of strategies to tackle SEPs, described in Section 4.4, above. It is evident then, that some limitations of PSP, in the MA, were considered, even though they were not reported in the overview paper by Carpenter et al.

Overall, however, any reflections on the limitations of PSP, and its relative merits and drawbacks, are poorly reported. This is presumably at least partially because of editorial restrictions, that impede the inclusion of lengthy discussions of limitations and failures. Indeed, as Baxter and Eyles (1997) explain in their seminal paper, editorial constraints typically mean that critical reflections on methods are left-out, to create space for presentation of research findings.

Another explanation revealed in the practitioner interviews is that limited provision is given for systematic evaluations of specific PSP workshops. This was particularly well-illustrated by the practitioner, Greg. He stated that, in his case, PSP had helped an indigenous community identify implementable actions that could help them adapt to environmental change. However, the funding for his project had ended before he was able to follow-up on the extent to which these actions were implemented:

"That was the end of the project that I had funding for and I wasn't able to concretely follow up on it. It would be a really good question to go back and see what had been adopted, what hadn't been adopted, but I don't have a very good sense of that." (Greg, 2016)

This was explained by another practitioner, Gavin, who had extensive experience using PSP in several global and regional processes. Gavin explained he had, along with a colleague, conducted a study into the extent that practitioners evaluated the PSP processes they conducted. He described how their findings showed there was a lack of provision for evaluation:

"Very few of them did very much [evaluation], and most of them said the reason why, was that they ran out of money, or ran out of time." (Gavin, 2016)

He went on to attribute the lack of provision to a limited mandate for evaluation of PSP processes, when they are conducted by academic organisations:

"If it's not happening, part of it is probably because people are working in contexts where there isn't such an impetus, or demand for showing impact." (Gavin, 2016)

In my two case studies, there was limited evidence of any provision being made for evaluation. However, in both cases, the facilitators welcomed my research, because they saw it as an opportunity for evaluation by an independent researcher. For instance, in FSF, the facilitator, Mike, asked me directly if I could include evaluative questions for him, in my interviews with participants. I politely declined as this would have compromised my position as an independent researcher. In PFSA, I was asked to meet with the monitoring officer for their funding body, to discuss my ideas about how learning could be evaluated. In this case, I agreed, because it did not influence my interviews with participants. The keenness of FSF and PFSA for me to help them evaluate their PSP processes indicates there was a desire for evaluation to happen, but the projects themselves lacked the mandate, or the capacity, to do it themselves.

This indicates that, although practitioners acknowledge the limitations of PSP, and would welcome opportunities for more systematic evaluation, the context in which they work does not create either the mandate or the resources necessary for such evaluation to occur. When I asked Gavin why he thought this was the case, he suggested two reasons. Firstly, he stated that in this field of practice, the focus had historically been on producing scenarios as ends in themselves, rather than as means for achieving other outcomes. As such, the outcomes of PSP processes had been given limited attention:

"In the early projects that I was involved in, the scenarios were seen as the end point, not really what you do with them or what they might trigger." (Gavin, 2016)

Secondly, he explained that whilst practicing scenarios in public-sector research, the mandate for evaluation had been much higher because of a public demand for accountability:

"More and more there is a push for us to be able to measure that and monitor it and evaluate it and report on that. We're not going to get supported to do this kind of work without being able to demonstrate that it is actually doing something good." (Gavin, 2016)

In comparison, he posited that academic practitioners have not been subject to the same demand for accountability:

"Academic researchers think they might be able to get away with not demonstrating impact." (Gavin, 2016)

Gavin's explanation alludes to both a relative immaturity and an isolation of PSP practice in the field of social-ecological resilience. Engagement with literature on the use of scenario planning in the more established field of corporate strategic planning shows that scenarios are emphasised as means through which to learn about the present, rather than as ends in themselves, for learning about the future (Ramirez and Wilkinson, 2016; Wilkinson and Kupers, 2014). However, Gavin went on to propose the practice of PSP in this context was beginning to mature, leading to an increased emphasis on evaluating the outcomes of PSP:

"I think the practice of scenario planning has evolved. Its maturing and it's going beyond just creating visions for the future, and talking about them but [thinking about] what are we going to do next? What do we do with them? How do we actually make something happen?" (Gavin, 2016)

However, as I reflect on in Chapter 7, PSP practitioners in this field could usefully engage with more established fields of PSP in corporate strategic planning and futures studies to improve their evaluation and understanding of outcomes from PSP.

4.5.2 Limited consideration of alternative methods

The evidence presented in Section 4.5.1 indicates that the limitations of PSP are recognised by practitioners. Despite this, only three of the 30 cases in my case review showed any consideration for methods that could have been used instead. Palomo et al. (2011), compare PSP

to quantitative models, and indicate that PSP is better at accounting for uncertainty, in the development of conservation plans for a 'protected area': '*in contrast to predictions and models, scenarios explore the uncertainty of future events, and thus decisions based on scenarios provide greater resilience to surprise*,' (p.24). Bohensky et al. (2009) also compare PSP to quantitative modelling. They assume it is better for exploring opportunities and uncertainties, in the development of eco-tourism in Papua New Guinea: '*in such uncertain, uncontrollable situations, scenarios have an advantage over more quantitative models… in their flexibility, transparency and space for narrative to describe possible futures in their complexity,' (p.2827).* The third case, Wollenberg et al. (2000) compares PSP to 'Participatory Rural Appraisal.' They show an assumption that PSP is better equipped to enable learning, by highlighting different stakeholders' assumptions about forest management: '*scenarios focus on the analysis of uncertainties, drivers of change and causal relationships… to a greater extent than do these other techniques*,' (p.67). However, not even these three cases offer a direct comparison, based on empirical evidence, of PSP with any other method.

The absence of explicit, detailed discussions about the strengths and weaknesses of PSP, as compared to alternative techniques, could make it difficult for practitioners to make an informed choice regarding whether PSP is the most useful method, in the context in which they seek to use it. In the practitioner interviews, the highly experienced professional facilitator, Terry indicated that this judgement could mean the difference between whether a PSP process was successful or not. However, he stated that other practitioners often use PSP without making such an informed choice:

"Well methodology and process being good or bad, I think is dependent on informed choice, knowing why you want to do it, where and when it can be useful and when not... You find loads, and loads, and loads of people do scenarios because they think, somebody else has done it, so it sounds like a good idea." (Terry, 2016)

Indeed, when I subsequently asked if PSP was just a fashion, Terry responded: "Yeah! Absolutely!"

Terry's assertion was reflected by the responses of another practitioner I interviewed, Greg. Greg had used PSP in his doctoral research, as a means of documenting indigenous knowledge about environmental change and connecting it with scientific knowledge. He admitted that his use of PSP had been strongly influenced by the knowledge it had been used, apparently successfully, by one of the lead scenarios coordinators on the influential MA: "I was familiar with the scenario work that he had done. He had done some scenario work with communities and it seemed like it might be a good approach to use." (Greg, 2016)

He went on to admit that, although he suspected other participatory methods could have helped connect indigenous and scientific knowledge, he had not considered them:

"To be honest, I didn't really go through a process of thinking about all of the different participatory processes that I could use, and the pros and cons of each... When I came across this method [PSP], it made sense in the context that I was working in, so I used it. I don't know how much more I can justify it." (Greg, 2016)

Greg did show some consideration of why PSP was useful in the context of his work:

"It just seemed like a very effective tool to be able to incorporate different sources of knowledge, in a format that was accessible to people in the community, who had a range of formal education." (Greg, 2016)

However, without any consideration of alternatives, it is difficult to say whether PSP was the most appropriate method to use, in this context.

The notion that PSP is a fashion is vindicated further by the example of the practitioner, Ronald. Like Greg, he used PSP in his doctoral research, which aimed to explore how different stakeholders used ecosystem services. In his case, PSP had already been written into the research proposal by his supervisors. The supervisors had previously been involved in PSP, through a process on dryland development, and through the MA. Ronald acknowledged that he had the option not to use PSP, but still chose to go along with it:

"Of course, I could have changed things round, so I could have chosen not to [use PSP], but it was already planned... They had an interest in the method and they wanted to explore it further... I thought it was really a cool idea. I think it's just intriguing, discussing and then exploring what can happen in the future, and drivers of change of today, and how to imagine them in the future." (Ronald, 2015)

When I asked Ronald to compare the benefits of PSP, to those of other participatory methods, he responded:

"I haven't studied other methods, so I can't really say, I'm afraid. I can't compare the two because I have much more experience with scenario planning." (Ronald, 2015)

In this instance, Ronald's use of PSP was undoubtedly influenced by his supervisors' past experiences, and by his own perception of it as a 'cool idea.'

In fairness, Ronald went on to show some evidence of considering alternatives, in that he compared the benefits of PSP for exploring ecosystem services, with key informant interviews. In fact, he reflected that PSP did not offer any additional value as a research tool:

"We could have just asked the experts that are working in the area." (Ronald, 2016)

However, his comparison focused on the benefits of PSP for extracting information on ecosystem services, and not on how useful PSP was for the participants involved, or as a method for participants' learning, or indeed, developing strategies to tackle SEPs.

The above evidence indicates that the use of PSP in the context of tackling SEPs is currently fashionable. It also shows that, although practitioners tend to believe it is a good idea when they use it, they show limited consideration of its strengths and weaknesses as compared to other methods. This finding demonstrates further evidence of the immaturity of PSP practice in this field and emphasises the need for greater reflection on its benefits and limitations, especially in comparison to other methods, as called for by Waylen et al. (2015).

An important caveat to this conclusion was highlighted in my two case studies, in that they were both designed around a variety of specific techniques for doing PSP. Moreover, the processes were supplemented with added features, like 'backcasting,' 'Futures Wheels,' and '3 Horizons.' This was done with the aim of adapting the PSP processes, to the contexts, as well as the purposes, for which they were used.

In the FSF case study, the workshop included the use of visioning and backcasting alongside PSP. As shown in Table 4.1, each of the methods played a different role in contributing to the outcomes of the workshop.

Table 4.5 Specific tools and activities used to adapt PSP to the needs of Food Security

Method	Purpose
Visioning	Exploring what an ideal future would be like if specific food and
	nutrition policy objectives were achieved.
Backcasting	Exploring the steps that would need to be taken for the ideal
	future to be realised.
Participatory scenario	Exploring what Tanzania would be like in alternative plausible
planning, using downscaled	futures. Then assessing what challenges could be faced in taking
'East Africa' scenarios	the steps to achieve the ideal future.

Futures, and their purpose

Source: author construct

When I asked the highly experienced FSF facilitator, Mike, to compare PSP to the other aspects of the workshop, he explained that PSP was better at identifying challenges for policies than the backcasting method they used alongside it:

"The visioning and backcasting helps us know what we want to see, and how we can reach that point, but it doesn't show us what challenges we might face. Scenarios help us to identify the challenges." (Mike, 2016)

It is evident then, that Mike had a clear understanding of the strengths and weaknesses of PSP, as compared to other methods, in that context.

In the PFSA case study, the workshop was designed around specific features, like 'Future Wheels,' and '3 Horizons,' with the aim of making the process of imagining just and sustainable futures more systematic. The role of each method is described in Table 4.2, below.

Table 4.6 Specific techniques, and tools used to adapt PSP to the needs of PositiveFutures for Southern Africa, and their purpose

Method	Purpose
Futures Wheels	Imagining the possible impacts of 'seeds,' or small-scale initiatives that encourage just and sustainable conditions, if they were mainstream
	practices.
Cross-impact matrices	Investigating how the seeds could influence each other.
Scenario	Exploring the events, conditions, and trajectories that would constitute the
development	Anthropocene if the seeds were commonplace and influenced each other.
3 Horizons	Investigating the dominant way things are in the present, the way things will be, in the futures represented by the scenarios, and the events, conditions and trajectories that would define the transition between the present and the future.

Source: author construct

This shows that, at least in my two case studies, there was some consideration of the strengths and weaknesses of PSP processes. Furthermore, there was some attempt to tailor them to fit specific contexts, by mixing it with different activities. Overall, though, the case review and practitioner interviews indicate that practitioners struggle to articulate the strengths and weaknesses of PSP, especially in comparison to other methods. It appears, therefore, that PSP is often used without an informed choice being made regarding why, and in what contexts, it is beneficial, and without being subject to stringent evaluation. This is reflected by an apparent lack of evaluative papers in the PSP literature. Indeed, as far as I am aware, the review paper by Oteros-Rozas et al. (2015) is the first to formally assess the benefits claimed of PSP.

The lack of provision for systematic evaluations, could be indicative of a wider reluctance, by practitioners and funders to consider the potential drawbacks and failures of initiatives pertaining to tackle SEPs. Indeed, as Mosse (2005) contends, the perceived efficacy of any initiative is often defined by the prevailing view held by those who commission it, rather than by the intended participants. The prevailing view of practitioners, and their funders, appears to be that PSP is a useful method, for achieving the benefits described in this chapter. It thus appears there is limited motivation for evaluation that could expose the potential drawbacks and failures of using PSP. In Chapter 7, I thus argue that there is a need for more systematic evaluation of the strengths and weaknesses of using PSP to help tackle SEPs, and for comparison with other methods.

4.6 Conclusion

In this chapter, I have explored the expected and reported benefits of PSP for tackling SEPs. My findings show that the use of PSP is typically underpinned by an assumption that it can help participants deal with uncertainty by breaking down, or structuring, future possibilities into alternative narratives of possible future events, conditions and trajectories. Structuring future possibilities is subsequently believed to help participants explore the complexity and uncertainty of SEPs, through highlighting the interconnections between different components in SES. However, I found that, although some informants indicated that structuring futures possibilities was a valuable outcome in its own right, it was only rarely reported as a benefit of PSP. I reason that this is because it is linked to and is sometimes a precursor to other benefits.

I found that learning is commonly expected as a benefit of PSP. This expectation was often reportedly achieved through bringing together stakeholders with different professional roles, disciplinary perspectives, and worldviews. However, despite the prevalence of learning as

both an expected and reported benefit, I found that practitioners had limited theoreticallygrounded understanding of how learning occurs and paid little attention to how, and why, learning varies between different participants. I propose that these issues should be given more consideration, to help assess whether PSP is being practiced effectively, as well as exploring who benefits from PSP, and who potentially loses out.

I also found that proponents of PSP commonly expected and reported that it would help develop responses to SEPs and test them in alternative futures. However, although I found some examples of these responses being implemented, evidence for PSP having an influence on realworld policy and action was scarce. I found two potential explanations for this: i) that PSP needs to relate to long-term processes of policy-making and action, rather than being conducted as a one-off intervention; and ii) that it requires the participation of influential stakeholders for responses to be implemented.

I also found limited evidence of practitioners considering the strengths and weaknesses of PSP, especially as compared to other methods for tackling SEPs. This could be because of limited provisions being made, by practitioners and their funders, for systematic evaluation of PSP processes. I therefore reason there is a need for greater critical reflection on PSP, its use and its benefits and limitations. Specifically, I argue there is a need for more stringent assessment of the strengths and weaknesses of PSP, and for comparison with other methods for tackling SEPs. In Chapters 5 and 6, I contribute to such an assessment, by investigating how learning occurs, and how and why it varies between different participants.

Chapter 5 – How and under what conditions does learning occur in Participatory Scenario Planning processes?

5.1 Introduction

In Chapter 4, I showed that learning is arguably the most important benefit associated with participatory scenario planning (PSP), that may help its practitioners and participants to tackle social-ecological problems (SEPs). However, I also found limited theoretically-informed information, regarding how learning occurs through PSP. As reasoned by one highly-experienced practitioner, Terry, PSP practitioners would benefit from a theoretically-grounded understanding of learning to help assess whether PSP is being used effectively. In this chapter, I contribute to such an understanding.

To do this, I draw on what insights could be gleaned from the case review, but as shown in Chapter 4, although the authors of the papers demonstrated an awareness of which aspects of PSP led to learning, a detailed, theoretically-grounded explanation was lacking. I therefore focus predominantly on findings from my practitioner interviews, as well as from my two case studies – 'Positive Futures for Southern Africa' (PFSA) and 'Food Security Futures (FSF). To interpret these findings, I draw on specific learning theories and concepts, the 'Zone of Proximal Development' (Vygotsky, 1978), 'boundary objects' (Star and Griesemer, 1989), and 'scaffolding' (Wood et al., 1976), as set out in my conceptual framework in Chapter 2. I understand learning as a change in understanding that occurs as a result of some external stimuli.

The chapter is structured as follows: In Section 5.2, I present evidence showing that learning in PSP occurs through interactions between participants from different worldviews, professional occupations, academic disciplines, and socio-economic backgrounds. Processes of learning in PSP can thus be explained as participants 'entering' their Zones of Proximal Development (ZPDs). In Section 5.3, I show how specific, structured activities in PSP processes can act as prompts for discussion. These encourage participants to make explicit their assumptions about the future, which exposes them to different ways of thinking about social-ecological systems (SES). Participants are thus encouraged to push beyond their usual range of thinking, which stimulates learning. PSP processes can thus act as 'boundary objects' (Star and Griesemer,

1989), that foster the exchange of knowledge between different participants, while maintaining their individual identities and disciplinary commitments.

In Section 5.4, I present evidence of three potential challenges, that can limit learning in PSP. Firstly, the types of discussions promoted by PSP processes, unsurprisingly, do not suit all participants, because they do not fit well with everyone's worldviews. Secondly, a lack of expertise in the topics of specific PSPs can limit learning about those topics, and a lack of diversity in participants' expertise, can limit their exposure to different knowledges. Thirdly, as in any participatory process, there are assertive personalities, as well as participants who are less willing to listen to the views of others. This can limit the contributions of less assertive participants. Moreover, it can lead to 'dead-ends' where tensions between participants, result in a break-down in discussions. These challenges highlight the importance of creating conducive conditions for learning to occur.

In Section 5.5, I present evidence showing that such conducive conditions can be encouraged by: Firstly, skilled facilitation can maximise opportunities for interactions, guide participants through specific activities, and ensure they interact in constructive ways. Secondly, careful selection of participants can increase the chances of including participants with relevant expertise, and who are willing to listen to, as well as learn from others. Thirdly, well-designed sets of activities help stimulate opportunities for interaction between different participants. The creation of conducive conditions for learning can also be assisted by holding PSP processes in a comfortable space, and by allocating sufficient time, human resources and, ultimately, funding to them.

5.2 *"The learning potential lies in interactions."* – Learning occurs through interactions between different participants

As indicated in Chapter 2, literature on PSP considers it useful as a method for incorporating the knowledges of different stakeholders on SEPs. There also appears to be an assumption that incorporating diverse knowledges is linked to learning. In their frequently-cited paper, Johnson et al. (2012) describe PSP as a 'vehicle for learning,' (p.10), because it can incorporate diverse ideas from different viewpoints. Similarly, in their review of PSP, Oteros-Rozas et al. (2015) indicate that PSP can expose participants to different knowledges, which enables them to learn about the interconnections between different components of SES. My findings

indeed show that learning can occur through interactions between participants with different worldviews, professional occupations, disciplinary perspectives, and social-economic backgrounds.

As described in Chapter 4, none of the papers in the case review provided a detailed, theoretically informed, account of how learning occurs. In 14 of the 30 cases, however, the authors appear to show an implicit awareness that learning was linked to interactions, specifically 'discussion,' 'collective thinking,' and 'knowledge exchange,' between different participants. This is illustrated in Table 5.1, below. In each of these cases, the authors subsequently report that learning occurred. I thus infer that learning is linked to these interactions, but this is not clearly explained in the literature. My findings from the practitioner interviews and case studies shed more light on this.



Table 5.1 – Evidence from the case review, that learning occurred through interactions between different participants.

Case of PSP	Evidence of interactions
Bohensky et al. (2006) - ecosystem services in South Africa	Creating links between different aspects of the scenarios encouraged discussion between
	participants at different spatial levels.
Brand et al. (2013) - ecosystem services in the Swiss Alps.	Discussions around consistency and surprise in potential futures occurred between participants from
	different disciplines and spatial levels.
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem services	'Diverse' participants collectively deliberated on the development of trends and drivers in alternative
in Tanzania.	futures.
Malinga et al. (2013) - ecosystem service assessment in South	Interactive workshops were held with stakeholders from different spatial levels.
Africa.	
Mistry et al. (2014) - ecosystem management in Guyana.	PSP created a 'platform for dialogue' (p.131) between participants with different worldviews.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	PSP encouraged interactions between participants with local, and specialised scientific knowledge.
Plieninger et al. (2013) - managing ecosystem services provided by	PSP encouraged discussion between scientists and local actors.
cultural landscapes in Germany.	
Ravera et al. (2011a); and Reed et al. (2013) -environmental	Interactions occurred between participants with different knowledges.
management and adaptation in UK uplands.	
Shaw et al. (2009) - adaptive action for climate change.	Inclusion of participants from different stakeholder groups expanded the amount of local-level
	information that was included and facilitated knowledge exchange.
Van Berkel et al. (2011) - rural development in Portugal.	Carefully selected stakeholders from different professional roles deliberated on challenges and
	opportunities for the future.
Vermeulen et al. (2013); and Vervoort et al. (2013) - climate change	Different stakeholders explored uncertainties and considered how to overcome potential future
and food security in East Africa.	challenges.
Schulz et al. (2015) - Payments for Ecosystem Services in Brazil.	Carefully selected participants, with different worldviews, deliberated on future threats and how
	they may be overcome.
Brown et al. (2001) - marine protected area (MPA) management in	Different stakeholders deliberated on trade-offs in the future.
Tobago.	
Wollenberg et al. (2000) - use scenario planning in adaptive co-	PSP encouraged knowledge exchange between different stakeholders.
management of community forests.	

Source: author construct



As I described in Chapter 4, all 16 practitioners I interviewed, indicated that learning is an important and highly desirable benefit of PSP. When I asked them to explain how they thought learning could occur in PSP, 10 of them indicated that learning occurred through interactions between different participants. This included interactions between: indigenous and scientific communities, local and national level stakeholders, smallholder and commercial farmers, and many more. In the other six, although they indicated learning occurred, they chose to focus their responses on other benefits of PSP. For example, one practitioner, Barry explained that he lacked a precise understanding of how learning occurred because:

"We're not an academic organisation, so we don't focus on learning theories and things like that..." (Barry, 2016)

Arguably, understanding precisely how learning occurs would be beneficial, regardless of the fact the organisation is not academic. However, this example indicates that not all practitioners focused on learning as a benefit of PSP, nor did they all consider it necessary to understand how it occurred.

Of the 10 practitioners who explained that learning occurred through interactions between participants, one informant, Gordon, who had experience of conducting PSP in several high-profile global processes, articulated this particularly concisely. He stated that:

"The learning potential lies in interactions across disciplines, where people's assumptions are questioned in a respectful way... This leads to learning about different drivers and learning about different people's visions and desires for the future." (Gordon, 2016)

Similarly, Gavin, another practitioner with a high degree of experience conducting PSP, explained that learning occurs through interactions between participants from different sectors and spatial levels. When I asked how he thought learning occurs, he responded:

"It is the interactions between stakeholders that are brought together. They are brought together, with people they don't normally interact with, across those different scales or across sectors, or areas of government, or industry." (Gavin, 2016)

The above evidence shows then, that learning is believed to occur through the interactions between different participants.

In my two case studies, I found that learning occurred through interactions between a range of different participants, including civil servants, teachers and academics in FSF, and artists, academics, and activists in PFSA. In the PFSA case study, all 13 participants I interviewed indicated that learning occurred through the interactions they had with other participants in the workshop. One participant, Penelope, who also had experience of other PSPs, stated, in a particularly eloquent way, that she learned through discussions with other participants, prompted by PSP:

"I learned something from every single person in that group, and that was really, really powerful...there were lots of interesting discussions, and the workshop provided opportunities to have those." (Penelope, 2016)

Another participant, Geoffrey, explained that interacting with different participants had exposed him to different 'angles' of looking at the future:

"[The workshop] brought together very artistic thinkers and creators, and academics, and us from intergovernmental processes, and civil society groups as well, and that enriched the dialogue in a lot of ways, because people were beginning to see and understand the whole forward-looking perspective from a different angle." (Geoffrey, 2016)

He subsequently gave a specific example of how he had learned, through discussions with other participants, about the role artificial intelligence (AI) could play in creating just and sustainable futures:

"I sat in that group with a totally different understanding of what artificial intelligence meant. [To me it meant] we're going to be taken over by aliens, but through Penelope's explanations, I thought 'oh, this is what it actually means, okay!' It's not necessarily just a computer; it's also the digital learning and all these different dynamics." (Geoffrey, 2016)

This provides a detailed example, in which interactions between two participants, who had apparently very different understandings of AI, resulted in learning about a specific topic. Indeed, the participants I interviewed commonly articulated particular points they had learned through interactions with others, or they thought others had learned from them. For example, Rachael recalled how she had learned about an interesting perspective on healthcare from another participant: "Miriam was talking about this notion of why are we obsessed with [spending money on] health, because for one thing to grow, the other must die. If you were to buy into that philosophy, you would save money for somewhere else, because you are thinking cyclically. So that was really, really powerful for me." (Rachael, 2016)

Another participant, Dillan, believed another participant, Lillian, had learned from his expertise on encryption technology:

"I felt I'd had an impact on Lillian, because she'd become a fan of [encryption]." (Dillan, 2016)

His claim was supported by a conversation I had with Lillian during the workshop, in which she said she had found it interesting to learn about encryption. Furthermore, she had already started doing some online enquiry into it.

It was a similar story in the FSF case study. In FSF, six of the 13 participants I interviewed, articulated that they had learned through interactions with different participants in the workshop. For example, one participant, Tracy, who was a teacher, described how she had learned through interacting with participants from other professional occupations:

"There were lecturers here at the University and there were teachers in schools, and there were officers in the education department, and other officers in agricultural department, but we shared information... We are all giving points, someone gives one point, another gives the next point, there are other points, which I learned from my friends, and therefore I learned something." (Tracy, 2016)

Similarly, Gerry, a participant who worked for the Ministry of Health, reported he had learned through interactions between participants from different professional occupations:

"People were interacting, because the group was having people from different areas, different angles. There were those from agriculture, representatives from primary schools and another from the University. Everybody was interacting... As a group, people were just contributing, we were asking ourselves what should be done to intervene for the issue and everybody contributed according to their experiences." (Gerry, 2016)

Another participant, Sally, an academic researcher, described how she had learned from interacting with participants who had different expertise:

"I met with people's different expertise, nutrition specialists, policy makers, one person from the pressure group, from NGOs. Those participants shared their skills, their knowledge, their experience accordingly." (Sally, 2016)

These findings show that learning in PSP occurs through interactions between participants with apparently different knowledges. These interactions create opportunities for participants to interact with different ways of understanding SEPs, which encourage learning. This corresponds with theory on the 'zone of proximal development,' (ZPD) as proposed by Vygotsky (1978). As I outlined in Chapter 2, Vygotsky reasons that people learn through interacting with others who are more capable in a specific field. He explains that, through these interactions, people draw upon the expertise of others, and are thus able to learn things that would normally be beyond their individual learning capacity. In Vygotsky's words, they 'enter' their ZPD. My findings indicate that PSP participants interact with participants who are more capable in different fields to their own, and thus, draw upon others' expertise to learn about ways of understanding SEPs that would be beyond their capacity to learn alone. In this sense, I reason, PSP enables participants to 'enter' their ZPDs.

However, this could arguably be said of any given participatory or educational process that brings participants with different knowledges into dialogue. It is therefore important to investigate if and how specific attributes of PSP stimulate learning. The data in the above paragraphs allude to attributes, such as questioning assumptions about the future, discussing common issues, and creating shared mindsets. I explore these, and other attributes in more detail in Section 5.3, below.

5.3 "Giving people a structure to push beyond where their thinking would normally take them." – PSP prompts discussion

In Section 5.2, I showed that learning in PSP occurs through interactions between participants with different knowledges. In this section, I focus on how the process of imagining alternative future possibilities encourages learning, through triggering and focusing discussions between different participants. This can create opportunities for participants to be exposed to the knowledge of others, and to 'push' their thinking beyond their usual capacity. In the case review, although information regarding how learning occurs is scarce, in 21 of the 30 cases the authors allude to specific aspects of their PSP processes that stimulated discussion. For example, Van Berkel et al. (2011) report that: 'The scenarios acted as prompts in the workshop discussions,' (p.135). The authors explain that deliberating on the effects of specific events, conditions and trajectories on the futures of participants' local area, stimulated discussion about local development issues. For instance, they describe how a scenario that focused on restoring traditional agricultural practices, stimulated discussion about different types of tourism that could be promoted in the area. The authors report that, because of these discussions, participants developed a: 'richer understanding of rural development issues,' (p.136) especially regarding acknowledging the interests of stakeholders from different sectors, including farmers, servicesector workers, and representatives from a national park. Evidence of PSP prompting discussions is prevalent in the case review. I illustrate this in Table 5.2, below.



Table 5.2 – Evidence from the case review that PSP 'prompted' discussions between different stakeholders

Case of PSP	Evidence of PSP 'prompting' discussions
Rivard and Reay (2012) - exploring the future of Malawi's energy sector.	Discussions encouraged by exploring 'structural uncertainties.'
Bohensky et al. (2006) - ecosystem services in South Africa.	Discussions prompted by creating links between different components of the scenarios.
Brand et al. (2013) - understanding ecosystem services in the Swiss Alps.	Discussions arose from exploring issues of consistency and surprise in potential future states.
Fisher et al. (2011); and Swetnam et al. (2011) - ecosystem service analysis	Collectively thinking about the development of trends and drivers in alternative futures
in Tanzania.	encouraged discussions.
Mistry et al. (2014) - ecosystem management in Guyana.	Creating a 'platform for dialogue' stimulated discussions between participants from different
	perspectives.
Palacios-Agundez et al. (2013) - ecosystem management in Spain.	Discussions arose from exploring plausible futures and thinking about how to avoid challenges.
Plieninger et al. (2013) – managing ecosystem services provided by cultural	Discussion occurred through participants being provided with a structure, with which to explore
landscapes in Germany.	future possibilities and responses to challenges.
Henly-Shepard et al. (2015) - improving adaptive capacity to hazards in Hawaii.	Discussions prompted by considering responses to challenges.
Pearson et al. (2010) - sustainability planning in Australia.	Discussions triggered through developing and testing responses to problems.
Ravera et al. (2011a); and Ravera et al. (2011b) - climate change adaptation	Discussions encouraged by considering adaptation options in different scenarios.
in Nicaragua.	
Tschakert et al. (2014) - climate change adaptation in Ghana and Tanzania.	Discussions arose from combining experiences of everyday life with climate projections and
	anticipatory views of the future.
Wesche and Armitage (2014) - understand environmental change in	Structured discussions occurred regarding the implications of different drivers, on livelihoods, in
northern Canada.	alternative scenarios.
Sheppard et al. (2011) - climate change action and awareness.	Visual methods helped stimulate discussions by making potential climate impacts seem real.
Van Berkel et al. (2011) - rural development in Portugal.	Scenarios 'prompted' discussions about rural development issues.
Vermeulen et al. (2013); and Vervoort et al. (2013) climate change and	Discussions encouraged by exploring how to overcome potential future challenges.
food security in East Africa.	
Bohensky et al. (2009) - ecotourism in Papua New Guinea.	Considering what would influence the outcomes of 'guiding questions' stimulated discussions
	between participants.
Schulz et al. (2015) - Payments for Ecosystem Services in Brazil.	Discussions arose through deliberating on challenges and responses to them.
Brown et al. (2001) - marine protected area management in Tobago.	Discussions stimulated by deliberating on trade-offs of different options in the future.
Palomo et al. (2011) - protected area management in Spain.	Discussions encouraged through exploring trade-offs between different options.
Jessel and Jacobs (2005) - planning for the European Water Framework	Considering the effects of different policy options in different scenarios encouraged discussions
Directive	between participants.

Source: author construct



The data from the case review thus indicates that specific attributes of PSP processes prompted discussions that enabled learning to occur. These attributes included exploring the potential future development of different components in SES, and the interconnections between them, as well as considering of responses to SEPs, and the challenges that would have to be overcome.

This was reflected in my practitioner interviews, in which the 10 informants with whom I discussed how learning occurred, all alluded to specific attributes of PSP processes that supported learning, by prompting discussion between participants. This included the future providing a point of focus for discussion between different participants. For example, the experienced practitioner, Gavin, explained that PSP stimulates learning by:

"Getting people together, from different places, around common issues. Everyone is interested in the future, right?" (Gavin, 2016)

He thus indicated that the future, itself, is a common issue, that can bring different participants into dialogue. This was reflected in the response of another practitioner, Belinda, who had a wealth of experience conducting PSPs in global projects. She stated:

"Everyone has expectations, aspirations and anxieties with regards to the future, which they are forced to make explicit when they imagine scenarios." (Belinda, 2016)

This evidence shows, that deliberating on the future is said to be a specific attribute of PSP, that provides a common issue, around which participants can discuss their assumptions. However, as I discuss further in Section 5.5, 'common' issues are usually contested. Careful management is therefore required to ensure discussions about them are constructive.

Another practitioner, Rick went into more detail, regarding how he thought learning occurred in a specific PSP process in which he was involved. He explained that learning occurred, specifically, through focused discussions on potential future developments, which encouraged participants to reflect on their existing assumptions about SEPs:

"through focusing people's minds on what they think are the most important developments and trends... People know it, but people don't necessarily have a chance to focus on it and pull it together." (Rick, 2016)

In the FSF case study, eight of the 10 informants who stated they had learned through the workshop, explained that this learning had occurred through imagining alternative future possibilities in a structured way. The other two did not clearly articulate how learning had occurred for them. It may be that they had not reflected on how learning had occurred, but it is also notable that neither of them spoke English fluently, which could have limited their ability to articulate how they thought learning had occurred.

One of the participants, Alan, who did explain how he had learned, indicated that the structured way of thinking about the future in PSP had helped him to learn about different aspects of food and nutrition security. He also stated he had never participated in PSP before, and that this focused way of thinking about the future was completely new to him. The novelty of this way of thinking about the future is an important factor in how learning varies between different participants, which I discuss further in Chapter 6. Alan stated:

"The methodology of using scenarios, and the planning by using the backcasting, that was the most interesting part because really it was new to me... It facilitates somebody to go step-by-step... It is difficult to miss something, to overlook something." (Alan, 2016)

The above evidence indicates then, that PSP encourages learning by providing a point of focus, around which participants can reflect on their assumptions about the possible future development of SEPs. Furthermore, I found evidence that PSP creates opportunities for participants to think outside the box, or to 'stretch' their thinking. This was particularly evident in the PFSA case study, in which nine of the 13 participants I interviewed indicated they had learned through thinking about future possibilities in a creative, but focused way. For example, one participant, Penelope, stated, fluently:

"I think imagining different futures, or different realities, is really powerful, because you're starting from a place of possibilities. When you are thinking of different futures like that, when you're doing scenarios, you're provided an opportunity to be creative... scenario planning provides an opportunity to be strategic, to be creative, and to start from a place of possibilities." (Penelope, 2016)

Another participant, Gareth said a similar thing, but emphasised the importance of PSP for providing a structure within which to focus participants' creative thinking:

"It helped people to 'think outside of their boxes,' but within some particular parameters." (Gareth, 2016)

Likewise, the participant, Elliot, enthusiastically stated that imagining the future, together with other participants, had inspired creative thinking:

"We were allowed the space to be in an imagination, in a collective imagination, to let go of those conditionings [that constrain our thinking]. The scenarios methodology allows for using that part of your brain that sometimes gets 'cobwebby,' that you don't turn on the lights very often. It helps with that space where inspiration comes from." (Elliot, 2016)

In both the PFSA and FSF case studies, I also observed examples of how the specific activities and framings of the two workshops stimulated this focused, creative thinking. This is illustrated in Table 5.3, below.



Table 5.3 – Examples of specific activities and framings of the case study PSP processes, that stimulated creative thinking

Specific activities and aspects of the case studies	Explanation of how they appeared to stimulate learning
'Future Wheels' activity in PFSA	On Day 1 of the workshop, participants imagined desirable future conditions in which small-scale initiatives (or seeds) were 'mature conditions,' or mainstream ways of doing things. Participants identified the direct 'primary' impacts of these initiatives as mature conditions, and then the more indirect 'secondary' and 'tertiary' impacts. These conditions were presented as 'wheels' with the initiatives at the centre and the impacts spreading outwards (see Table 3.4 on p.60 for a more detailed explanation). Imagining desirable future conditions in the Futures Wheels provided the initial stimulus for participants to think creatively. For example, in one discussion group, the participants imagined a future in which the division between rural and urban spaces became blurred. In another group, the participants imagined how the effects of gene technology on human health could ultimately lead to immortality. The ideas generated in the Futures Wheels were subsequently expanded on and developed, in the later activities. The creative thinking, they fostered was maintained throughout the rest of the workshop.
Linking the 'Seeds' in PFSA	As outlined in Chapter 3, the 'seeds' were small-scale initiatives that can be described as conducive to creating just and sustainable futures. Each of the four discussion groups worked with three such initiatives, which they used to develop their scenarios. One thing I observed was that connecting the initiatives, and imagining the effects they could have on each other, helped participants to think creatively. For example, one discussion group connected an initiative involving artificial intelligence (AI) with another promoting more equitable and inclusive access in urban spaces. This led to them imagining 'fluid infrastructure,' in which urban infrastructure could physically change shape, to meet different purposes and, thus, encourage more equitable and sustainable use of space.
Testing responses to food and nutrition poverty in FSF	A key objective of the FSF workshop was to explore plausible future conditions and identify challenges and opportunities for food and nutrition security (FNS) in Tanzania. I observed that this aspect of the workshop prompted participants to think creatively about challenges and opportunities for FNS, as well as responses to them. For example, one discussion group came up with the idea of a 'taskforce' to help foster cooperation across different sectors dealing with food and nutrition issues.

Source: author construct



The observations I outlined in Table 5.3 corresponded with the responses of the participants I interviewed. One PFSA participant, Penelope, spoke about how imagining desirable futures in the 'Futures Wheels' had provided a stimulus for focused, creative thinking:

"It gave people a structure to push beyond where their thinking would normally take them... we did, in some ways, get beyond the standard ways of thinking about how things will evolve. It was a genuine shift in my understanding of what is possible." (Penelope, 2016)

Another PFSA participant, Dillan, indicated that imagining how different seeds could influence each other to bring about preferred outcomes, had enabled him to think creatively about possibilities for developing sustainable communities:

> "I was working with [a sustainable community initiative] that already exists, it is a little community. So, if you bring in a digital currency, and bring in that thinking there, you could probably use that as a springboard for similar communities." (Dillan, 2016)

The participant, Paul, indicated that the creative thinking fostered by these specific activities had been sustained throughout the workshop. He stated that this had been achieved through continuously adding to ideas about how the seeds could develop in a desirable way in the future:

"We would start with these seeds and then end up with complete scenarios. [For example] with the 'Artificial Meat,' we started off thinking about something in a test tube, and then we ended up resolving world hunger. So, something as small as that, could end up with major repercussions." (Paul, 2016)

In the FSF case study, two of my informants, Sally and Fiona, indicated that exploring plausible challenges, as well as responses to them, with others, had encouraged them to think creatively. Specifically, they described how they had identified the idea of a 'taskforce' to overcome the challenge of a lack of cooperation: "We were in a scenario, whereby there was no cooperation. So, when we discussed about the implementation of health programmes in schools, we discovered there was no cooperation, so we had to design a committee, which would be responsible to create that cooperation... There are challenges, but with ideas given by others then you get through." (Fiona, 2016)

"We said we need the task force to include different people with different backgrounds, from different sectors, because the issue of food security and nutrition is a cross-cutting issue." (Sally, 2016)

Another FSF participant, Tracy, also indicated that discussing how to overcome challenges in the scenarios had enabled her to think creatively. Specifically, she indicated that creative thinking had been stimulated by identifying responses to food poverty. She explained:

"[We thought about], what are the causes of food poverty in Tanzania, and how will those problems be overcome? Now what should we do to overcome those problems? We have to change the policy developments to create political will, so that food production and food security can be given the first priority. If the government will give first priority to the food production, then it would overcome that problem." (Tracy, 2016)

The above evidence shows that imagining future narratives of SEPs can provide a point of focus for discussions between different participants. This can take the form of both imagining how future conditions could develop in a desirable way and exploring plausible challenges and responses. This point of focus provides opportunities for participants to reflect on their assumptions about the future of SEPs, as well as to encounter the knowledges of others. PSP processes thus fit well with the concept of 'boundary objects.' As outlined in Chapter 2, these are phenomena, first proposed by Star and Griesemer (1989), that occupy several interacting worlds and remain relevant and outwardly acceptable to all of them. They can thus help to facilitate effective communication between diverse actors, as well as encourage the exchange of knowledge between them. As described by Star (2010), boundary objects can be concepts that people work towards and with, rather than material objects, or completed products.

My findings thus correspond with the assertion of Chaudhury et al. (2012), that PSP processes are boundary objects. PSP processes can provide a point of focus that stimulates communication between different participants. If each participant contributes their assumptions to imagining plausible events, conditions and trajectories, or to preferred conditions, PSP

processes can incorporate multiple identities and therefore remain outwardly acceptable to different participants. They can thus create opportunities for participants to exchange knowledge with each other. I therefore reason that PSP processes can represent boundary objects that encourage learning, through exposing participants to different knowledges, and encouraging them to reflect on their assumptions about the future.

My two case studies show that boundary objects in PSP can take different forms, depending on the framing of the workshops. In PFSA, the boundary object was the creation of scenarios that were mutually desirable for the participants involved. In FSF the boundary object was the creation of scenarios that included the assumptions and expertise of different participants regarding plausible developments in FNS. In both cases, opportunities were created for participants to exchange knowledge with each other, but in PFSA this occurred through participants contributing their ideas about how just and sustainable futures would look, whereas in FSF, it was through participants contributing their assumptions about plausible future developments in FNS.

My findings also show that PSP can stimulate creative thinking about the future, especially when it includes specific, carefully-designed activities that help to prompt and focus such thinking. As shown in the case studies, such activities can include exploring the development of initiatives that could help create just and sustainable futures, as well as deliberating on how to overcome potential future challenges. PSP can thus enable participants to 'push' their thinking beyond where it would normally go.

This links to the Zone of Proximal Development (ZPD), proposed by Vygotsky (1978). In Section 5.2, I showed that PSP can encourage interactions between different participants. The evidence presented in this section indicates that PSP can go further than just enabling interactions, because it provides a stimulus for focused, creative thinking. Thinking about the future in this way can thus galvanise participants to push beyond their usual range of thinking, which enables them to learn things that would normally be beyond their individual capacity. I therefore propose that, by stimulating focused, creative thinking, PSP processes can help participants to enter their ZPD. However, the prominence of specific activities, like the Futures Wheels in PFSA, highlights the importance of carefully designing PSP processes, such that they include different tools for creating and supporting structured, creative thinking. Facilitators can thus play an important role in learning, by designing and guiding participants through PSP processes.

My case studies indicate that this creative thinking can occur in both normative and exploratory approaches, but in slightly different ways. In the PFSA workshop, participants engaged in structured, creative thinking through imagining how small-scale initiatives could develop to produce desirable, just and sustainable future conditions. In the FSF workshop, structured, creative thinking was fostered through participants exploring responses to plausible challenges to FNS in Tanzania. It is evident then, that there are different ways in which structured, creative thinking about the future can be encouraged in PSP. As I discuss further in Chapter 6, the framing of the two case studies as either normative or exploratory influenced what participants considered relevant, as well as how facilitators directed participants' discussions, with potential repercussions for what is learned.

I also found evidence that learning occurred simply by being introduced to the concept of PSP. As I discuss further in Chapter 6, 12 of the 13 participants I interviewed in FSF stated they had learned a new way of thinking about the future, through being exposed to PSP for the first time. Equally, in Chapter 6, I also show that some PFSA participants learned just by becoming aware of others' perspectives on their fields of expertise. It therefore appears that these participants learned, without necessarily engaging in the creative, focused thinking, described above. Hence, it could be the case that, although this way of thinking stimulates learning, participants can still learn without it, through other aspects of PSP processes. This emphasises the importance of exploring how, and why, learning varies, which I do in Chapter 6.

5.4 Scaffolding

My findings, in Section 5.3, indicate that PSP can provide opportunities for focused, creative thinking. It can thus help participants 'enter' their ZPDs. However, Vygotsky (1978) adds that people require assistance from others, not just interaction, to effectively enter their ZPDs. I found that participants in PSP were helped to engage in focused, creative thinking about the future, through assistance from facilitators.

The authors of papers in the case review, typically gave little attention to the role of facilitators in the cases they described. However, in five cases, the authors indicated that facilitators designed PSP workshops to help participants engage in focused, creative thinking about the future. For example, Plieninger et al. (2013) state that: *'the workshops were prestructured regarding their form and central aims, but remained completely open for the*

participants regarding content,' (p.44). Hence, the facilitators appear to have focused participants' thinking, by providing a template, which the participants filled in. In another case, Bohensky et al. (2009) describe how the facilitators began the workshop by helping participants think of a 'guiding question' (p.2828). Participants were subsequently prompted by facilitators to think about the components of SEPs, that could influence this guiding question. The authors state that the guiding question was developed by the participants, but it is likely to have been influenced by the aims of the workshop, set by the facilitators. The influence of facilitators on shaping learning is explored further in Chapter 6.

In the practitioner interviews, 13 of my 16 informants, acknowledged the importance of facilitation for guiding participants through PSP processes. One practitioner, Vera, who was a highly experienced professional facilitator of futures-thinking methods, emphasised that facilitation is a key condition for learning in PSP. She acknowledged that participants can find it difficult to think about the future, in the way that PSP proposes, which means facilitators need to help ease them into it:

"The challenge is to get people to engage with these scenarios... The important thing is to get participants to at least entertain the idea [of thinking about possible futures] and play with it. The role of the facilitator must be to pick up on what incremental changes people are willing to consider in the future, and build on those." (Vera, 2016)

Another practitioner, Rick, described a specific example of how he had helped different participants to engage with PSP in a project he facilitated. He emphasised the importance of designing PSP processes to include different strategies that could help diverse participants engage in PSP:

"I think having different methods of engaging people is really important. So, you try to capture different kinds of knowledge that are within the different ways that people think... The visual methods helped [because] sometimes people can't express things in words, but they can do that in visual forms... So, using a whole load of methods helps to bring more information." (Rick, 2016)

The role of facilitators in supporting participants to engage in PSP was demonstrated strongly in my two case studies. In PFSA, Mavis, the lead facilitator, provided a particularly good example of how a facilitator could guide participants through a workshop. She was a highly experienced facilitator of futures-thinking methods, and had designed the structure of the PFSA

workshop, in collaboration with the other facilitators. During the workshop, she assisted participants in two main ways: Firstly, she provided an overview of the entire process, before explaining each activity, in turn. This included providing visual demonstrations of the activities. For example, when she demonstrated the 'Futures Wheels' activity, described in Section 5.3, she used an example of how the primary, secondary and tertiary impacts of 'Solar Tiles' could develop in the future. Similarly, when she described the '3 Horizons' activity, described in Table 3.4 (p.55), she illustrated the process, using a diagram that visualised the trajectories of the different 'Horizons.'

In my interviews with PFSA participants, they indicated that these explanations had helped them engage in the process. For example, Dillan explained:

"[The activities] were nicely set out, so you understood where you were going to, clearly... It was good to have it just in bitesize chunks." (Dillan, 2016)

Similarly, Geoffrey described how he had felt confused prior to the workshop, but Mavis' explanations had helped him engage in it:

"Initially I was a little bit lost. When I read the outline of what we were going to do, I didn't get a clear sense, until we met in person. We then had this outline that Mavis made and that schematic. I think that schematic helped me get a much better sense of what was expected and what exactly was needed in the scenario planning process." (Geoffrey, 2016)

Secondly, in addition to explaining the process, Mavis roamed between the four discussion groups, providing clarification and advice on the activities, to help participants progress through the workshop. For example, I observed an instance when one of the discussion groups were unsure whether to connect the 'Futures Wheels' with another activity, the 'Cross-impact Analysis,' (described in Table 3.4 on p.55) in which they explored how different initiatives could connect to and influence each other. The group were concerned about losing the connections they had already made, between different aspects of the Futures Wheels. Mavis suggested they do the 'Cross-impact Analysis' separately but explained how they would link the activities together at a later stage. One participant, Gareth, who had been in that group, thus explained that Mavis helped: "ccalm things down when it got confusing."

The fact that Mavis explained and clarified the process made it easier for the other facilitators to concentrate on encouraging discussions between participants, in the discussion groups. I observed that they did this by asking questions and prompting participants to discuss

specific points. For example, the facilitator, Pamela, asked participants to think about where people would live, and how, in the world described by her group's scenario. This helped the participants to imagine future conditions in great detail. Another facilitator, Doris, encouraged participants in her discussion group to consider the divide between rural and urban spaces, and how it might change in their scenario. This prompt helped to stimulate the creative thinking about 'blurring' the urban-rural divide, described in Table 5.3.

The role of the facilitators was reflected in my interviews with participants. For example, one informant, Miriam highlighted how the facilitator in her group had prompted discussions by asking questions about how the 'seeds' could develop:

"She was just bringing questions in, like: 'okay what's next, and what's next, and what's next?' 'What are the limits of this?' 'Do you think this is bad?'" (Miriam, 2016)

Miriam thus indicated that the facilitator created a *"space for discussion."* It is evident then, that the PFSA facilitators provided assistance that helped participants to engage in the PSP process. Mavis explained and clarified the specific activities, while the other facilitators stimulated discussions between participants, using prompts and questions.

In the FSF case study, although the facilitators played essentially the same role as in PFSA, they were limited by only having three facilitators, as compared to five in PFSA. This meant the lead facilitator, Mike, had to play multiple roles, including: explaining the process, prompting discussion in one of the discussion groups, and roaming between the groups to check on progress and provide clarification. He was also constrained by the lack of time available for the workshop, which meant he had to explain things very quickly and was unable to provide demonstrations like Mavis in PFSA. At the end of the workshop, he reflected to me that it had been the least-funded, shortest workshop he had ever done. He felt that this had limited his ability to roam between groups and make sure they were engaging with the activities. Indeed, when I interviewed one participant, Fiona, she explained that not having Mike roaming around had sometimes been frustrating, because it took a while to get help when they required clarification on specific activities:

"But he was so busy, of course he apologised to us sometimes [because] he could not come in time." (Fiona, 2016)

Another challenge for the facilitators in the FSF workshop was that one of the three facilitators, Thomas, had never facilitated a PSP process before. Thomas would therefore have

benefited from more input from a more experienced facilitator, but because Mike had little time to roam around and provide clarification, it was difficult for him to provide this support. Thomas' discussion group therefore appeared to regularly become confused and struggle to make progress on the workshop activities. Two of the participants I interviewed, Keith and Mark, had both worked in two discussion groups (as the groups were mixed-up half-way through the workshop), including the one facilitated by Thomas. They reflected that Thomas had been limited by his inexperience. Keith explained that Thomas had thus been less able to encourage participants to contribute their knowledge to the discussions:

"I think facilitator matters. I know the other facilitator, he is very much experienced, so he knows how to 'pick' things from out of people, but in this group, you could see, he is not much experienced of scenario-creating things." (Keith, 2016)

It is evident from the above findings that skilled facilitation can help participants to engage in PSP processes. The support provided by facilitators to participants can be explained as 'scaffolding,' a concept put forward by Wood et al. (1976). As I set out in Chapter 2, Wood describes scaffolding as assistance, provided by experts in a given topic, to help non-experts achieve learning. The idea of scaffolding is commonly linked to the zone of proximal development (ZPD) (Vygotsky, 1978). Fernández et al. (2001) explain that scaffolding structures and guides participants, through tasks that help them to push their thinking 'outside the box,' in a controlled way. This helps participants to enter their ZPD and achieve learning they would not have achieved alone.

The learning scholars, Van der Pol et al. (2010) state that scaffolding typically occurs in three stages: contingency (tailoring support to learners' existing ability), fading (decreasing the level of assistance as the learner becomes more competent), and transfer of responsibility (for learning from the expert to the learner). This is evident in the assistance provided by Mavis to the participants in PFSA. She provided contingency by designing, as well as explaining the activities to participants, in a way that corresponded with their existing ability to engage in PSP. She then provided fading, by clarifying the activities and advising participants on how best to tackle them. Ultimately, she decreased the level of assistance, until participants had responsibility for their engagement in each activity. Assisting participants to engage in carefully-designed sets of activities, can thus stimulate focused, creative thinking about the future, which encourages learning.

As Fernández et al. (2001) explain, scaffolding can also occur in the form of 'exploratory talk,' which typically occurs between peers, rather than between an expert and a non-expert. In exploratory talk, people propose new ideas and receive critical and constructive feedback from others. My observations of the two case studies show that facilitators proposed new questions and ideas to get participants thinking and discussing. These prompts and questions could be viewed as attempts to stimulate such exploratory talk. In this sense, the facilitators provided scaffolding as peers involved in the discussion with participants.

I conclude, therefore, that facilitators of PSP processes play an important role in enabling learning, specifically by providing scaffolding. This includes explaining and guiding participants through carefully-designed sets of activities in PSP processes, and prompting discussions. In this way, facilitators can enable learning, by helping participants to engage in creative, focused discussion. However, even with assistance from facilitators, there exist several challenges and pitfalls that can limit participants' engagement in PSP processes, and thus hinder learning. I explore these further in Section 5.5 below.

5.5 Conducive conditions for learning in PSP

As Henrichs et al. (2010) state, in their guide concerning how to set-up PSP processes, there is no definitive approach to conducting a PSP process. Instead, they assert that individual processes should be designed according to their specific contexts, objectives, and participants. Nevertheless, Henrichs et al. set out some general principles that they believe should be considered in PSP processes, to ensure a range of outcomes. In this section, I also lay out a set of conditions that my data indicate are important considerations in PSP processes. In contrast to Henrichs et al., however, I concentrate specifically on the conditions that appear most conducive for learning.

5.5.1 *"Everyone felt very sharply distinct" -* Diversity of participants is conducive to learning

In Sections 5.2 and 5.3, I showed that learning in PSP can occur through interactions between different participants. Collectively thinking about the future can expose participants to different knowledges and encourage them to reflect on their assumptions about the future. Unsurprisingly, therefore, I found that learning is encouraged by including a diverse cohort of

participants in PSP. As I stated in Chapter 4, 23 of the 30 cases in the case review assumed that bringing different participants together to deliberate on the future could enable learning to occur. Similarly, in the practitioner interviews, there was an assumption, implicit in my informants' responses, that the diversity of participants was an important enabler of learning. For example, Gordon stated that learning was stimulated by: *"interactions across disciplines."* Terry stated it was encouraged by: *"sensitisation to multiple perspectives."* Gavin indicated it was catalysed through: *"bringing people together, with people they wouldn't normally interact with."* Vera, who had decades of experience facilitating PSP processes, described: *"including a diverse range of perspectives,"* as a key condition for learning to occur.

A particularly clear example, of how a high degree of diversity can stimulate learning, was provided by the highly experienced practitioner, Deborah. She attributed learning in a major global PSP process, in which she took a lead role, to disagreements that arose between participants from different academic disciplines:

"We had close to a day of pretty acrimonious disagreements, especially among some of the younger ecologists with the young economists. [Another facilitator] and I had to just have a series of individual conversations with people that said 'hey, cool it,' we all need to work together to get this done... But actually, when there is a genuine disagreement among experts that generally means that there's a real uncertainty there that should be noted." (Deborah, 2016)

This description shows that the differences between the participants were valuable for learning, in that their disagreements could result in the identification of uncertainties. This indicates that the greater the diversity of participants, the more likely they are to learn through exposure to different knowledges.

This was especially evident in the PFSA case study. All 12 of the informants, who claimed to have learned through the workshop, indicated their learning had been encouraged by the diversity of participants. For example, one participant, Geoffrey, stated that the diversity of participants had encouraged learning, in a way that other participatory meetings he attended had not: "I go to a lot of meetings and it's not always such a diverse group... in my opinion, the more we include people that we normally don't talk to, the better... because what that did was to challenge our everyday understanding of these issues." (Geoffrey, 2016)

Similarly, the participant, Penelope, who had prior experience of other PSP processes, indicated very clearly, that distinctions between the participants had encouraged learning, for her. She reflected:

"Everyone felt very sharply distinct... I learned from every single person in that group."

"The organisers should be commended, for trying to bring together people that do have different views, and I don't think these processes are very valuable if everyone can just be happy and jolly and all get along." (Penelope, 2016)

It is evident then, that a diverse cohort of participants is believed to encourage learning, by increasing the different knowledges, to which participants are exposed. This corresponds with Henrichs et al. (2010), who assert that the participation of a wide range of stakeholders is vital for PSP processes that aim to produce learning. They emphasise that, when designed and facilitated effectively, processes that include diverse participants can foster learning about the complexity, as well as the dynamics of social-ecological systems.

However, as indicated by the practitioner, Deborah's example, including a diverse cohort of participants also has pitfalls, which can limit or even prevent learning in PSP. She indicated that, although differences between participants ultimately resulted in learning, this required careful management by facilitators, and a willingness to cooperate by participants. As I discuss in the following sub-section, disagreements between participants can become counterproductive.

5.5.2 "We hit a dead stop." - Challenges of including diverse participants

In the practitioner interviews, three of my informants described how they had conducted PSP processes in situations where there was a significant amount of conflict between different participants. In these instances, the practitioners each chose to keep different groups of participants separate, to avoid exacerbating the existing tensions between them. The respective practitioners therefore created scenarios in separate workshops, with different groups of

participants, and then synthesised them to create scenarios that represented the inputs of different groups. One of these practitioners, Greg, justified this, saying:

"You know, if you put people who have very different perspectives together, I don't think that's a problem. It's just more about, there are some tensions and some divisions in that community and I didn't want to exacerbate those... I wanted people to express their perspectives in a way that they felt free and not under fire and not judged... [If they had been in the same room], I don't think that people would talk freely. They would be very concerned with what they say to each other... I think people were free to talk with just me listening." (Greg, 2016)

Similarly, Ronald justified keeping participants separate, in his case, by saying:

"I mean we didn't bring them together in the same room, because to draw on their experiences and integrate them, and treat their values and perspectives equally would have been difficult if they had been together." (Ronald, 2016)

Greg and Ronald clearly believed that the conflicting interests of different groups would have created counterproductive tensions, if they had interacted face-to-face. As such, they thought participants would have been unable to make their assumptions about the future explicit, which could have limited learning.

This was reflected in one instance in the PFSA case study, in which disagreements in one of the discussion groups appeared to result in counterproductive tensions, that hindered learning. Three of the four informants I interviewed from that group complained that their interactions with other participants in the workshop had been too intense. One participant, Morty, was especially vocal about this. He stated that the intensity of the interactions had left him feeling exhausted, and at one point he had been forced to withdraw from his discussion group for a rest. He therefore indicated that he would have maintained a higher degree of enthusiasm for the workshop if there had been more time for individual reflection:

"It was just too much! Too much talking time and not enough integrating time... I would have, for practical, and for process reasons, I would've had more alone time." (Morty, 2016)

Another participant, Dillan, who had been in the same discussion group as Morty, indicated that the interactions had been too intense. He explained that this intensity had led to
the group becoming 'fatigued,' which resulted in tensions between the group members. He went on to explain that these tensions brought the discussions to a complete halt:

"I think towards the end, we were really fatigued, and that's when tempers started boiling. We were not getting anywhere. We had just come to a dead stop." (Dillan, 2016)

The fact that discussions came to a 'dead stop,' implies that the interactions in that group were not conducive to the creative, focused thinking that appears to stimulate learning. Interestingly, I observed that the participants in that discussion group were all very forceful in the way they articulated their ideas. They were also prone to criticising the ideas of others on a very detailed, technical level, which became contentious. It could therefore be the case that these interactions were less conducive to learning, because the disagreements became counterproductive. Indeed, another participant from the same group, Darren, reflected:

"Within our group, I thought that, at times, the disagreement was counterproductive... Sometimes people were so insistent in what they believe and what they do, that they were not very good at listening, and I think there was a lot of missed opportunity, because people were so headstrong." (Darren, 2016)

This example appears to correspond with the assumptions of the practitioners, Greg and Ronald, that disagreements can be counterproductive to learning in PSP. In the guide set out by Henrichs et al. (2010), the authors also recognise that the different interests of participants can lead to conflict. Furthermore, they indicate that conflict can limit the achievement of desired outcomes. This seems at odds, however, with my findings in Sections 5.2 and 5.3, that exposing participants to different knowledges is integral to learning. Hence, there appears to be a tension between exposing participants to different knowledges and avoiding counterproductive tensions.

However, the fact discussions in the above discussion group came to a "dead stop," was, arguably, just as much about the personalities in the discussion group, as it was about the disagreements between them. Indeed, in the account of Darren, he states that participants were "not very good at listening," and were very "headstrong." This reflects another challenge, associated with including diverse participants, that some participants are, inevitably, more vocal and assertive than others.

This was demonstrated in the PFSA case study, in that not even one of the participants mentioned the plenary sessions, when I asked about which aspects of the workshop had

encouraged learning. One participant, Paul, who described himself as an introvert, explained this, when I interviewed him. He indicated that interactions in plenary, were difficult for more introverted participants, because the plenary sessions tended to be dominated by the most assertive participants:

"When we were all together again, certain people would always be talking. I mean, not always, but there were always certain voices, and then the more introverted people would speak more in their [discussion] groups... In the large group, you start to hear the same things being repeated." (Paul, 2016)

Similarly, another participant, Penelope, complained that the plenary discussions had been dominated by radical ideas from the most assertive participants that 'shut down,' the discussion:

"I did find it challenging in the larger group discussions, when some participants would be very strong on ideas, in a way that threatened to throw the process away. That whole digression, I felt like [it] shut down the interesting direction that conversation could have gone in terms of thinking about the future, and instead made it into a polemic." (Penelope, 2016)

This indicates there was less opportunity for participants to contribute and be exposed to different knowledges, and thus less opportunity for learning to occur.

It is evident, therefore, that learning in PSP can be hindered by assertive participants, creating counterproductive tensions, especially if facilitators are not alert to this. Hence, it seems that certain types of interactions are more conducive to learning than others. Diversity and disagreement appear to be important, but can become counterproductive, if they are not managed effectively. This links to learning literature, regarding 'exploratory talk.' As described in Section 5.4, Fernández et al. (2001) describe exploratory talk as a form of scaffolding, that can help people enter their ZPDs, and thus learn. Fernandez et al. emphasise that any criticism of others' ideas should be constructive for such exploratory talk to occur. Similarly, Barnes (2008) highlights that for exploratory talk to be effective, it is vital that learners feel at ease, and not in danger of being aggressively, or disrespectfully criticised for the ideas they express. The importance of respectful interactions is also acknowledged by Henrichs et al. (2010). They advise that, to avoid counterproductive tensions in PSP, it is vital that facilitators carefully manage interactions between diverse participants. The importance of facilitators, in this respect, is reflected in my findings. I explore this further, in the next sub-section.

5.5.3 "You need a very good facilitator to come in and make sure that everyone has a voice." - Importance of facilitators for managing interactions

The evidence presented in this section thus far, indicates that interactions between different participants need to be carefully managed, if they are to result in learning. As I set out in Section 5.2, the experienced practitioner, Gordon, specified that learning occurs through participants questioning each other's assumptions *"in a respectful way."* As indicated above, skilled facilitation is, thus, required to ensure interactions between participants are 'respectful.'

In the case review, I did not find evidence of the role played by facilitators in managing interactions between participants. This may be because editorial constraints meant they were unable to discuss this in their papers, rather than because they did not consider it an important issue. Indeed, in the practitioner interviews, 10 of the 13 informants, who discussed the role of facilitation, acknowledged that facilitation was vital for managing interactions between different participants. For example, one practitioner, Barry, who had a wealth of experience conducting PSP processes with diverse groups of stakeholders, emphasised the importance of facilitation for ensuring different participants are able to contribute in PSP:

"When it comes to doing the participatory work, that's when you need a very good facilitator to come in and make sure that everyone has a voice, because there's always going to be people who dominate." (Barry, 2016)

An example of effective facilitation was provided by the experienced practitioner, Deborah, in which effective facilitation, apparently turned potentially counterproductive disagreements into valuable learning points. She described how her role, as a facilitator, had helped with this:

> "My job was a mediator, basically [I said] 'look guys, I realise there's a disagreement about how to do this, but the alternative is, we can either make a decision and do something, or we can do nothing, and doing nothing looks like a really bad idea so, let's find a way to muddle through here.' We had few disagreements we really had to manage, and we managed it, so in the end it came out well." (Deborah, 2016)

In this instance, the facilitators reportedly encouraged divergent participants to work together, rather than allowing a complete break-down in discussions. The facilitators appear to have

encouraged participants to question each other's assumptions, in a respectful way, which enabled learning to occur.

Another experienced practitioner, Mary considered facilitation to be a "hugely powerful" tool for managing interactions between different participants. In contrast to Deborah's success story, Mary illustrated the importance of facilitation, using an example in which poor facilitation resulted in a complete break-down of discussions. In this instance, the organisers had been unable to hire the professional facilitator they originally approached, forcing them to use someone else. When the workshop started, the facilitator did not effectively manage the relationships between different participants. This led to a situation where the disagreements between participants just led to counterproductive arguments:

"The process started, and she just stood there, she didn't say a word, she just stood box-still and let all these people argue with each other, and people were getting upset, people were shouting at each other, voices raised. It was all the same old arguments coming out. The break time went, the break time was over, and they were still arguing, people were getting upset, you know, someone had stood up now and was shouting at someone." (Mary, 2016)

It is evident then, that practitioners recognise that skilled facilitation is required, to ensure interactions in PSP are respectful. However, the three above accounts came from the practitioners who facilitated the very workshops they described. It is therefore possible they emphasised the importance of their own roles, as facilitators. My interviews with participants in the two case studies provided valuable insights on facilitation, from the perspectives of the participants.

In the PFSA case study, I observed that the facilitators did not begin the workshop with conventional, formal introductions. According to the lead facilitator, Mavis this was *"deliberately engineered,"* to avoid participants introducing themselves based on their titles and the organisations they represented. They expected this would help encourage more equal interactions between participants and avoid domination by the participants with the highest credentials, or from the most influential organisations. In six of my interviews with participants, they indicated that this had worked. For example, Penelope indicated that this made it easier for participants to interact with each other:

"I think one of the best things that the organisers did, is that they didn't have this whole thing of telling everyone who you are, focusing on your credentials,

or creating power imbalances. It was like, throw everyone in a room who know almost nothing about each other and just go for it, and that, I think was an immense strength, because it meant that we could connect to participants on a human level." (Penelope, 2016)

This indicates that the PFSA facilitators effectively enabled participants to interact in a respectful way. However, this does not appear to correspond with my findings, presented earlier in this section, that the plenary discussions were dominated by the most assertive participants. Equally, it is at odds with my finding that one discussion group was hindered by counterproductive arguments that resulted in a 'dead stop.' Indeed, Dillan, who had been in that group, thought the facilitator had been too weak:

"I think she was maybe not strong enough to keep everyone in check. I felt sorry for her in our group. Sometimes I just wanted to stand up and say: 'enough now! Just listen!' She certainly had her hands full with us." (Dillan, 2016)

My findings from the FSF case study also showed that the facilitators were instrumental in managing the interactions between participants. Four of the FSF participants who I interviewed, reflected that the facilitators had prevented the discussions from being dominated by any individual, or group of, participants. For example, when I asked Tracy about whether she thought any participants had contributed more than others, she responded:

"The facilitator made sure every one of the members of that group participated. Therefore, I can say that he did not segregate. He was just helping all of us. That is why it is very difficult to find the one who contributed most, and the one who contributed least. So, everyone contributed well." (Tracy, 2016)

This indicates that the facilitators' approach of encouraging every individual to contribute, helped to minimise the domination of discussions by any individuals, or groups of, participants.

It is evident then, that facilitators play an important role in managing the interactions between different participants. This matches the assertion of Henrichs et al. (2010), who state that skilled facilitation is needed to moderate discussions, as well as encourage effective participation by all participants. They emphasise, specifically, that facilitators need to deal effectively with assertive participants, and capitalise on diversity in participants' worldviews, whilst ensuring that discussions remain constructive. The importance of skilled facilitation is also

reflected on by Reed (2008) in his paper on best practice in participatory processes for environmental management. He contends that the facilitation of a participatory process is critical in determining whether it achieves its desired outcomes. He specifically states that facilitators need to be capable of managing conflicts and dominant participants, all the while ensuring that the interactions between different participants remain constructive. This, in addition to the role of facilitation in providing 'scaffolding,' as described in Section 5.4, emphasises the vital role played by facilitators, in enabling learning. It is concerning, therefore, that there was limited information in the case review regarding the role of facilitators.

5.5.4 *"Scenario does not exist in Swahili."* – Some participants are less able, or willing to engage in PSP

In the above sub-sections, whilst the data show that learning can be hindered by counterproductive disagreements, they also demonstrate that the types of interactions, that unfold in PSP, are not conducive to the learning of all participants. As described above, the apparent intensity of the discussions, in PFSA, seems to have contributed to the counterproductive disagreements, in at least one discussion group. Equally, the plenary discussions seem to have lent themselves to domination by more assertive participants. This could be because some participants found certain types of interactions more conducive to learning than others.

This was illustrated strongly by an example provided by the experienced practitioner, Mary. She described a situation in which three participants arrived at a PSP workshop, uninvited:

"We'd invited 12 people, but actually there were 15 people. There were three guys sitting at the end of the table who I didn't know. I made the mistake of not working out who they were and either chucking them out, or at least trying to get a handle on who they were." (Mary, 2016)

When Mary asked the participants to engage in an activity, involving writing things on posters, the three uninvited participants did not do so. She went on to describe how these three participants had explained they were unable to engage in the task, because they were illiterate:

"So, we did that, and these three guys just weren't doing it, so I went and said: 'so you do this and this, and then you stick it there,' and they said 'yeah, we get it'. So, I said 'why are you not doing it?' At this point, all eyes were on these

three guys because they still hadn't done it and I was asking them 'why can't you do it?' And one of them admitted 'we are illiterate, we can't read and write.'" (Mary, 2016)

In this instance, the participants were unable to engage in in a specific activity in PSP, because of their level of education.

In the PFSA case study, in which most participants had a similar level of education, one participant, Gareth, indicated that he had been more comfortable with certain types of interaction than others. Specifically, he stated he had learned more, through interactions in the informal periods, like tea breaks and meal times, than the focused discussions in the workshop itself. He explained that he had learned about how a particular initiative could develop, in the future, to assist marine conservation:

"I found out about the benefits of the gene technology, like it could help to resurrect destroyed ecosystems, like coral reefs. But that came up in a discussion with another participant, at dinner, not in the group discussion itself." (Gareth, 2016)

In this instance, the informal discussions appeared to be more conducive to Gareth's learning than the focused discussions in the workshop. However, it is unlikely that the informal conversation, leading to this learning, would have happened if Gareth had not been engaging in a PSP process that involved imagining the development of initiatives like gene technology. Gareth's learning was arguably stimulated by the focused, creative thinking in PSP, even though he attributed it to an informal conversation.

In FSF case study, I found evidence that some participants were less able and/or willing to engage in PSP overall, because it represented a new and unfamiliar way of thinking about the future. In one instance, I observed that a participant, Erica, raised moral objections to the process. She contended, during the first day of the workshop, that PSP equated to *"gambling."* She misinterpreted the role of PSP, as making definitive decisions, based on predictions, rather than exploring alternative, plausible futures. The facilitators tried to reassure her that the idea was not to predict the future, but Erica remained unconvinced. She did not attend the second day of the workshop, which could indicate she was unwilling to continue with the process. Unfortunately, due to illness, she was unavailable for me interview and hence, I was unable to ask her about this.

In the FSF workshop, it took noticeably longer for the participants to grasp the process than it did in PFSA. In the workshop, I observed how the facilitators had repeatedly to explain the

idea of PSP, to confused participants. Four of my interviewees also reflected they had found it difficult to understand. For example, Diane reflected that PSP had been *"unclear,"* to her, while Keith stated, he had found it *"difficult."* The other 10 interviewees did not explicitly state they had found it difficult, but this may have been because they did not want to be critical of the organisers.

The difficulties that FSF participants appear to have faced in engaging with PSP, may be explained by one participant, Sally, who indicated that PSP was a very new and unusual activity for the participants:

"It is a paradigm shift! Scenario planning is new to us." (Sally, 2016)

Indeed, when I asked one of the facilitators, Thomas, about challenges in the workshop, he indicated that the concept of 'scenarios' did not even translate into the local language:

"Some words are not there in Swahili, like 'scenario.' 'Scenario' does not exist in Swahili." (Thomas, 2016)

It is evident, therefore, that the way of thinking required by PSP was very unfamiliar to participants in FSF, which made it difficult for them to engage in the process.

In complete contrast, one PFSA participant, Barbara, indicated she had been less engaged in the PSP process because she was *too* familiar with it. She was an academic who specialised in thinking about future social innovations, and was thus accustomed to thinking about different possible futures in processes including PSP. She therefore stated that she had not learned through her participation in the workshop:

"You know Sam, this is my job. I do this every day. It was a great event, but it hasn't been an eye-opening experience for me... This is what I do in life." (Barbara, 2016)

However, Barbara was not the only PFSA participant who was familiar with PSP. In fact, Penelope also had prior experience engaging with scenarios, but enthusiastically indicated that she had learned much from the workshop. Barbara therefore appears to have been less open to engaging in the specific PFSA workshop than other participants. I explore why this might be the case in Section 5.5.5, below.

The above evidence shows that some participants are less able, or willing, to engage in PSP processes than others. Henrichs et al. (2010) also acknowledge that different groups of

participants have different levels of capacity to think in an open-minded way about the future. They subsequently advise that facilitators should make efforts to enable all participants to engage in PSP, through clearly explaining the process and building trust with participants. However, in FSF I observed that, despite the facilitators' best efforts, the discussions in one group appeared to marginalise some participants. In this group, there were three female participants, who were younger than the other members of the group. They were mostly very quiet and only spoke, briefly when they were asked to. Hence, they appeared quite marginal to the group's discussions. This may be because of traditional gender roles in Tanzania, that mean women typically remain quiet in the presence of older men (Verma, 2001). This indicates that, despite the best efforts of facilitators, some participants were still less comfortable, perhaps largely because of cultural reasons, with engaging in the types of interactions that unfolded in the workshop.

It is therefore likely that the effectiveness of PSP as a method for learning, varied between different participants. This raises important questions regarding for whom PSP is meant to be beneficial. Is the main intention for everyone to learn from it? Alternatively, as implied in Chapter 4, is the primary intention to elicit stakeholders' knowledge, for research purposes, with learning as a secondary objective? One further issue, that can affect the efficacy of PSP for different participants, is the extent to which participants have relevant expertise. I explore this further below.

5.5.5 *"I want to be operated on by a surgeon."* – The importance of relevant expertise

In the practitioner interviews and the two case studies, my data show that to encourage learning, it was important for participants to have relevant expertise in the topics that specific PSP processes focused on. One practitioner, Mary, emphasised the importance of carefully selecting participants with relevant expertise. She was an academic, with expertise in participatory processes of knowledge-sharing across different groups of stakeholders. She described a PSP process, in which she had conducted a 'stakeholder analysis,' to help select participants with expertise and influence regarding particular SEPs: "We used these matrices, where you're asking for a level of interest and influence, which you might rate as high or low. To ensure that we were representing each of the different categories, and to identify the groups that were most marginal to the network and ensure that they were included as well." (Mary, 2016)

She was confident that, in doing so, she ensured participants had diverse, but relevant expertise with regards to the SEPs considered in her project. Indeed, she went on to critique other projects for not doing likewise:

"I believe it [stakeholder analysis] is a step that is very rarely included in scenario analyses, and bizarrely is rarely included in any participatory process. People think they've done it, but it's very rarely systematic." (Mary, 2016)

The importance of ensuring participants have relevant expertise was emphasised by one participant, Barbara, in the PFSA case study. When I asked her about what she had found challenging in the workshop, Barbara, an accomplished social innovation scholar, bemoaned what she perceived as a lack of expertise among the other participants:

"What I found most challenging was, to deal with the ignorance. By that, I mean technically, literally ignorance, not in an offensive way. I mean, at times, you have to strike a good balance between the diversity within the group, and the knowledge... When it comes to knowledge, I want to be operated on by a surgeon, who knows exactly what she or he is doing. I don't think we should have equal say when it comes to certain things. A lot of people [in the workshop] perceived that all opinions have the same weight, and everything is equally important, and often this is not the case. You either know something, or you don't. You cannot fabricate it. Managing the trade-off between the democratic participation in the process, and the knowledge-driven part of the process is not easy, so it needs to be carefully managed." (Barbara, 2016)

Barbara clearly felt the 'knowledge-driven' part of the workshop had been compromised by participants with limited technical expertise. However, she later appeared to undermine this assertion by stating the workshop had been valuable for her to gain feedback from experts on her ideas about the future. She illustrated this with an interesting example about pizza-making:

"It's like if you know how to make pizza, and then you meet another pizza guy, and you share a lot of notions that, even if you knew how to make pizza really well, you know even better now, because a couple of tips have come out of that conversation." (Barbara, 2016)

To get 'tips' on pizza-making, the other 'pizza-guy' presumably needs to have some technical expertise in making pizza. Barbara's argument that the other PFSA participants all lacked relevant expertise may have thus been inaccurate.

The data from the FSF case study indicates that the relevance of participants' expertise is important. My interviews with FSF participants and facilitators indicated that participants' lack of expertise was a significant constraint on the process. For example, one of the facilitators, Thomas, showed some frustration with the level of participants' expertise:

"Yeah, in the group I would say that... they needed a lot of guidance, these people... There was a lot of knowledge limitation... In terms of participants who had little knowledge, of course I had to educate them, I had to elaborate theories, [so that] they were able to engage in the discussions." (Thomas, 2016)

Similarly, three of the FSF participants I interviewed, also stated that they found other participants' lack of expertise to be a challenge. Two of them, Beth and Tristan, worked for government ministries. The third participant, Sally, was a doctoral student who was studying international development. All three had substantial technical expertise in food and nutrition security. Tristan contended that the discussions in the workshop had been compromised by other participants' lack of expertise on food security:

"I just came to realise that a lot of people are talking about food security, but are not really grounded in food security... People [didn't] understand a lot about food security, because even when I was trying to talk about food availability, accessibility and utilisation, people [didn't] understand... So, if people were grounded really in food security, the discussion could have even been alive." (Tristan, 2016)

This was echoed by Sally, who indicated that the lack of expertise slowed the discussion down, which made it less intellectually stimulating:

"Some of the members were trainees... they were not informed much with the prevailing situations... It slowed it down, because we had to explain to each other at least 'what do we mean by this?" (Sally, 2016)

This shows that the inclusion of participants with relevant expertise is an important factor in enabling learning. This echoes the finding, presented in Chapter 4, that including relevant participants is vital for implementing responses to SEPs. This finding also corresponds with the guidance of Henrichs et al. (2010), who advise that participants should be selected such that they represent the interests and worldviews of broader stakeholder groups, who are affected by the topic that is deliberated on. Reed (2008) goes further than this, to contend that participants should be selected systematically using specific tools like 'stakeholder analysis.' He reasons that such systematic selection can ensure participants have expertise that is relevant to the SEPs being considered.

However, although three informants thought that the FSF workshop had been limited by a lack of relevant expertise, the evidence presented in Chapter 4, and earlier in this chapter, indicates that learning still occurred. The relevance of participants' expertise thus appears to impact on the learning of individuals, but not necessarily on whether learning occurs overall. The participants who complained about a lack of expertise may have learned less, or learned something different from what they expected, because they did not consider the different knowledges they encountered relevant to their interests. This emphasises the importance of considering how and why learning varies, which I discuss in Chapter 6.

5.5.6 Space, time and resources

The final 'conducive conditions,' for learning in PSP, concern the physical space, time, human and financial resources available for specific PSP processes. To do this, I compare the FSF and PFSA case studies. The physical space in which the workshops took place differed significantly. I illustrate this in Table 5.4, below.

Table 5.4 – Comparison of space, time and resources available in the Positive Futures

Conditions	Positive Futures for Southern	Food Security Futures
	Africa	
Physical space	Four large, well-lit, well-ventilated	Single, cramped, uncomfortable room.
	rooms in a five-star hotel.	Resulted in discomfort for participants
	Participants and facilitators also	and facilitators, which made it difficult
	had access to the extensive hotel	to focus on the workshop activities.
	grounds, swimming pool, and	
	sports facilities. The pleasant	
	setting helped participants engage	
	in the workshop.	
Time	Four-day workshop, which	Two-day workshop, which meant
	provided adequate time for	everything was rushed, especially as
	participants to complete the	participants had to commute from
	workshop activities and have	home to the workshop venue.
	plenty of time for breaks.	
Human and	Sufficient funding to arrange	Insufficient funding to pay for
financial	accommodation in a pleasant	participants' accommodation at the
resources	environment. This allowed	workshop venue, which meant they
	participants to socialise and	had to commute. Facilitators were
	continue discussions in an informal	short-staffed, meaning the lead
	setting. Having five facilitators	facilitator was unable to roam
	enabled one of them to roam	between different discussion groups
	between different discussion	as much as he wanted.
	groups and guide them through the	
	process, whilst each discussion	
	group had a facilitator to help	
	prompt discussions.	

for Southern Africa and Food Security Futures case studies

Source: author construct

As indicated in Table 5.4, the FSF workshop was conducted in far less comfortable conditions than PFSA. This appeared to have a negative effect on participants' concentration on the workshop activities. Indeed, during the FSF workshop, one of the facilitators, Mike, suggested to me:

"The heat could be an issue. It's less comfortable for people to work and less conducive to thinking." (Mike, 2016)

One participant, Keith, even left the workshop early, on the second day, complaining that the cramped and stuffy conditions were making him feel unwell:

"Yeah, it was very hot! People were feeling tired. I was not feeling well." (Keith, 2016)

Another participant, Beth complained that the noise of other participants was distracting, because they were in such close proximity:

"There was an issue of movement, and also noise, people are discussing and they're a bit close, so your attention would be drawn at times when it's not necessary." (Beth, 2016)

Nine of the 13 FSF participants, I interviewed, also complained that the workshop had been too short. For example, one participant, Alan reflected:

"The time was so short because there was so much to do, to make sure that all the things to be done [in the workshop] have been taken on board. So, by the time we had begun to understand what is to be done and to start implementing it, we were running short on time." (Alan, 2016)

Similarly, Tristan, compared the time available for the FSF workshop with a larger PSP process he had participated in previously:

"In the other one, it was not under pressure, we were having ample time to do the thing." (Tristan, 2016)

In contrast, two of the PFSA participants I interviewed, Dillan and Elliot, commented on how the pleasant setting had helped them to engage in the workshop:

"For me it was taking time out of my reality and hustle and bustle, and just stepping out into an environment where everything is taken care of. It lends itself to people being able to just focus and concentrate on what we had to do." (Dillan, 2016)

"Being in a space of such luxurious comfort, made it feel like a different dimension, which helped to concentrate on the workshop activities." (Elliot, 2016)

It is evident then, that the conditions in the PFSA workshop were far more conducive to participants focusing their energy on engaging in PSP. This means they were more able to engage in focused, creative thinking, and reflect on their assumptions about the future. Indeed, the importance of having ample time and funding for PSP processes is well-acknowledged. Henrichs et al. (2010) advise that PSP processes should encompass a minimum of two workshops, each of one or two days, to enable in-depth discussions, and indeed learning, to occur. They recognise

that this requires sufficient funding to pay for participants' travel, food, and accommodation. This suggests that, for learning to occur, PSP processes require ample time as well as funding. However, it is notable that learning occurred through the FSF workshop, despite the sub-optimal conditions. The influence of space, time and resources on learning may therefore be relatively small. I thus conclude that, although a comfortable space, sufficient time, as well as adequate human and financial resources, encourage learning in PSP, learning can still occur in their relative absence.

5.6 Conclusion

In this chapter, I have contributed a theoretically-informed explanation of how learning can occur through PSP, focusing primarily on PSP practice in the field of social-ecological resilience. Additionally, I have set out specific conditions that can help make PSP conducive to learning. I argue that learning occurs through interactions between participants with different worldviews, professional occupations, academic disciplines, and socio-economic backgrounds, who are actively and purposively brought into dialogue with each other. The process of PSP can act as a 'boundary object,' (Star and Griesemer, 1989) that encourages participants to make explicit their assumptions about the future, and exposes them to different knowledges. It can also prompt discussion, and focused, creative thinking about the possible futures of SEPs. PSP therefore helps participants to enter their 'zones of proximal development,' (Vygotsky, 1978) and thus to push beyond their usual range of thinking. Participants can be assisted to engage in this focused, creative thinking, by facilitators. Facilitators provide 'scaffolding,' (Wood et al., 1976) by explaining and guiding participants through specific activities, and by using prompts and questions to stimulate discussion.

I found that PSP is most conducive to learning when it includes a diverse group of participants, with relevant expertise. However, careful facilitation is vital for ensuring that the interactions between diverse participants are respectful, and lead to learning. Conducting PSP in a comfortable space, with adequate time, and sufficient human and financial resources, can also help make it more conducive to learning. However, as indicated in Section 5.5, the interactions that occur in PSP are not always favourable to every participant. Some participants are less able or willing to engage in PSP than others. Equally, the engagement of some participants can be hindered if they perceive others to lack relevant expertise. It is therefore important to investigate how and why learning varies between different participants, which I turn to in Chapter 6, below.

Chapter 6: How and why does learning vary between different people in participatory scenario planning processes?

6.1 Introduction

In Chapter 4, I identified learning as the key outcome of participatory scenario planning (PSP), that can help people to tackle social-ecological problems (SEPs). In Chapter 5, I showed that PSP encourages learning by bringing different people together, to deliberate and reflect on their assumptions about the future. Learning can therefore occur, through participants being exposed to different knowledges, which pushes beyond their usual range of thinking. However, I found that practitioners of PSP have limited explicit understanding of and how and why learning varies between different participants. This is an important consideration since, as shown in Chapter 5, some participants appear less able to engage in PSP processes than others, which could affect their learning. In this chapter, I investigate how and why learning varies in PSP. I draw primarily on data from my case studies - Positive Futures for Southern Africa (PFSA) and Food Security Futures (FSF) (described in Chapter 3), because they enabled me to explore the learning of different individuals, as well as groups of participants in detail. Moreover, I draw on my case review and practitioner interviews, to explore general themes in what was learned by whom.

This chapter is structured as follows. In Section 6.2, I present evidence showing who learned what in PSP. All the participants I interviewed in my two case studies appeared to have learned something through PSP, but different participants learned different things. The section is organised around the different themes I identified in what participants learned. Firstly, I show that some participants refined their existing understanding of specific aspects of SEPs, through being exposed to others' views on them. These participants were those who had existing expertise in the topics they deliberated on. Secondly, I demonstrate that most participants in both case studies learned new content about specific topics, especially those who were unfamiliar with a specific topic. The topics about which participants learned, varied depending on their individual fields of expertise, and on the aims of the process in which they participated. Thirdly, my findings illustrate that most participants also identified opportunities for tackling the problems on which they deliberated. The specific opportunities also varied according to participants' existing fields of expertise. Finally, I observed that some participants learned about new ways of thinking about the

future. This was particularly prominent in the FSF case study, in which PSP was a new experience for most participants. In PFSA, although most participants had prior experience of PSP, some participants learned to think differently about the future, through the specific PFSA process.

In Section 6.3, I provide evidence that helps to explain why learning varies between different participants. My findings indicate that learning is shaped by the extent to which participants consider the different knowledges they encounter to be relevant to their interests. I found that learning is strongly influenced by power imbalances. Participants can be marginalised by the roles expected of them in everyday social contexts. Moreover, marginalisation can occur if participants fail to comply with the views of the majority. Furthermore, I show that facilitators of PSP processes can shape learning by defining what is considered acceptable for discussion. I find that facilitators are themselves often influenced by the objectives of wider projects, in which specific PSP processes are embedded. These power imbalances can limit the extent to which participants consider PSP processes to be fair and unbiased. I draw on literature on knowledge co-production and adult learning, to explain that participants are less likely to accept the knowledge they encounter in PSP, if they do not find it to be credible, salient and legitimate (Cash et al., 2003). Hence, they are less likely to push beyond their usual range of thinking which means they are less likely to learn.

6.2 Who learned what?

In this section, I explore who learned what in PSP. The findings that I present, show four different themes in participants' learning: 1) refining existing understanding on specific topics of expertise; 2) new content about specific topics; 3) identifying opportunities for tackling SEPs; and 4) new ways of thinking about the future. I draw primarily on data from my two case studies, but also include data from the few papers in my case review that accounted for variations in what was learned by whom, as well as the practitioner interviews. Table 6.1, below, illustrates the different learning themes, and which participants showed evidence of learning about them. In the following sub-sections, I expand on this table by describing who the participants in each theme were, beginning with those who appeared to refine their existing understanding regarding specific aspects of SEPs.

	Refining existing understanding on specific topics of expertise	New Content about specific topics	Identifying opportunities for tackling SEPs	New ways of thinking about the future
Positive Futures for Southern Africa	Barbara, Morty, Zak, Paul, Penelope	Darren, Dillan, Gareth, Rachael, Paul, Penelope, Geoffrey, Tabitha, Elliot	Dillan, Gareth, Mirriam, Rachael, Paul, Penelope, Geoffrey, Tabitha, Elliot, Facilitators	Dillan, Gareth, Zak, Rachael, Paul, Penelope, Geoffrey, Tabitha, Elliot
Food Security Futures	Alan	Fiona, Sally, Edmund, Mark, Christina, Alan, Tracy	Fiona, Sally, Mark, Christina, Alan, Tracy	Beth, Keith, Fiona, Sally, Edmund, Mark, Karla, Christina, Diane, Alan, Gerry, Tristan, Tracy

Table 6.1 Who learned what in the two case studies, based on participants' responses in my interviews. The headings refer to learning themes I identified in my analysis.

Source: author construct

In both the FSF and PFSA case studies, the organisers invited participants from a range of different professional and disciplinary perspectives. The backgrounds of the participants I interviewed are shown in Tables 6.2 and 6.3, along with the specifics of what they learned. I expand on the information in these two tables in the Sub-sections, 6.2.1 - 6.2.5, below.



Table 6.2 – PFSA participant backgrounds, based on how they described their occupations and interests to me, compared with what participants learned, based on their responses in my interviews. The table shows how learning is linked to participants' backgrounds.

Name	Professional and disciplinary expertise	Learning identified by participants in my interviews
Darren	Artist	Synergies with others, especially on the benefits of small-scale governance
Zak	Academic – climate change adaptation and PSP	Learned about how participants did not consider climate change when thinking about the future, and about the value of the arts for making the future seem more real.
Barbara	Academic – social scientist, innovative governance	Refined existing understanding of SEPs, through exposure to the views of others
Paul	Academic – social scientist, conservation	Value of structured thinking about the future, the role of participatory governance, other participants' views on taking an 'ecocentric ethic' and the potential role of 'artificial meat' in resolving food crises.
Penelope	Academic – social scientist, natural resource governance, futures thinking, artificial intelligence (AI).	Refined understanding of AI, through other participant's views on it, specifically their concerns about its human implications. Learned about a potential opportunity, for AI to contribute to creating just and sustainable futures
Morty	Activist - small-scale urban development, disciplinary background in humanities	Refined understanding of small-scale initiatives through others' views on them. Was concerned that participants saw them as a dogma, and had started thinking of measures to avoid this
Dillan	Activist - user and proponent of encrypted currency	Learned about the roles played by academics, in creating more just and sustainable futures; identified an opportunity for encrypted currency to help contribute to just and sustainable futures.
Gareth	Activist – equitable urban spaces, using artistic methods	Learned about the role gene technology could play in resurrecting coral reefs, about the diversity of perspectives and about the complexity of SEPs
Mirriam	Activist – alternative education and food	Learned about how nurturing and connecting small-scale initiatives, could help create just and sustainable futures
Elliot	Activist - sustainable food	Learned new content about AI, as well as how it could help create just and sustainable futures.
Rachael	Social entrepreneur and academic – disciplinary background in economics	Learned about the importance of supporting environmental causes, and the value of PSP for thinking about the future.
Geoffrey	Practitioner – global environmental governance	Learned new content about AI, the value of thinking about the future in this way. Moreover, he started thinking about including different stakeholders in planning processes.
Tabitha	Practitioner – futures thinking	Learned about AI, as well as possibilities for creating just and sustainable futures in Southern Africa, such as the relative lack of existing infrastructure.

Source: author construct



Table 6.3 – Food Security Futures participant backgrounds, based on how they described their occupations and interests to me, compared with what

participants learned, based on their responses in my interviews. The table shows how learning is linked to participants' backgrounds.

Name	Occupation/Disciplinary	Learning identified by participants in my interviews
	Background	
Fiona	Schoolteacher	Value of PSP, importance of bottom-up governance, task force for coordination, as a means of tackling food poverty.
Edmund	Schoolteacher	New content about nutrition.
Diane	Schoolteacher	Value of PSP for thinking about the future.
Tracy	Schoolteacher	Technology as a driver of food security, and other drivers. Value of PSP for tackling SEPs.
Sally	Academic – agriculture and	Value of PSP for tackling SEPs. Moreover, she expressed a desire for further training in it. Improving cross-sector coordination,
	food systems	as a strategy to improve food security.
Karla	Academic – economics	Value of PSP for thinking about the future
Beth	Government Intern – Finance	Value of PSP for thinking about the future
Mark	Government officer –	Value of PSP for thinking about the future, importance of including different stakeholders in processes of governance, PSP as
	Agriculture	an opportunity to help tackle SEPs.
Christina	Government Intern – Finance	Value of PSP for thinking about the future, taskforce for coordination as a response to food poverty.
Alan	Government Officer - Social	Value of PSP for thinking about the future, understanding of nutrition as a driver of development, and new content about the
	welfare	influence of different aspects of the food system on nutrition.
Gerry	Government Officer - Health	Value of PSP for thinking about the future
Tristan	Government Officer -	Conditions that encourage successful PSP processes.
	Agriculture	
Keith	Businessperson – small-scale	Value of PSP for thinking about the future
	food processing operation	

Source: author construct



6.2.1 *"It was a good exercise for re-organising thoughts that I've had for many years.'* - Refining existing understanding in topics of expertise.

One participant in the PFSA case study, Barbara, indicated that her participation in the workshop had not led to a significant change in her understanding of SEPs. As a professor, with substantial experience of thinking about innovative ways to create just and sustainable futures, she thought the discussions in the workshop replicated ideas she was already writing about:

"I don't mean to sound patronising, but the scenarios that we came up with, you're going to see a book of mine published, with the stuff behind it, it's the same story. It's exactly the same." (Barbara, 2016)

However, although she thought her understanding had not changed significantly, she indicated that she had refined her existing understanding of how to create just and sustainable futures. She explained that she had refined her understanding through receiving feedback on her ideas from other participants:

"Certainly, for me it was a good exercise for re-organising thoughts that I've had for many years. It was a useful exercise, because I got to share my thoughts, and I got a bit of feedback, and I came out even more convinced that the work I do is relevant." (Barbara, 2016)

This was reflected in the interviews I conducted with the other academics who participated. Each of them indicated that they had refined their existing understanding of their relative fields of expertise. For example, Penelope was an academic researcher with expertise in environmental governance, as well as artificial intelligence (AI). She reflected that she had refined her understanding of AI's human implications:

"I'm quite interested in technology, and how it affects things, but I sometimes find it difficult to think more deeply about the human implications of that, and I think this workshop helped me to do that." (Penelope, 2016)

Similarly, Penelope indicated she had learned about the other participants' views of local-level governance:

"[There was] this idea that the local is inherently good, and that no matter what it is, whether it's urban policy, whether it's food, or whatever, it should be local. And, as someone who's worked a lot on things like global governance and large-scale institutions, I found it interesting that there was almost this unquestionable idea that the local was good." (Penelope, 2016)

It was not only academics that refined their existing understanding of topics in which they were experts. One activist, Morty, who had extensive professional expertise in local-level governance, described how he had refined his understanding of those topics. He reflected:

"What I realised, is that... people worship small-scale, and localism as well. It's become this kind of dogma. And, interestingly, we had this whole discussion where I was trying to point out the fact, that people might still get marginalised [in local-level governance]. And people not accepting that, made me realise that actually, we have to be really careful, because people are investing too much in this as a utopian concept." (Morty, 2016)

It is evident then that Morty learned more about other participants' perceptions of local-level governance and recognised the dangers of it becoming a dogma. In this sense, he refined his existing understanding, by considering others' views, on his field of expertise.

In contrast, learning of this nature was not prevalent in FSF. Indeed, as I highlighted in Chapter 5, most FSF participants lacked specific expertise in food and nutrition security (FNS). Only four of my informants had expertise directly linked to FNS: Sally, who was an academic, studying international development, Mark, who was a government officer working on adapting agriculture to climate change, Gerry, who was a nutrition expert, and Tristan, a government officer, who focused on food systems. None of them indicated they had refined their existing understanding in their fields of expertise. Instead, as indicated in the following sections, they learned new content about specific topics, about opportunities for tackling complex problems, as well as new ways of thinking about the future. This could be explained by a quote, given by Tristan, when I interviewed him. He complained that other participants had lacked expertise in FNS:

"I think that people didn't understand a lot about food security, because even when I was trying to talk about food availability, accessibility and utilisation, people didn't understand." (Tristan, 2016)

As I showed in Chapter 5, interacting with participants who have relevant expertise is an important condition for encouraging learning. This indicates that because other participants' levels of expertise were apparently low, there may have been little these four experts could learn from them regarding FNS.

In PFSA, I found that the facilitators also refined their understanding of their fields of expertise. In the week following the workshop, they met to share what they had learned. In this meeting, they all demonstrated they had learned, through being exposed to others' views on how just and sustainable futures could be created. In particular, they indicated they had learned that diverse groups of participants had similar ideas about how just and sustainable futures might look. These ideas included decentralised, bottom-up governance, an 'ecocentric ethic,' and an emphasis on interconnections between humans and nature.

In the practitioner interviews, one of the 16 practitioners, Gavin, also explained that facilitators had learned in processes in which he had been involved:

"We want to learn from participants as much as they might learn from us. And often what happens is we learn more from them than the other way around." (Gavin, 2016)

These two examples illustrate that, as well as encouraging learning for participants, PSP processes can also encourage practitioners to refine their understanding of their fields of expertise, through learning about the views of participants. Indeed, as Gavin stated, PSP processes may be more valuable for the learning of practitioners than for participants. This highlights the question I previously raised in Chapter 5, regarding who PSP processes are really for. The finding that practitioners may learn more from it than participants indicates that the learning reflects researchers' needs for knowledge to develop a more holistic understanding of SES, as according to Lewin's (1997) theory of change. The question remains as to what extent PSP in this context enables participants to take specific actions to intervene in SEPs. This emphasises the importance of exploring how the aims of PSP processes influence what is learned by whom. This will be explored further in Section 6.3.

In contrast, there was no evidence that the FSF facilitators had refined their understanding of FNS through their PSP workshop. As pointed out by the participant, Tristan, most of the FSF participants had limited expertise that was directly relevant to FNS. Indeed, when I interviewed one facilitator, Thomas, he complained that the participants had limited understanding of the issues being discussed:

"There was a lot of knowledge limitation... People do not read a lot... They don't know about issues." (Thomas, 2016)

If interacting with participants who have relevant expertise is a condition for learning, as indicated in Chapter 5, the lack of participants' expertise could have hindered facilitators' learning.

6.2.2 *"I learned a lot around the AI stuff."* – Learning new content about specific, unfamiliar topics

In both case studies, a common learning theme, was that participants learned new content about specific topics, in which they were not overtly experts. In PFSA, nine of my 13 informants indicated they had learned new content about the topics discussed in the workshop. For example, Elliot, an activist for sustainable food, asserted that he had learned new content about AI:

"I learned a lot from Penelope around the AI stuff, because my understanding was completely different. I didn't realise that AI existed in as much as it does already, and that I'm using it every day." (Elliot, 2016)

Similarly, another PFSA participant, Paul, who was a researcher, specialising in conservation, indicated he had learned new information, regarding 'artificial meat' technology, as well as how it could help tackle food crises:

"With the artificial meat you, start off with something in a test tube and then you could end up with resolving world hunger, or the water crisis, for example. I'm oversimplifying it, but something as small as that, food in a test tube, could end up with major repercussions." (Paul, 2016)

In FSF, seven of my 13 informants demonstrated they had learned new content about specific topics. For example, Edmund, Mark and Alan, all learned new content about nutrition. Alan, who represented a government social welfare programme, described:

"Really, I was enlightened on the importance of nutrition as a driving factor of development... It should start from the farming stage. They choose the right soil, the right seed, the right manure, and all those things." (Alan, 2016)

Similarly, Mark, who worked on climate change adaptation, indicated he had learned about the different nutritional requirements of school children at different ages. He described:

"You have to consider the age, because if you have class 1 - 7, the requirement for Class 1 might not be the same as the requirement for Class 7. Their nutritional requirement is quite different." (Mark, 2016)

Learning varied according to the unfamiliar topics participants encountered

In both case studies, the specific topics about which participants learned, varied according to their existing fields of expertise. In the above examples, Elliot was not an expert in AI, nor was Paul an expert in artificial meat. In the FSF example, Mark and Alan did not have prior expertise in nutrition. It is evident then, that the specific content of what different participants learned, was shaped by their existing fields of expertise and encountering knowledge about topics that were unfamiliar to them.

My case review also showed that the content participants learned about, varied according to participants' fields of expertise. Most of the 30 papers in the review did not explicitly include a discussion of who learned what, but of the five that did, one, Palacios-Agundez et al. (2013), indicated that learning varied between participants from different professions:

"Public administration personnel explained to NGO members the different aspects regarding the pace and proceedings of the administration, while the NGO members expounded on relevant aspects that should be considered in ecosystem-based management," (Palacios-Agundez et al., 2013 p.18).

This shows that, through interacting with each other, participants from the two professions learned about aspects of the others' roles and expertise, which they were previously unaware of.

Learning varied according to the design of the workshops

The specific content of participants' learning also varied according to the discussion groups in which participants worked, as well as the design of the overall workshops. In PFSA, as outlined in Table 3.4 (p.60), participants were asked to deliberate on initiatives, or 'seeds' that could be described as promoting just and sustainable futures. Each of the four discussion groups were given three initiatives as discussion prompts and were asked to imagine how they might develop and interact with each other in the future. I found that all four of the PFSA participants who said they learned about AI had been in a discussion group in which participants were asked to deliberate on the potential development of AI. The eight informants who did not indicate that they had learned about AI had not been in groups that deliberated on it. Similarly, the participant who learned about artificial meat, Paul, had been in a discussion group in which participants were

asked to deliberate on the possible future progress of artificial meat. In FSF, participants, who did not have prior expertise in nutrition, learned about it because the workshop aimed to explore food and nutrition security.

It is evident then, that participants in PSP commonly learned new content about specific topics in which they were not overtly experts. The specific topics about which participants learned varied according to their existing expertise, in that they learned when they encountered knowledge about topics they were unfamiliar with. This form of learning may thus be explained as reframing, as described by Ramirez and Wilkinson (2016). According to Ramirez and Wilkinson, scenario planning provides opportunities for participants to explore different perspectives and assumptions about the future, often including expertise in topics that are unfamiliar to them. This enables the development of new knowledge.

The topics about which participants learned also vary according to the topics that specific PSP processes deliberate on. This emphasises the importance of facilitation, as well as the aims of specific PSP processes in shaping what was learned by whom. This will be explored further in Section 6.3.

6.2.3 "It might apply new sets of relations between humans and ecosystems." - Identifying opportunities to tackle social-ecological problems

My findings in Section 6.2.2, above, indicate that participants in PSP were able to learn new content about unfamiliar topics, which I have characterised as reframing, according to Ramirez and Wilkinson (2016). As I described in Chapter 2, reframing can lead to reperception, which involves participants exploring the challenges and opportunities they could face in alternative future contexts, and then identifying responses to them. Correspondingly, another theme in participants' learning involved identifying opportunities for tackling SEPs. In PFSA, 10 of my 13 informants indicated they had learned about such opportunities. In FSF, I found that six of my 13 informants had learned about opportunities for tackling SEPs. I present evidence for this, and how this type of learning varied, below.

Variation aligns with participants' existing fields of expertise

The specific opportunities identified by participants appeared to align with their existing fields of expertise. For example, one participant, Penelope, who had some expertise in AI, described how she had identified the possibility for 'fluid infrastructure' as a means of improving

the sustainability of urban land-use. This was an idea I observed her group developing in the workshop. It developed from an idea put forward by the participant, Tabitha, that in the future urban infrastructure could be constructed from small, solar-powered 'Lego-blocks,' the composition of which could be changed through small electrical signals. The idea was that this could create opportunities for more sustainable and equitable composition of urban spaces. When I asked Penelope what she had learned, through the workshop, she reflected:

"I think this idea of fluid infrastructure was a super-interesting way that I hadn't really thought about things. And this idea that the whole of the built environment might radically shift into something quite different, that might apply new sets of relations between humans and ecosystems, like the collapse of the rural and urban distinction." (Penelope, 2016)

Penelope appears to have learned about an opportunity involving AI, because it aligned with her existing interests and expertise.

Similarly, Morty, who had expertise in local-level governance, reflected on how he had changed his thinking on local-level initiatives, to account for their potential limitations:

"We have to be really careful because people are investing too much in this as a kind of utopian concept... that was actually really important and we've [subsequently] had a lot of conversations about how we promote this, without selling it as a panacea." (Morty, 2016)

This emphasises the importance of relevance for explaining variations in learning, as will be explored further in Section 6.3.

Conversely, in FSF there was little evidence that the opportunities participants identified, directly aligned with their fields of expertise. This is likely because most participants lacked expertise in topics related to FNS. One participant, Tracy, who was a schoolteacher, described how she had learned that FNS could be encouraged by improvements in science and technology:

"If we have improved science and technology, it means that all the land will be used, water we have would be used, and then we can cultivate more, and we will harvest more. Therefore, food poverty would be overcome in Tanzania." (Tracy, 2016)

It may be that he thought he could encourage improvements in science and technology, through his role as a teacher, which would indicate this learning was relevant to his field of expertise.

However, since he did not explicitly say this, there is no obvious link between his learning and his existing expertise.

This could also be explained by the fact that the FSF workshop had a less normative framing than PFSA. The opportunities identified by PFSA participants could have aligned with their fields of expertise, precisely because they viewed them as conducive to their own preferred future conditions. In FSF, on the other hand, participants focused on opportunities to deal with plausible challenges to FNS, rather than how to achieve their own preferred futures. Normative preferences were therefore not explicitly implicated in the opportunities participants considered, and thus what they learned. This emphasises the influence of the workshop design in shaping what is learned.

Learning varied, based on the topics for discussion, set by facilitators

The opportunities identified by participants in FSF were predominantly aligned with the challenges they deliberated on, rather than their existing fields of expertise. As I set out in Table 3.4 (p.60), each discussion group was given a set of parameters, within which they had to try and achieve goals for FNS. Participants identified opportunities to overcome the challenges, posed by certain parameters. For example, participants in one discussion group identified opportunities to overcome the challenge of disjointed governance on FNS. Accordingly, the three participants I interviewed from that group, Fiona, Sally and Christina, all indicated that they had identified the creation of a taskforce, to encourage cross-sector coordination, as a strategy to improve FNS. Indeed, Sally stated:

"We said we need a task force to include different people with different backgrounds, from different sectors, because the issue of food security and nutrition is a cross-cutting issue, so we need different groups of people to be involved." (Sally, 2016)

Likewise, Fiona described:

"We discovered there was no cooperation, so we had to design a committee, aside, which would be responsible to create that cooperation." (Fiona, 2016)

I also found evidence in the PFSA case study, that the opportunities participants identified, varied between different discussion groups. For example, one participant, Paul, recognised how an initiative his group discussed, could be developed into a form of participatory democracy:

"We ended up with this 'Hot Democracy' where everybody counts, where everybody's voice matters. So, as you govern, you learn, and as you learn, you govern." (Paul, 2016)

These two examples from the two case studies show how learning reflected the specific topics that facilitators prompted participants to deliberate on. Again, this emphasises the important influence that the aims of specific PSP processes have, on shaping what is learned by whom. This will be explored further in Section 6.3.

Learning varies according to participants' social and economic background

In PFSA, one participant, Rachael, demonstrated that the opportunities she learned about for tackling SEPs, had been shaped by her own social and economic background. When I interviewed her, she described how she hailed from an ethnic group that was typically associated with poverty and disempowerment in South Africa. She explained that this had influenced her attitude towards environmental, or 'green' causes:

"Some could debate, in this country it's a luxury to become green. The real cause is how to feed our stomachs. Someone could argue: 'why are we investing in the green economy, when people are dying?'" (Rachael, 2016)

She reflected that, through participating in the workshop, she had learned about the importance of environmental causes:

"I see this is an important cause now, as well... I would invest my money in green causes, whereas before I wouldn't, I'd take all my money and put it on developing small businesses. Now I would say 'okay, these small businesses have to thrive, but if I don't contribute to the green cause, there will be no world for them to participate in."" (Rachael, 2016)

Rachael thus learned about the importance of green causes, because she came from a background where those causes were not considered important. This indicates learning, about opportunities for tackling SEPs, can vary according to participants' social and economic backgrounds.

Learning varied according to the spatial levels at which participants work

In the case review, three of the papers that explicitly describe differences in participants' learning, demonstrate that learning about opportunities for tackling SEPs can vary between participants at different spatial levels. In one of these cases, described by Mistry et al. (2014), the

authors specify that the opportunities participants learned about, varied between global and regional level, and local level participants. Regional level participants predominantly identified opportunities relating to policy: '*At the global and regional scales, we see a strong focus on policies influencing society and the environment,*' (Mistry et al., 2014. p.140). In contrast, the authors report that local level participants' learning was more about the actual implementation of strategies in practice: '*This focus on policy is not reflected at lower scales where the uncertainties lie around practice; issues around the actual operationalisation and implementation of effective development and environmental management,*' (Mistry et al., 2014. p.141). This indicates learning about opportunities for tackling SEPs can vary according to the spatial level at which different participants focus.

In my two cases studies, however, the spatial level at which participants focused, did not appear to influence their learning. As demonstrated in Tables 6.4 and 6.5, below, participants in both case studies worked at a range of different levels, but the differences in what they learned do not appear to be linked to this. In PFSA, individual participants from both the regional and local levels learned new content about different aspects of SES, as well as opportunities to create just and sustainable futures. In FSF, participants from both the national and local levels learned new content about specific aspects of FNS, as well as about the value of PSP as a tool for thinking about the future.



Table 6.4 - Spatial levels at which Positive Futures for Southern Africa participants focused in their work, compared with what they learned. The

table shows that differences in participants learning do not appear to be linked to the spatial level at which they work.

Name	Spatial level of working	Learning identified by participants, in my interviews	
Barbara	Global	Refined existing understanding of SEPs, through learning about the views of others.	
Penelope	Global	Learned about other participant's views on AI, specifically their concerns about its human implications, and on local-level governance. Learned about a potential opportunity for AI to contribute to creating just and sustainable futures.	
Dillan	Global	Learned about the roles played by academics, in creating more just and sustainable futures; identified an opportunity for 'encrypted currency' to help contribute to just and sustainable futures.	
Geoffrey	Global	Learned new content about AI, learned about the value of thinking about the future in this way, learned to think about including different stakeholders in planning processes.	
Tabitha	Global	Learned about AI, and other possibilities for creating just and sustainable futures in Southern Africa.	
Darren	Local	Learned about the views of others on the benefits of small-scale governance.	
Zak	Local	Learned about how participants did not consider climate change, when thinking about the future, and about the value of the arts for making the future seem more real.	
Morty	Local	Learned about others' views on small-scale initiatives and was concerned that participants saw them as a dogma.	
Gareth	Local	Learned about the role gene technology could play in rejuvenating coral reefs, learned about the diversity of perspectives and about complexity in SES.	
Mirriam	Local	Learned about how nurturing and connecting small-scale initiatives could help create just and sustainable futures.	
Elliot	Local	Learned new content about AI, as well as how it could help create just and sustainable futures.	
Rachael	Local	Learned about the importance of supporting environmental causes, as well as the value of PSP for thinking about the future.	
Paul	Local/regional	Learned about the value of structured thinking about the future, the role of participatory governance, other participants' views on taking an 'ecocentric ethic' and the potential role of 'artificial meat' for resolving food crises.	

Source: author construct



Table 6.5 - Spatial levels at which Food Security Futures participants focused in their work, compared with what they learned. The table shows

that differences in participants learning do not appear to be linked to the spatial level at which they work.

Name	Geographical level of	Learning identified by participants in my interviews	
	work		
Keith	Local	Value of PSP for thinking about the future	
Fiona	Local	Value of PSP for thinking about the future, importance of bottom-up governance, task force for coordination as a means of tackling	
		SEPs.	
Edmund	Local	New content about nutrition.	
Diane	Local	Value of PSP for thinking about the future.	
Tracy	Local	Technology as a driver of food security, and other drivers. Value of PSP for tackling complex problems.	
Sally	National	Value of PSP for thinking about the future, desire for further training in PSP as a tool to help tackle complex problems. Coordination	
		taskforce.	
Karla	National	Value of PSP for thinking about the future.	
Beth	National	Value of PSP for thinking about the future.	
Mark	National	Value of PSP for thinking about the future, importance of including different stakeholders in processes of governance, PSP as an	
		opportunity to help tackle complex problems in SES.	
Christina	National	Value of PSP for thinking about the future, taskforce for coordination.	
Alan	National	Value of PSP for thinking about the future, understanding of nutrition as a driver of development, and new content about the	
		influence of different aspects of the food system on nutrition.	
Gerry	National	Value of PSP for thinking about the future	
Tristan	National	Conditions that encourage successful PSP processes.	

Source: author construct



These two tables indicate that although learning varied between different participants, there was no discernible link between the spatial levels at which participants focused, and what they learned. However, it is important to note that the two workshops focused on different levels overall. As I set out in Table 3.2 (p.57), PFSA focused on the implications of local level initiatives at the regional, Southern African level, while FSF focused on the influence of national and regionallevel drivers at the national level. This emphasises that the differences in learning varied according to the different processes, rather than between participants from different levels. This indicates, therefore, that learning tends not to vary according to differences in participants' spatial level of focus, within the same workshop. However, learning does appear to vary between different workshops that focus on different levels.

6.2.4 *"Scenario planning is a panacea!"* - Learning about new ways of thinking about the future

The final learning theme I observed in my data was that participants learned about PSP as a new way of thinking about the future. This was a particularly strong theme in FSF, in which 12 of my 13 informants, including a variety of different participants, indicated they had learned about the value of PSP. A particularly strong example of this was Sally, who went as far as saying:

"Scenario planning is a panacea." (Sally, 2016)

Sally subsequently explained she thought PSP was a useful tool to help her prepare for future challenges. She also emphasised her newfound enthusiasm for PSP by talking, enthusiastically, about her desire for further training in using it:

"Now we need to plan, because now the situation is not determined, it is not determined. That's why I'm saying we need to adopt scenario planning. We then have the information of the present situation and then we can forecast the future, informed by the prevailing situation currently... Now I'm eager to understand more about scenario planning! I've started to read some of the articles and publications, with regards to scenario planning." (Sally, 2016)

Another participant, Mark explained that he had learned about the potential value of PSP for government planning:

"It was very beneficial to me as I'm working on the level of policy making... So, using the way of scenario analysis can be used to prepare certain strategies, or certain plans." (Mark, 2016)

Similarly, the participant, Fiona, also indicated she was enthused by the value of PSP, for tackling complex problems:

"For me, that was a very good method of tackling many things. And I'm still interested to know more about how to make use of the method." (Fiona, 2016)

Tellingly, none of the 12 informants who indicated they had learned about the value of PSP had previous experience thinking about the future, in the structured way proposed by PSP. Indeed, Sally strongly indicated this way of thinking about the future was new to her and the other FSF participants:

"It is a new paradigm. We are not used to scenario planning." (Sally, 2016)

Similarly, when I asked Fiona if she had ever encountered this way of thinking about the future before, she responded:

"Never! Never before! This was a new experience." (Fiona, 2016)

It is evident then, that the participants who learned about the value of PSP as a mode of thinking about the future did so, because this way of thinking was a new experience for them.

Indeed, the one participant who did not indicate he had learned about PSP, Tristan, was unique among the FSF participants, in that he had previously participated in another PSP process:

"I participated in the food security and climate change scenario building work in Nairobi." (Tristan, 2016)

Participating in PSP was therefore not a new experience, which could explain why he apparently did not learn about it. However, Tristan went on to compare the FSF workshop unfavourably with the one he had participated in previously:

"Well, I could say that the other one was not under pressure, we [had] ample time to do the thing, and participants to the workshop were gathered in a place where we were just working on that, so we were confined to a space where your mind is just on the workshop, and nothing else." (Tristan, 2016)

In doing so, he showed he had learned about the importance of having sufficient time to conduct PSP, and of being in a space that helped participants focus on the workshop. He therefore, arguably, did learn something about PSP. Specifically, about the conditions that make PSP more conducive to learning.

In the PFSA case study, all but four of my 13 informants indicated they had learned about new ways of thinking about the future, through participating in the workshop. The four participants who did not indicate they learned this had experienced this way of thinking before. For example, when I interviewed Barbara, who had substantial previous experience of participating in PSP processes, she opined that structured thinking about the future was a mundane experience for her:

"This is my job. I do this every day... So, this is bread and butter, you know" (Barbara, 2016)

Another participant, Darren, described how focused thinking about the future had not been new to him, because he often did so in his role as an artist:

"The imaginary part of the future is something I do daily anyway. And I think that's the difference between the creative bunch that was there, and the academics, it's that we do that kind of interpretation every day." (Darren, 2016)

Of the nine informants who indicated they had learned about new ways of thinking about the future, six had not previously encountered PSP. For example, the participant, Gareth, stated:

"I have never experienced this scenario planning before." (Gareth, 2016)

When I asked if he would participate in a PSP process again, he stated he would, and indicated he had learned about the value of focused, creative thinking about the future:

"Yes, I would do it again. It was useful, because it helped people to think outside of their boxes, but within some parameters." (Gareth, 2016)

The participant, Elliot, also described how he learned about the benefits of PSP for creative, but structured thinking about the future:

"To be allowed the space to be in an imagination, in a collective imagination, in a space that wasn't all fun and visioning... It was critical. So, that was really interesting." (Elliot, 2016)

It is clear then, that participants who had not previously encountered PSP learned about its potential value for thinking about the future in a creative, focused way.

However, I also found that three of the nine participants who indicated they had learned about new ways of thinking about the future, did have prior experience of PSP. The most pertinent example of this was Penelope, an academic, who had previously studied and used scenarios in her research. She explained that she had learned about aspects of the PFSA process that she thought could improve PSP practice:

"I'd been involved in a number of scenarios processes, and often I've found the things that came out [of them] reinforced the society we live in, or certain ways of doing and thinking... And I think one thing that was really distinct with this process, was there were guidelines [to direct us] through the process, but there was that ability to go much further 'out there' than a lot of other processes... That was quite empowering." (Penelope, 2016)

In this instance, Penelope evidently learned about the value of being able to think 'outside the box' for exploring more than just incremental change. This characteristic was new to her, even if PSP was not, as the PSPs she had previously encountered, had not featured such imaginative thinking. It is evident then, that although the majority of participants who learned about PSP had been new to it, the participants who had experienced it previously could also learn about new ways of thinking about the future in specific processes.

In the practitioner interviews, one informant, Rick, indicated that learning about the value of PSP had varied according to the spatial level at which PSP was conducted. He described how national-level policy-makers had learned about the value of thinking beyond their immediate futures:

"[They said] it was really useful just to think about what could change in the future, because they don't really think so much about that... It was very much the day-to-day of the present, and fire-fighting everything that's going on, rather than having time to sit and reflect on what the future could bring." (Rick, 2016)

This was in contrast with what the local level participants in his example had learned. As described in Section 6.2.3, local level participants learned more about the 'internal workings' of their community.
This was reflected by another practitioner, Greg, who had also conducted PSP with local level indigenous participants. He indicated that engaging in systematic thinking about the future had been difficult for the elder members of the community. Greg stated:

"I think it was easier for younger people who had had a more diverse exposure beyond their community, so to different ways of thinking and doing things than for the elders... [The elders] didn't have as much facility to speculate or to think about things in a hypothetical way." (Greg, 2016)

This demonstrates that local level, indigenous participants found it difficult to engage with PSP as a way of thinking about the future, especially elder participants who had less exposure to the world outside their community. As such, they may not have found as much value in PSP, as did the younger participants. As I explain further in Section 6.3, the extent to which participants find value in the knowledge they encounter has a significant bearing on what, if anything, they learn.

The evidence presented in this section shows that learning in PSP can be split into four themes: 1) refining existing understanding on a topic of expertise; 2) learning new content about specific, unfamiliar topics; 3) identifying opportunities for tackling SEPs; and 4) learning about the value of PSP for thinking about the future. It is evident that learning can vary, according to participant's level of prior expertise in specific topics, their social and economic backgrounds, their previous experience of PSP, the spatial levels at which they focus, as well as the design of specific PSP processes. In Section 6.3, below, I draw on literature regarding knowledge coproduction, and power, to explain why learning varies in these ways.

6.3 Why does learning vary?

In this section, I explore potential explanations for why learning varies in the ways described in Section 6.2, based on my data, combined with insights from literature on knowledge co-production and power. In their paper on exchanging knowledge in PSP, Chaudhury et al. (2012) reason that participants are unlikely to engage with the knowledges they encounter in PSP if they do not find it credible (based on valid evidence), salient (relevant to their requirements) and legitimate (produced through a fair, inclusive and unbiased process). They draw on the work of Cash et al. (2003), who explain that credibility, salience and legitimacy are key conditions for participants to consider new knowledge. This implies that, if PSP participants do not find the knowledge they encounter to be credible, salient and legitimate, it is unlikely that being exposed

to it will encourage them to enter their 'zone of proximal development' (Vygotsky, 1978) and push beyond their usual range of thinking. As I reasoned in Chapter 5, these are the critical aspects of PSP processes that enable learning.

In this section, I show that salience and legitimacy are especially important in shaping who learns what in PSP. Credibility did not emerge as strongly in my research. This could be because PSP is commonly used to integrate different types of knowledge, as observed by Oteros-Rozas et al. (2015). It may therefore be that practitioners and participants in PSP buy into an idea that all knowledges are equally valid. Indeed, as described in Chapter 5, one PFSA participant, Barbara, emphasised precisely this when she complained about the technical ignorance of other participants.

I begin by showing that learning varies, according to the extent that participants find knowledge relevant to their interests. I then explain that learning varies, according to power imbalances between different participants, which can limit the engagement of individuals. Furthermore, I show that facilitators of PSP processes have considerable power over the engagement and contribution of different participants, through defining what is considered acceptable for discussion in PSP processes. Facilitators themselves are influenced by the aims of wider projects, in which specific PSP processes are embedded.

6.3.1 Learning varies because of the relevance of information for different participants

As mentioned in Section 6.2, only five of the 30 papers in my case review explicitly addressed variations in the learning of different participants. In one of these cases, Ravera et al. (2011a) describe:

"Frustrations of higher level stakeholders (e.g. public administration at the regional level) emerged because there was a mismatch between the scale at which the models operated (i.e. plot, household level) and the scales at which information was requested by these stakeholders." (Ravera et al., 2011a p.448)

This indicates that learning, by regional level stakeholders, was limited, because they considered the information discussed in the scenarios irrelevant.

In the practitioner interviews, all seven informants, who explicitly discussed variations in learning, specified that learning varied between participants from different professional and

socio-economic backgrounds. In these cases, the variation was apparently linked to the extent participants found knowledge to be relevant to their interests. For example, the practitioner, Greg, described a PSP process he facilitated with indigenous communities, which aimed to explore possible future threats and opportunities for their livelihoods. He indicated that specific participants, who he described as 'community leaders,' learned more than others:

"We had a few community leaders, in the project, so they were probably more engaged than the average workshop participants... We did a survey with all the participants [and] the signs were that change was most evident among those community leaders." (Greg, 2016)

In this instance, the 'community leaders' likely had a responsibility, as leaders, to understand the threats and opportunities to the livelihoods of their communities. In this sense, the knowledge they encountered in PSP was particularly relevant, which could explain why they were more engaged, and why learning was subsequently more evident for them than for others. However, as these leaders were in a position of authority, the other participants could have been marginalised, and thus less engaged in the discussions. This highlights the role power can play in shaping learning, which I discuss further in Section 6.3.2.

Another practitioner, Rick, explained that in PSP processes exploring different stakeholders' priorities for ecosystem services, he had found differences between the learning of national level policy-makers, and local level indigenous communities. The local communities learned more about the *"internal workings of their community,"* whereas the national level policy makers learned about the value of using PSP to think about the future. In this case, learning about the actual process of PSP was more salient for the national-level policy-makers, because they recognised how it could help them in their professional roles. In contrast, the local-level participants found the elicitation of micro-level information about their own communities more relevant.

The issue of relevance at different spatial levels was emphasised by two practitioners, Lizzie and Gordon, who both had extensive experience conducting PSP across different levels in the Millennium Ecosystem Assessment (MA). Lizzie emphasised the difficulty of making global level knowledge relevant at the local level, and vice versa:

"How you try to balance the top-down contextualisation and the bottom-up, local information remains a huge challenge." (Lizzie, 2016)

Gordon also indicated that making global level information, relevant to participants at the local level had been a challenge in the MA:

"There was a whole range of scenarios, and that was why there was so much discussion on how we bring these different scales together. How related do they have to be to still be consistent with the global scenarios, but still salient enough to say something meaningful, or to support learning going on for the people who are building or using these scenarios, at a particular scale, for the question at that particular scale?" (Gordon, 2016)

It is evident, therefore, that the relevance of knowledge encountered in PSP can vary between participants at different spatial levels. This corresponds with literature on PSP by Kok et al. (2007). Kok et al. reason that the relevance of knowledge varies, because participants at different levels are likely to be affected by the development of different aspects of SEPs in diverse ways. As Gordon indicated, the knowledge encountered in PSP needs to be relevant for it to *"support learning."* The relevance of knowledge, for different participants, could therefore have important implications for who learns what.

This was reflected in the PFSA case study. As described in Section 6.2, the participants, Penelope and Morty both identified opportunities for tackling complex problems that were related to their own fields of expertise (artificial intelligence and small-scale, localised urban development, respectively). This implies that they identified those opportunities specifically because they were relevant to their own professional roles. Similarly, in FSF Alan, who learned about the importance of nutrition as a driver of development, did so because he identified the connections between nutrition and his own work promoting social and economic development:

"Our side, we are trying to ensure that there is reliable income in a particular household... It's linking together! So, if we have healthy people, everybody will be contributing to the economy." (Alan, 2016)

It is evident, therefore, that participants in PSP processes learn when they encounter new knowledge that appears relevant to their interests. This corresponds with the argument of Cash et al. (2003), that people are more likely to engage with knowledge they consider to be credible, salient and legitimate. The above evidence indicates that the issue of salience is especially important, for encouraging participants to consider the knowledges they are exposed to. My findings also show that, as Hegger et al. (2012) contend, different participants have different criteria for what they consider relevant. These differences appear especially significant between

participants at different spatial levels. In PSP processes including diverse groups of participants, it is inevitable that not all of them will find all the knowledge they encounter relevant to their interests. Variation in learning is therefore likely to reflect the knowledge that participants find most relevant.

This could be explained by the learning scholars, Knowles et al. (2005), who outline a framework delineating the necessary conditions for adults to learn. They argue that adult learners need to know why the knowledge they encounter will be useful to them in their everyday lives. Knowles et al. also reason that an adult's ability to engage with different knowledges depends on their prior life experiences, as well as their motivation. This indicates that participants are most likely to learn when they encounter knowledge that is relevant to their interests, and when it appears acceptable, based on their prior life experiences.

I therefore reason that learning in PSP varies according the relevance of different knowledges, to participants' diverse interests. This appears to be influenced by differences in participants' worldviews, particularly between participants at different levels, and their professional roles and responsibilities. This resonates with the argument of Bandura (1977), who argues that people select, organise and transform the knowledge they encounter in ways that are shaped by their social, economic and political characteristics. The social context thus shapes what individuals learn. Indeed, as Jasanoff (1996) explains, knowledge and social organisation are 'coproduced' such that the ways people come to know the world, and the ways that they live in it, are closely entwined.

In her influential critique of participatory processes, Kothari (2001) contends that the socially constructed nature of knowledge, means processes bringing different knowledges together are heavily infused with power imbalances. She thus argues that these power imbalances can shape the outcomes of such processes. It is therefore important to investigate how power imbalances shape learning in PSP. In Section 6.3.2, I present evidence showing that power imbalances between participants themselves, as well as between participants and facilitators, can influence the legitimacy of knowledge encountered by different participants in PSP.

6.3.2 Learning varies because of power imbalances

In Chapter 5, I explained that learning can occur through interactions in which participants are exposed to different knowledges, through PSP prompting discussions between them. It follows that learning could vary according to participants' relative engagement in these discussions. Following on from Section 6.3.1, which highlights the importance of relevance in shaping who learns what, in this sub-section I show how power can shape participants' engagement in PSP, and how this subsequently influences the legitimacy of the knowledge they encounter. As I set out in Chapter 2, I follow the broad definition of power, proposed by Avelino and Rotmans (2011), as: 'the capacity of actors to mobilise resources to achieve a certain goal' (p.798). In this definition, resources are viewed as assets, such as people, knowledge and money, that help actors to achieve certain goals. I begin by exploring the influence of power imbalances, between participants, which typically reflect power as having greater capacity than others to mobilise resources.

In one of the papers in my case review, Ravera et al. (2011a) describe how the voices of marginalised participants, like women and young people, contributed less than others in PSP workshops on climate change adaptation. The authors state: *'The voices of powerless people, such as workers, women and young people, had less influence in the formal decision-making spaces,'* (Ravera et al., 2011a p.449). The less powerful participants thus had less capacity to contribute their knowledges, so the knowledges that participants encountered were more likely to reflect the contributions of powerful participants. Indeed, in another case described in the same paper, Ravera et al. (2011a) show that the initial phases of a PSP process in upland conservation were: *'dominated by more formally educated stakeholders, which left some participants feeling uncomfortable,'* (Ravera et al., 2011a p.450) This indicates that the knowledge participants encountered was more likely to reflect the perspectives of more powerful participants. Moreover, the marginalised participants appear to have been less likely to engage, enthusiastically in the PSP process.

This was reflected in my practitioner interviews, in which all 16 of my informants indicated that power influenced learning in PSP. One practitioner, Barry, had substantial experience conducting PSP with diverse groups of participants. He provided a detailed account of how PSP workshops are often dominated by certain types of participants: "I've noticed in the groups that I've facilitated, people who are more senior, [especially] older men tend to dominate... Or if you have a table where government agencies are represented, an NGO and private sector, it is most likely that the government agency and the private sector would dominate over the voice of [someone from] an NGO." (Barry, 2016)

When I asked Barry to explain how these power imbalances could influence learning, he reasoned that the domination of PSP processes by individuals or groups of participants could create what he described as 'bias' in discussions:

"You need to make sure that everyone has a voice, because there will always be people who dominate, and that's where you have that power issue... I think legitimacy is where the power issue comes in, because that is all about [people being able] to legitimately participate... If certain people will keep speaking up, they will probably create a bias, in their decisions on developing the scenarios." (Barry, 2016)

This indicates that the knowledge to which participants are exposed is more likely to reflect the knowledge of powerful participants. If other participants thus fail to consider this knowledge to be legitimate, they may be less likely to learn from these discussions.

Two practitioners, Danielle and Mary, both highlighted one characteristic that could influence power imbalances and therefore learning, which was participants' command of language. Danielle, a highly experienced practitioner, reflected on a PSP process she was planning at the time, in which she intended to include participants from different ethnic backgrounds. She expressed a concern, about engaging participants with different first-languages:

"I'm quite worried, because I don't know how much language capability we can build into the project, [and] doing it all in English seems like a big problem." (Danielle, 2016)

When I subsequently asked Danielle to explain what the influence of language might be, she responded by raising the important question:

"Well, what's lost in translation?" (Danielle, 2016)

Indeed, if participants cannot fully understand the discussions that occur, their engagement and their learning are highly likely to suffer.

This was demonstrated in a different way, in a case described by the experienced practitioner, Mary. As I set out in in Chapter 5, Mary reflected on a situation in which some participants had been unable to participate fully, because they were illiterate. The participants had been asked to complete a task that involved writing on posters, but the illiterate participants had not been able do so. As such, they were unable to contribute their own ideas to the workshop. The other participants therefore inevitably missed out on learning from their knowledge, and their inability to engage limited their own learning.

In the PFSA case study, the influence of power on learning was illustrated by the relative engagement of the artists in the workshop discussions. In my own observations from the workshop, I noted that the artists were often quieter and more reserved during the discussions than other participants. One of the participants who I interviewed, Gareth, who was not a professional artist, but engaged with the arts to a large degree in his professional and personal life, showed a concern that the artists had not felt comfortable engaging in the discussions:

"I wondered if some of the artists felt a bit uncomfortable in the workshop space, because it was all about speech and discussion, whereas they are used to expressing themselves through the body and dance." (Gareth, 2016)

This was reflected by one of my informants, Darren, who was an artist. He reflected that he, along with the other artists in the workshop, had been less comfortable than other participants in articulating idea through verbal discussions:

"I'm not very good at debating, or arguments and stuff. I'm just not [used to] that kind of environment... I think that the strength of the artists there was really good for the scientists, but I personally don't know if the artists felt completely that they contributed their best." (Darren, 2016)

When I subsequently asked Darren what he had learned, he indicated his learning was limited, especially in terms of informing his own work:

"I don't think there was anything that could directly inform my practice... There wasn't like a couple of 'a-ha' moments for me." (Darren, 2016)

The structured discussions, therefore appear to have limited the engagement of participants from an artistic background. This corresponds with my finding in Chapter 5, that PSP is not necessarily conducive to the learning of all participants.

In the FSF case study, I observed there was a certain discussion group in which the discussion was dominated by a small group of men, and the younger, female participants appeared quieter and more subdued. Indeed, Harry, the facilitator for that group reflected:

"I had this group with this really vocal group of guys… They were really emphasising one thing, which we had already discussed, and that really slowed the process down." (Harry, 2016)

He described how, although he encouraged the participants to move on, they still slowed the process down. He explained that the effect of this was:

"we didn't have time to go into so much detail at the end." (Harry, 2016)

The young women were likely influenced by Tanzanian culture, in which women typically stay quiet in the presence of older men (Verma, 2001). The dominant, male participants thus appear to have focused on what they found most relevant, which meant the female participants were unable to focus on topics, that might have been more relevant for them. The knowledge discussed in the group could therefore have had limited the legitimacy, and thus learning, for the young, female participants. Indeed, when I interviewed Christina, a young, female intern, she struggled to articulate what she had learned through the workshop.

It is therefore evident that, as indicated in Chapter 5, the interactions that take place in PSP are not always appropriate for all participants. This is especially the case when participants are not fluent in the language used in specific PSP processes, when they are not used to verbal expression and debate, and when they are already marginalised in everyday life. These participants can thus become side lined in PSP processes, which can limit their learning.

In the PFSA workshop, one of the scientists, Barbara, indicated she had felt excluded at a certain point in the workshop. Interestingly, I did not expect this because she was an accomplished professor with substantial expertise in her field and was very vocal and assertive in her contributions during the discussions. However, I observed that she often appeared to be side-lined during breaks and mealtimes, in that other participants did not interact with her during these periods. During the plenary discussion at the end of the workshop, Barbara made a very impassioned speech, in which she took a rather revolutionary tone about actively *"disrupting institutions,"* and stating that *"the present system must collapse!"* The other participants quickly changed the direction of the discussion, away from Barbara's spiel and turned to thinking about more incremental changes. When I interviewed Barbara and asked her what she had found most

challenging about the workshop, she complained that her contributions in the plenary had been ignored, in this instance:

"There was an opportunity for changing people's attitudes and it was just kind of brushed away by this very generic, collective conversation we were having." (Barbara, 2016)

In this instance, although Barbara had a lot of expertise and appeared vocal and assertive, her voice seemed to be marginalised by the other participants. This emphasises that, as Kothari (2001) contends, power imbalances in participatory processes are complex and manifest themselves in multiple ways. Barbara's voice appears to have been marginalised, not because she was less able to engage in the PSP process, but because her speech about disruption did not fit with the direction of other participants' collective thinking. Barbara therefore appeared not to have found much of value in the 'generic' discussion that occurred in the workshop. This could explain why, as described in Section 6.2, she stated that the workshop had merely helped her reaffirm her existing ideas about creating just and sustainable futures and did not indicate she had learned anything new.

As Kothari argues, participatory processes can, and often do, reproduce existing inequalities, and serve the interests of the powerful. My findings show that this is reflected in PSP processes. Participants who are already marginalised in everyday life can also be disengaged in PSP processes. Similarly, participants can become marginalised if they are not fluent in the language in which specific PSP processes are conducted, and if they are not comfortable with the modes of expression it requires. Another participation scholar, Kapoor (2005), emphasises that power imbalances can act as a self-surveillance mechanism, which means participants feel incapable of acting and speaking in ways that oppose the roles expected of them in society or in a particular social context. This is evident in the example of Barbara in PFSA. She did not act in such a way that the other participants thought was conducive to the prevailing discussion, and thus appears to have been alienated.

The work of adult learning scholar, Brookfield (2005) offers an explanation for the influence of these power imbalances on learning. He contends that in group learning processes, power imbalances can lead to conformity in participants' ideologies, or 'groupthink.' He explains that groupthink typically reflects the interests of the most powerful members of a group. The knowledges of less powerful participants can thus become excluded, especially if they dissent from what the majority considers beneficial. The consequence of groupthink is that PSP participants are most likely to be exposed to the knowledges of the most powerful participants.

This can reduce the legitimacy of the knowledge that participants encounter. It is therefore less likely, that their exposure to it will encourage them to push beyond their usual range of thinking, which means they are less likely to learn.

Facilitation and power

In Chapter 5, I showed that facilitation plays a vital role in enabling learning by explaining and guiding participants through PSP processes, as well as by prompting discussion, and managing interactions between participants. I also found that because of these roles, facilitators have a significant influence on who learns what. Power, in this context, typically takes the form of the capacity to mobilise people, in that facilitators can influence who speaks at what point and whose contributions are considered acceptable, and whose are not.

As I stated in Chapter 5, the papers in the case review typically did not reflect on the role of facilitators in specific PSP processes. Accordingly, I did not find any evidence of a reflection on how facilitators influenced learning in the case review. This could be because of editorial constraints on what could be incorporated into the papers (Baxter and Eyles, 1997), but also indicates a lack of mandate for reflection on the role of facilitators.

In the practitioner interviews, six of the seven informants who talked about how learning varied recognised that facilitation could play an important role in shaping what was learned. One way in which facilitation can influence learning is through shaping the discussions that occur in PSP. This was aptly demonstrated in an example described by one practitioner, Rick. The PSP processes he described had been facilitated by a team of academics from a range of different disciplines including himself, a human geographer. When I asked him about the challenges he faced in that process, he reflected on how another facilitator had tried to push the process in a certain direction, whilst he had wanted to keep the process more open:

"In the scenario work, the person who joined me to help facilitate, was very [keen] on being quite rigid and setting the boundary of the scenarios, so in the scenarios they must just think about conservation. And I thought, no, because actually, everything affects everything, so why does it have to be focused on a particular topic?" (Rick, 2016)

As I described in Section 6.2, local-level participants in the process described by Rick learned about the 'internal workings' of their communities, rather than solely about conservation, as the other facilitator desired. This indicates that the less rigid approach, advocated by Rick, enabled participants to concentrate on discussing what was most relevant to them. If the other

facilitator in this example had succeeded in focusing the process exclusively on conservation, the learning would likely have been different. Indeed, when I asked Rick how taking a more rigid approach might have affected learning, he mused that participants could have identified more opportunities for action on environmental problems:

"If it was much more focused then you could potentially have much more specific action linked with it." (Rick, 2016)

It was unclear, however, whether the eventual learning was more useful, and for whom, than would have been the case if the process had been more 'focused.' The PSP process described by Rick, was intended to identify ways that local conservation strategies could be supported by national and regional-level policies. The less rigid approach clearly detracted from identifying specific actions for supporting local conservation strategies, even though it gave participants more freedom to deliberate on their own interests. This highlights the important question of who the benefits of PSP are intended to be for, which I return to in Chapter 7.

Another practitioner, Nigel described how he had controlled what was considered plausible in a PSP process in which he was involved, because he later had to translate the scenarios into spatial models:

> "What happened was that, as the person that got to implement it in the spatial modelling, I had to look at [the scenarios] and go 'look, this is not going to be practical, I can't implement that...' I then went back [to the participants] and said 'okay this is what you gave us, this is how we've interpreted it in terms of it being a tractable modelling response, and you've got to remember, it's got to be practical when you're modelling."" (Nigel, 2016)

In this instance, by iteratively refining what participants considered feasible, Nigel shaped what was considered an acceptable contribution to the discussion.

Similarly, the highly-experienced practitioner, Deborah, described a situation, in which two participants started suggesting ideas that she considered disruptive to the prevailing direction of thinking in the workshop. As a result, she took these participants aside and reined them in: "We had a couple of economists who just flipped out and became communists on us, and it was near the end and I remember thinking like 'what the hell, these guys have gone off the deep end!' And [we] had to go off into a smokefilled room with them and get things settled down. They were really advocating for a centralised economic solution, which was really not something anyone considered a viable mechanism for addressing the problem." (Deborah, 2016)

These participants were therefore prevented from exploring that particular avenue. Hence, their exploration of possible futures was directed towards what Deborah, and the other facilitators considered more constructive.

The above three instances show that facilitators can define what is considered an acceptable contribution to discussions in specific PSPs, and what is not. In this way, they inevitably privilege some voices and marginalise others. This could influence the legitimacy of knowledge produced in PSP. If participants do not find this knowledge to be legitimate, it is less likely they will learn through their exposure to it.

It is important to note, however, that in each of these examples the facilitators were aiming to meet the objectives of specific projects. For example, the process Rick described aimed to explore how community responses to environmental change might fit in with global conservation initiatives. He explained that this objective allowed him to take an open-minded approach:

"I think for what we wanted to do, in terms of just exploring what would happen, whether local solutions would fit into the future, I think keeping it broader was better." (Rick, 2016)

The process Nigel referred to, aimed to inform a national government's negotiations for carbon storage payments. He explained that this meant the process needed to be focused and realistic to capture policy-makers' attention:

"We were definitely aiming for realistic, because this was not a research exercise that was out in blue sky, this was about delivering things that could be presented in Copenhagen at the REDD negotiations." (Nigel, 2016)

These differences in the project objectives, inevitably influenced what the facilitators deemed appropriate for discussion. In fact, Nigel indicated a particular trend had become quite a

prominent issue in Tanzania, but had only been considered briefly in the workshop, because it was considered too unrealistic:

"It's no good [thinking about] 'just as a surprise, why don't we just sell off the whole of the country to China,' which, interestingly, is something that we did look at a little bit and has actually become quite an issue." (Nigel, 2016)

In this instance, the fact that selling off land in Tanzania was not considered in the workshop, meant that participants were not able to learn as much about it as they might have if a more exploratory approach had been taken.

The influence of specific process aims was clearly visible in the PFSA workshop. The overall 'Positive Futures' project aims to encourage learning about what just and sustainable futures would look like for different participants, as well as to explore how these futures could be created. In the workshop I observed in South Africa, the facilitators pushed this normative framing right from the start, by emphasising the themes of transformation, creativity, and positivity. Indeed, the introductory session began with a presentation by one of the facilitators, Doris, which highlighted present unsustainable conditions in SES, and emphasised the importance of thinking about opportunities for positive change:

"Meeting the challenges of the 'Anthropocene' requires complete transformation." (Doris, 2016)

This shows that the facilitators primed the participants to consider significant changes towards more just and sustainable conditions, as well as to focus on imagining future conditions they considered desirable. This subsequently shaped participants' discussions and thus what they learned.

The facilitators continued to emphasise these themes throughout the workshop, by prompting participants to think creatively about transformations and complexity in their discussions. For example, I observed one facilitator, Tina, encouraging participants to make sure they were thinking about *"really innovative,"* links between different seeds. This emphasised the need for participants to consider new ideas about how change could happen. Similarly, Nichola emphasised, in her discussion group, that participants needed to focus on *positive* futures, when they were beginning to think about dystopian ideas. In doing so, she stressed the normative framing of the workshop to consider positive futures. In this way, the PFSA facilitators shaped what was discussed in the workshop, and, thus, what participants learned.

Another way the facilitators shaped discussions in PFSA, was through acting as a scribe (writing notes on flipchart paper, as participants spoke), and encouraging participants to note things down. During the discussions, I noted how each of the facilitators listened to the discussions that were occurring in their groups and noted down summarised points on stickynotes, which were then used to record participants' thinking. I also noted that they regularly suggested participants write their ideas down themselves. Indeed, one participant, Elliot, described how the facilitator in his group, Tina, had encouraged participants to write down points she thought were important:

"[She would say]: 'okay guys, that's great, but write it down.' So, we would write it down." (Elliot, 2016)

The facilitators, therefore determined what was committed to paper, according to what they considered most relevant in the discussions. In this way they inevitably advocated some contributions over others, which shaped what was discussed, and thus, what was learned.

Indeed, the facilitators' influence was evident in the types of opportunities participants identified, for creating just and sustainable futures. As described in Section 6.2, one discussion group deliberated on how artificial intelligence could help encourage just and sustainable futures. This same group was the one the facilitator, Tina encouraged to be 'really innovative.' Accordingly, the participants I interviewed from that group indicated they had identified opportunities including 'fluid infrastructure' where artificial intelligence was used to help make urban land use more sustainable, and the possibility of using robots to rebuild coral reefs. This demonstrates the emphasis placed on innovation led to the identification of highly innovative and imaginative ideas, rather than more incremental solutions.

The PFSA facilitators were also influenced by the aims of the overall 'Positive Futures' project. In this case, the facilitators told me, the overall aims set by the funders of the project, were to improve their selection of small-scale initiatives to fund. They expected the PFSA project to yield information on how small-scale initiatives could be supported to enable positive change. The influence this had on the facilitation of the workshop was demonstrated most strongly in the discussion group facilitated by Nichola. The group had been exploring how the possibility of 'gene technology' could result in immortal human beings, and thus, the possibility of interplanetary travel. Nichola became concerned this idea was too radical, despite the emphasis on creativity and innovation. As such, she directly explained to the participants that the funders required practical information about how specific initiatives could develop. In doing so, she encouraged the group to consider more plausible possibilities that could be more palatable to the funders, but

discouraged participants from exploring particular future possibilities, which inevitably influenced what they learned.

However, although the PFSA facilitators evidently influenced learning by emphasising the aims of the process, three of my informants commended them for not imposing their own views on the discussions. For example, one participant, Miriam reflected, that the facilitator in her group had openly indicated strong views on a discussion point related to genetic engineering. However, he commended her for not trying to push her views on the participants:

"Besides the fact she was completely pro-genetic modification, she realised the group was going in the opposite direction, and she didn't try to resist. You know, she was challenging us by asking questions, but she never pushed the discussion according to her own values." (Miriam, 2016)

This indicates that, although the facilitators promoted the aims of the process by prompting participants to think about positive and creative ways to create just and sustainable futures, they did not try to promote their own personal views.

In the FSF case study, I also observed that the facilitators played an important role in shaping what was discussed, and thus what was learned. The FSF workshop took an exploratory approach to exploring challenges and opportunities for food and nutrition security (FNS) in Tanzania. In doing so, it aimed to contribute to the Tanzanian government's 'Five-Year Development Plan'. Accordingly, I observed that the facilitators regularly prompted participants to explore a wide range of different aspects of FNS, as well as to consider how they were linked to policy. For example, I observed how one of the facilitators, Harry, prompted participants to think about the implications of different policy objectives, and kept asking them to think about whether there were any factors they had not yet considered. This encouraged participants to explore wide-ranging, plausible event, conditions and trajectories, as well as their implications for FNS.

I also observed how the FSF facilitators engaged individual participants by posing questions directly to them. For example, I noted one moment where a facilitator, Mike, asked one participant, Edmund, to describe how his everyday role would look in the scenario he was imagining. In doing so, the facilitators aimed to encourage a more even contribution from different participants. Indeed, six of the 13 participants I interviewed indicated that the facilitators had encouraged an even contribution of participants across the groups. For example, one participant, Tracy, described how the facilitator, Mike, had used questions to encourage the contributions of each and every participant:

"He tried to make sure that everyone has given points. He asked questions to this one, you answer, this one you answer. Everyone was asked a question and answered it." (Tracy, 2016)

In contrast, another facilitator, Thomas, allowed each participant to make a short speech about their contribution, and then moved on to the next person, rather than prompting discussion by asking short, punchy questions. When I discussed these different facilitation styles with Mike, he suggested that Thomas's style was sub-optimal:

"Letting everyone make a little speech takes a long time and leads to people getting bogged down on one particular point, which leaves less time to deal with a range of different issues." (Mike, 2016)

However, because they were under-staffed, Mike had difficulty facilitating his own facilitation team and guiding the less experienced Thomas, simultaneously. He was therefore unable to change Thomas's facilitation style. If participants thus did not find the points they got 'bogged down on' to be relevant and legitimate, it is unlikely they would learn from those discussions. This could explain why, as shown in Section 6.2, only one of the participants who had been in Thomas's discussion group, Mark, indicated he had learned new content and identified opportunities for tackling complex problems.

In the same way as in PFSA, I observed that the FSF facilitators actively influenced learning through synthesising and summarising the discussions that occurred. In the FSF workshop, I observed how Mike and Harry regularly summarised participants' discussions and asked them to write down their ideas. For example, I observed one point in Mike's group when he decided that the scenario participants were creating was becoming too negative, so he encouraged them to consider positive points and asked them to write them down, rather than the negative ones. The facilitators thus defined what was considered an acceptable contribution and what was not.

The above evidence shows that facilitators of PSP processes have significant power over what is discussed in them. Facilitators can influence what is considered a valid contribution, and thus, who contributes what knowledge. This corresponds with the findings of learning scholar, Tett (2016). Tett demonstrates that facilitators of group learning processes have a significant influence on what is learned. She argues that facilitators can advocate or oppose the contributions of participants simply through acknowledging them, asking participants to expand on ideas, and instructing them to write things down. This closely mirrors my findings, that facilitators can set the tone of the workshops they facilitate and can privilege the contributions of

some voices over others. Facilitators can thus create 'groupthink' as described by Brookfield (2005), in that shaping what is an acceptable contribution can result in participants conforming to certain ways of thinking.

If participants thus fail to consider PSP to be a fair, inclusive and unbiased process, they may not consider the knowledge they encounter to be legitimate. Participants may subsequently be less likely to engage with the knowledges they encounter, which means they will not push beyond their usual range of thinking and will be less likely to learn. I therefore reason that learning varies because of facilitators' power to influence what is considered a valid contribution to discussions, and what is not.

However, it is also evident that facilitators are influenced by the objectives of wider projects in which specific PSP processes are embedded. Indeed, in their review of PSP processes, Oteros-Rozas et al. (2015) acknowledge that PSP processes are typically designed to meet research objectives, rather than learning objectives. As such, what facilitators consider to be acceptable for discussion in PSP, is shaped by their obligations to wider projects, including their funders. This resonates with the critique of participation by Cornwall (2008), that participatory processes are typically owned by those who initiate them. Likewise, another participation scholar, Chambers (1997), contends that the facilitators of participatory processes ultimately have power over their outcomes. The objectives of projects in which specific PSP processes are embedded thus seem to have a significant influence on how learning varies between different participants.

6.4 Conclusion

In this chapter, I explored the question of how and why learning varies between different participants in PSP. I found four discrete themes of learning. Firstly, participants can refine their existing understanding of specific topics. These participants are typically those who have existing expertise in the topics discussed by specific PSP processes. Secondly, participants commonly learn new content about specific topics, such as artificial intelligence, governance, and nutrition, in which they are not, overtly experts. The specific topics about which participants learn vary according to their existing levels of expertise and ignorance in the topics discussed, and according to the design of specific PSP processes. Thirdly, participants can identify opportunities for tackling SEPs. The sorts of opportunities they identify vary according to their fields of expertise, their social and economic backgrounds, and the design of specific PSP processes. Finally, participants can learn about new ways of thinking about the future. This is especially common in participants,

for whom PSP represents a new way of thinking about the future. However, participants who are familiar with PSP can also learn from new ways of doing it, in different specific processes.

In the second half of this chapter, I explained that these variations in learning are shaped by the relevance of knowledge to participants' interests, and the extent they consider it to be produced through fair and unbiased processes. The extent to which participants find knowledge to be relevant is shaped by whether they consider it useful to their interests. The extent to which participants find knowledge to be legitimate is influenced by power imbalances, which can lead to conformity in participants' thinking, or 'groupthink.' Moreover, facilitators of PSP processes have significant power to define what is considered acceptable for discussion in specific PSP processes. If participants do not find knowledge to be salient and legitimate, it is unlikely that being exposed to it will encourage them to push beyond their usual range of thinking, and thus, to learn.

It is important to note that the ways in which facilitators design and facilitate PSP processes are themselves shaped by the aims of wider projects in which PSP processes are embedded. It is therefore vital to consider the question of for whom, and indeed at what level PSP is intended to result in learning. Do those who design and use it just intend to elicit different knowledges to inform a macro-level understanding of SEPs, as Lewin (1997) argues is a vital prerequisite for intervening to tackle them? Or, do they go further and intend for it to help stakeholders identify specific actions at the micro-level, as per the reperception aspect of Ramirez and Wilkinson's (2016) framework? If the knowledges discussed in PSP are defined by broader project objectives, to what extent is it really intended to enable learning for participants? I discuss this further in my concluding chapter below.

Chapter 7 – Conclusion

7.1. Thesis Overview

Participatory scenario planning (PSP) - interactive processes in which diverse groups of participants develop multiple, plausible narratives of future events, conditions and trajectories has increasingly been applied to tackling social-ecological problems (SEPs) in a variety of global locations, at different geographical levels, and by a mixture of academic, governmental and thirdsector organisations. As Oteros-Rozas et al. (2015) show, PSP is commonly used to incorporate the knowledges of diverse stakeholders in processes of dialogue on tackling SEPs. This reflects assumptions in wider literature on social-ecological resilience, that dialogue and knowledge exchange between different stakeholders, can help promote changes towards more sustainable conditions in social-ecological systems (SES) (Boyd and Folke, 2012). However, as indicated in the influential critique of participation, Kothari (2001), processes that claim to incorporate different knowledges, typically privilege the knowledges of some participants over others. Indeed, as argued by Chambers (1997) and Cornwall (2008) the outcomes of participatory processes are typically owned and defined by those who initiate them. This research was therefore motivated by a need to investigate the benefits of PSP for those whose knowledges it incorporates.

As indicated in the recent review of PSP by Oteros-Rozas et al. (2015), PSP may be particularly useful for learning. I therefore focused on exploring learning, as a potential benefit of PSP. I structured my research around a conceptual framework, that I developed, informed by learning theories. I used the cognitivist theory of the 'zone of proximal development,' (Vygotsky, 1978) to explain that learning in PSP occurs, through interactions between participants with different fields of expertise. I used the concept of 'boundary objects' (Star and Griesemer, 1989) to explain how PSP prompts discussion between different participants. I also drew on the concept of 'scaffolding' (Wood et al., 1976) to explain how facilitators can assist participants to engage in PSP processes. I then used theory from adult learning, knowledge co-production, and participation to explain variations in what participants learned through PSP processes.

I designed my research around three sources of data. I began by conducting a case review of 30 carefully selected cases in the peer-reviewed literature in which PSP had been used to help tackle SEPs. I built on this by conducting semi-structured interviews with PSP practitioners, and investigating two case studies of carefully selected PSP workshops, as they occurred. This methodology enabled me to elicit rich, qualitative data, which I analysed to explore the expected

and reported benefits of PSP, how, and under what conditions, learning occurs, and how and why it varies in PSP. I present a summary of my findings in Section 7.2, below.

7.2 Research Findings

7.2.1 What are the expected and reported benefits of participatory scenario planning?

The first set of findings from my research concern the expected and reported benefits of using PSP to tackle SEPs. These findings are presented in Table 7.1, below.

Table 7.1 – Summary of research findings, regarding the expected and reported benefits

Expected and reported benefits of PSP	Details
Helping participants explore complexity and uncertainty in SEPs	The use of PSP is typically underpinned, by an assumption that it can break-down, or structure, future possibilities into alternative narratives of plausible future events, conditions and trajectories. This is believed to help participants explore the complexity and uncertainty of SEPs. This is not frequently reported as a benefit, because it is linked to, and sometimes acts as a precursor to other benefits.
Learning	Learning is commonly expected as a benefit of PSP. It is often reportedly achieved through bringing different participants together to deliberate on the future. However, practitioners have limited theoretically-grounded understanding of how learning occurs and appear to give limited attention to how and why learning varies for different participants.
Developing and testing responses to SEPs	Proponents of PSP commonly expect, and report, that it helps develop responses to SEPs, as well as test them in alternative future conditions. However, I found limited evidence of these responses being implemented. This could be because PSP needs connecting with long- term processes of policy-making and action. Moreover, it requires the participation of influential stakeholders, to encourage the implementation of the responses it helps identify.
Limited provision for evaluation of PSP	There is limited evidence of practitioners considering the strengths and weaknesses of PSP, especially compared to alternative methods for encouraging dialogue for tackling SEPs. This could be explained by limited provision being made by practitioners and ultimately their funders for systematic evaluation of PSP processes.

of PSP processes, based on my analysis.

Source: author construct

7.2.2 How and under what conditions does learning occur in Participatory Scenario Planning?

My second set of findings explain the attributes of PSP processes that enable learning to occur. They also describe the conditions that can make PSP more conducive to learning. I present these findings in Table 7.2, below.

Table 7.2 – Summary of research findings regarding how and under what conditionslearning occurs in PSP processes, based on my analysis.

How and in what conditions learning occurs	Details
Learning occurs through interactions	Learning occurs through interactions between participants with different worldviews, professional occupations, and social and economic backgrounds.
PSP prompts discussion between different participants	PSP can act as a 'boundary object,' (Star and Griesemer, 1989) that encourages participants to make explicit their assumptions about the future. It therefore exposes participants to different knowledges. Hence, it can prompt discussion, as well as creative, focused thinking about the future.
PSP enables participants to push beyond their usual range of thinking	PSP therefore helps participants to enter their 'zones of proximal development,' (Vygotsky, 1978) and hence, to push beyond their usual range of thinking.
Facilitators assist participants to engage in PSP	Facilitators of PSP processes can help participants to engage in focused, creative thinking, by providing 'scaffolding,' (Wood et al., 1976). This involves explaining and guiding participants through specific activities, as well as stimulating discussion using prompts and questions.
Conducive conditions	PSP is most conducive to learning when it includes a diverse group of participants with relevant expertise. Careful facilitation is required to ensure interactions are respectful and constructive. Conducting PSP in a comfortable space with adequate funding, time and human resources can also make it more conducive to learning. However, some participants are inevitably less able and/or willing to engage in PSP processes than others.

Source: author construct

7.2.3 How and why does learning vary in participatory scenario planning

processes?

The third and final set of findings from my research concern how learning varies between different participants in PSP, and why these variations occurred. These findings are presented in Table 7.3, below.

Table 7.3 – Summary of research findings, regarding how and why learning varies fordifferent participants, based on my analysis.

How and why learning varies	Details
Refining existing understanding.	Participants with existing expertise in the specific topics, deliberated on in PSP processes can refine their understanding of those topics.
Learning new content about specific topics.	Participants most commonly learn new content about specific topics, in which they are not experts. The specific topics vary, according to their existing fields of expertise, and according to the design of specific PSP processes.
Identifying opportunities for tackling SEPs.	The opportunities participants identify vary according to their fields of expertise, social and economic backgrounds, and the design of specific PSP processes.
New ways of thinking about the future.	Participants for whom PSP is a new experience can learn about its potential value as a way of thinking about the future. Participants who are already familiar with PSP can learn about new ways of thinking about the future in different specific PSP processes.
Learning varies according to relevance of knowledge.	Variations in learning are influenced by the extent to which participants encounter knowledge they consider relevant to their interests. As Knowles et al. (2005) explain, people need to understand how knowledge can help them in their everyday lives for them to learn from it.
Learning varies according to legitimacy of knowledge.	Variations in learning are influenced by the extent participants consider knowledge to be legitimate. This is shaped by power imbalances that can create conformity in participants' thinking, or 'groupthink' (Brookfield, 2005)
Facilitators shape learning	Facilitators also have significant power to shape learning, through defining what is considered acceptable for discussion. However, facilitators are influenced by the objectives of wider projects, in which specific PSP processes are embedded.

Source: author construct

7.3 Original Contributions

As shown in the above three tables, my research has contributed to understanding on PSP in several ways. Firstly, I have contributed to previously limited understanding regarding the expected and reported benefits of PSP. The review by Oteros-Rozas et al. (2015) provides a detailed analysis of the expected and reported benefits from 23 cases of PSP. However, the authors of the review are all practitioners who were directly involved in the PSP processes they analyse. My research represents the first analysis, that I am aware of, which investigates the expected and reported benefits of PSP processes, from an independent perspective. My findings show that learning is the most significant benefit associated with PSP, which corresponds with the findings of Oteros-Rozas et al. However, my findings go further than Oteros-Rozas et al., in that I reveal a lack of explicit, theoretically-informed understanding of how learning occurs. Furthermore, I show that little attention has thus far been paid to how and why learning varies for different participants.

My findings, presented in Chapter 5, contribute a theoretically-grounded understanding of how learning occurs, that was previously absent in PSP practice in the field of social-ecological resilience. My findings correspond with the assertion of Johnson et al. (2012), who reason that PSP encourages participants to engage with different knowledges. However, my findings go further than those of Johnson et al., in that I explain the specific attributes of PSP that can prompt discussion between participants, as well as stimulate creative, focused thinking. I also delve into the cognitive processes through which learning occurs, by linking my findings to a specific learning theory, the 'zone of proximal development' (Vygotsky, 1978). In doing so, I show that the creative, focused thinking, stimulated by PSP helps participants to push beyond their usual range of thinking, thus entering their zones of proximal development. As Terry, a highly experienced practitioner, indicated, such a theoretical understanding of learning can help practitioners to understand when PSP is being done well, and when it is not.

My research also contributes to previously limited understanding of the roles played by facilitators in PSP. Some PSP scholars, like Bennett et al. (2016b), and Rounsevell and Metzger (2010), have alluded to the importance of skilled facilitation in PSP processes, but do not analyse the roles that facilitators can play, in a detailed manner. My findings show that facilitation is especially important for providing assistance to participants that enables them to engage in discussions, and thus, to push beyond their usual range of thinking. Facilitators also have an

important role to play in selecting participants who have relevant expertise. The importance of skilled facilitation is not a new finding, in its own right. Indeed, skilled facilitation is highlighted as a key condition for successful stakeholder participation in processes of dialogue for tackling SEPs (Henrichs et al., 2010; Reed, 2008; Wittmayer and Schäpke, 2014). However, my findings focus specifically on the role of facilitation in encouraging learning through PSP, whereas others have focused on a range of outcomes from participatory processes in general. Furthermore, by linking the role of facilitation to learning theory on 'scaffolding' and the zone of proximal development, I emphasise that facilitators are vital for helping participants engage in creative, focused thinking, and thus, to push beyond their usual range of thinking.

As described in Chapter 6, my findings show that facilitators also play an important role in shaping who learns what in PSP. They do so through defining what is considered acceptable for discussion in PSP processes, something which influences the extent that participants encounter knowledge they consider relevant and legitimate. Facilitators are themselves influenced by the objectives of wider projects in which specific PSP processes are embedded. This links to wider debates about power and participation. Indeed, the influential critic of participation, Chambers (1997) contends that facilitators ultimately control the outcomes of participatory processes. My findings appear to support and add to these critical assertions about participation, by showing that the facilitators, and predefined objectives of PSP processes, have a significant influence on who learns what.

The previous lack of theoretically-informed understanding, regarding how learning occurs, together with the lack of information in the literature, regarding who learned what, and the paucity of reflection on the role of facilitation in enabling and shaping learning, indicates there is a need for greater critical reflection on the use of PSP. As I showed in Chapter 4, practitioners struggle to articulate the strengths and weaknesses of PSP, especially as compared to other methods. My findings, from the practitioner interviews and case studies show there is limited provision for systematic evaluation of PSP processes, and that limited consideration is given to alternatives. Indeed, as one practitioner, Terry, opined, the use of PSP for tackling SEPs is a fashion, that occurs with insufficient critical reflection.

7.4 Implications for practice and future research

The apparent lack of critical reflection, described above, could be indicative of a wider reluctance by practitioners and funders to consider the potential drawbacks, as well as the failures, of participatory initiatives pertaining to address SEPs. Indeed, as Mosse (2005) contends, the perceived efficacy of any initiative is often defined by the prevailing view held by those who commission it, rather than by the intended beneficiaries. The predominant view of PSP appears to be that it is beneficial for enabling learning that can help develop a more holistic understanding of SEPs.

My research supports this assumption, but given the significant role played by the design and facilitation of specific PSP processes in defining what is learned by participants, my research also highlights the importance of considering for whom learning in PSP is intended, as well as the level at which learning in PSP is intended to focus. Are PSP processes primarily intended to extract participants' contextual, local, and disciplinary knowledge to add to a more holistic understanding of SEPs, according to Lewin's (1997) theory of change? Or are they also intended to encourage micro-level learning about specific responses to SEPs, as in the reperception aspect of Ramirez and Wilkinson's (2016) framework? My findings in Chapter 6 indicate that participants' learning is predominantly directed towards end-points that are predetermined by the objectives of wider research projects.

This point emphasises the importance of considering the role of power in tackling SEPs. As Patterson et al. (2017) point out, changes that can lead to more sustainable conditions in SES are highly contested, and will inevitably result in winners and losers. Indeed, they contend that questions regarding what a sustainable future should look like, as well as who gets to decide this, are deeply political. However, they contend that these questions are not often addressed directly in sustainability research. Indeed, as Olsson et al. (2014) point out, power has typically been underplayed by scholars who are thinking about tackling SEPs. They emphasise that redistribution and sharing of power are key conditions for encouraging sustainability. However, they also contend that initiatives purporting to encourage sustainability are often top-down, pre-packaged, standard solutions, rather than innovations adapted to local contexts. Similarly, Lang et al. (2012) observe that participants. My findings support this, in that learning in PSP processes appears to be heavily influenced by those who initiate them.

I therefore argue there is a need for more robust governance of PSP practice in the field of social-ecological resilience to ensure best practice and encourage critical reflection. Improved

governance should include greater provision for systematic evaluation of the societal impacts PSP processes may have. Indeed, as Lewin (1997) argues in his theory of change, whilst general systems need to be understood holistically to explore possible conditions and outcomes, knowledge of specific situations is needed to take action, and evaluation is needed to evaluate its success, once implemented. As explained in Chapter 4, systematic evaluation of PSP has previously been hindered by a lack of mandate for evaluation in the field of practice in social-ecological resilience. The practitioner, Gavin, attributed this to a focus on scenarios as end-points, rather than means to achieving other outcomes, as well as a lack of accountability for impact among academics practicing PSP. This highlights the relative immaturity of PSP practice in the field of social-ecological resilience.

However, as this field matures there is increasing commitment to critical reflection on the benefits and limitations of PSP, as shown by Oteros-Rozas et al. (2015), Waylen et al. (2015), and indeed this thesis. Equally, there have been some recent attempts to link PSP to theory. For example, Duckett et al. (2017) study the role of power dynamics in PSP by applying Habermas' theory of communicative action. As the practitioner, Terry, alluded to, developing closer links to theory could improve governance of PSP by helping practitioners to understand when it is being done well and when it is not. Practitioners of PSP in the field of social-ecological resilience could also usefully draw to a greater extent on the wealth of theory and experience that exists in more established fields of PSP practice, such as corporate strategic planning and futures studies. For example, governance could be improved by encouraging sharing of excellence and best practice through practitioners' seeking membership of the Association of Professional Futurists (APF, 2018).

My emphasis on improved governance of PSP in this field echoes wider calls for improved evaluation regarding the effectiveness of participatory processes purporting to encourage dialogue on tackling SEPs. For example, Lang et al. (2012) emphasise that evaluating the societal impact of such participatory processes is a key condition of participatory, transdisciplinary research for tackling SEPs. Similarly, Wiek et al. (2014) point out that the emphasis of participatory processes on enabling change for sustainability calls for evaluation of their effectiveness. Wiek et al. thus propose a framework for evaluating the societal effects of participatory processes, which includes assessing participants' learning, as well as how it may enhance their capacity to tackle SEPs. My research can be used to support such an evaluation by helping to identify the causal links between specific aspects of PSP (as well as other participatory processes) and learning.

In any evaluation of PSP processes, it is important to remember that, as Henrichs et al. (2010) point out, different processes have different purposes, which influence the mode of participation that is used, and indeed what learning is intended to involve. For instance, in a PSP process for research and scientific exploration a successful learning outcome would involve experts gaining insights into the topic of study. In contrast, for a PSP process with the aim of educating people, successful learning would involve a wide range of stakeholders challenging their assumptions about the future and learning from the perspectives of others. In a process aimed at supporting decision-making, learning could be considered successful if key decision-makers were enabled to identify new, implementable strategies. Henrichs et al. therefore argue it is vital for the purpose and goals of a PSP process to be clearly defined from the start. They add that, as a minimum standard, PSP practitioners should ensure they conduct a thorough analysis of the wider context in which PSP is to be used, as well as engage relevant stakeholders throughout the process.

Wiek et al. (2014) also highlight the importance of evaluating the efficacy of usable products (like technologies and publications), network effects (like increased trust and new contacts), and structural changes and actions (like implemented policies and actions), which were beyond the scope of this research. Indeed, future research could usefully explore the effectiveness of PSP for achieving these outcomes. As Lang et al. (2012) point out, such an evaluation is complicated, because the effects of participatory processes are often delayed and diffuse, which creates difficulties in attributing them to specific interventions. In Chapter 4, I showed there is limited evidence for responses to SEPs, identified in PSP, being implemented in practice. However, as indicated by one practitioner, Greg, learning may be linked to diffuse impacts, such as an increased capacity to make better informed decisions. Future research that evaluates PSP, and other participatory processes, could therefore use learning as a starting point, and track how it may inform subsequent decision-making, as well as action.

My research highlights the importance of facilitation in supporting as well as shaping learning as an outcome of PSP. It is therefore vital that future evaluation of PSP processes includes a critical reflection on the roles played by facilitators. This links to ongoing debates in sustainability research regarding the role of researchers in participatory processes that aim to tackle SEPs. Scholars, including Miller (2013), and Pohl et al. (2010), enthusiastically encourage researchers to take an active role in facilitating such participatory processes. This is reflected in my research, in that all but two of the practitioners I interviewed, were academic researchers.

However, Wittmayer and Schäpke (2014) take a more cautious approach, questioning the suitability, as well as the level of appropriate training, researchers have for facilitation. They subsequently encourage researcher-facilitators to give themselves space for self-reflection, and adequate training, to ensure they have the necessary skills for facilitation. They also call on universities and funding bodies, to give institutional space for researchers to acquire appropriate training. Indeed, as Ramirez and Wilkinson (2016) contend, there is presently no requirement for practitioners to attain any form of qualification before using PSP. As this field of PSP practice matures, practitioners could be encouraged to uphold best practice by receiving a professional accreditation before using PSP. I argue that facilitations of PSP processes should engage with facilitation guides, like the 'sourcebook' of facilitation best practice by Chambers (2002), as well as the wealth of online resources for facilitation, and facilitation training programmes.

My findings can be used to guide facilitators, and ensure they are adequately prepared to support learning in PSP. My analysis highlights the importance of creative, focused thinking that enables participants to push beyond their usual range of thinking. This emphasises the importance of enabling participants to 'enter' their 'zones of proximal development,' as explained by Vygotsky (1978). Facilitators can achieve this through: designing PSP processes around specific activities, like 'futures wheels,' that stimulate discussion and creative thinking; explaining and guiding participants through specific activities; prompting discussion; and managing interactions between carefully selected participants. They could also usefully engage more systematically with the extensive learning and educational psychology literature that stems from the ZPD (Chaiklin, 2003).

In Chapter 4, I revealed that there is limited information in literature on PSP, regarding its relative usefulness, as compared to other methods for dialogue on tackling SEPs. This creates difficulties for assessing whether PSP is more, or less beneficial than other such methods. As shown in my practitioner interviews, this can make it difficult for practitioners to take an informed decision, regarding whether PSP is the most appropriate method to use, in specific contexts. A useful line of inquiry, for future research, could therefore be to compare PSP against other methods. This could involve comparing it to other futures thinking tools, like Delphi technique (Mukherjee et al., 2015), or Horizon Scanning (Sutherland and Woodroof, 2009). It could also include comparing PSP to other participatory methods, that centre around the development of narratives, such as participatory theatre (Brown et al., 2017), or participatory video (Mistry et al., 2016). A 'pure' comparison of the extent and content of learning using different methods would be impossible, unless the different methods were utilised in identical contexts, which is unlikely.

Comparisons of PSP against other methods should therefore focus on the extent to which it stimulates the creative, focused thinking that can encourage learning.

7.5 Conclusion

Participatory scenario planning is a useful method for stimulating learning about socialecological problems. It can enable learning by prompting discussion between different stakeholders, which can encourage creative, focused thinking about the future. In doing so, it can push participants to push beyond their usual range of thinking, or to 'enter' their 'zones of proximal development.' Learning can be supported by facilitators, through assisting participants to engage in such creative, focused thinking. The type and content of learning in PSP varies, according to the extent participants encounter knowledge they consider relevant to their interests, and to be produced through a fair and unbiased process. Facilitators have significant power to shape what is learned in specific PSP processes, by defining what is considered acceptable to discuss. Facilitators are themselves influenced by the objectives of wider projects, in which specific PSP processes are embedded.

There is therefore a need for stringent governance of PSP processes for tackling SEPs. This should include greater critical reflection regarding who learning in PSP is primarily intended to be for. This could include further research into the societal impacts of PSP, especially regarding if and how learning helps participants to take more informed decisions, as well as actions to tackle SEPs. There is also a need for critical reflection on the role of facilitators, and for adequate training of those intending to facilitate PSP processes. Greater understanding is also required, regarding the relative strengths and weaknesses of PSP as compared to other futures-oriented, and narrative-based methods, for tackling SEPs.

Bibliography

Alcamo, J. & Henrichs, T. (2009). Towards Guidelines for Environmental Scenario Analysis. Environmental Futures: The Practice of Environmental Scenario Analysis, **2**, 13-35.

Amer, M., Daim, T. U. & Jetter, A. (2013). A review of scenario planning. Futures, 46, 23-40.

APF. (2018). Association of Professional Futurists Homepage. URL: <u>https://apf.org/</u> [23.03.2018].

- Avelino, F. & Rotmans, J. (2011). A dynamic conceptualization of power for sustainability research. *Journal of Cleaner Production*, **19**, 796-804.
- Badjeck, M. C. & Diop, N. (2011). The future is now: how scenarios can help Senegalese and Mauritanian fisheries adapt to climate change. *Nature and Faune*, **25**, 68-74.
- Balint, P. J., Stewart, R. E. & Desai, A. (2011). *Wicked environmental problems: managing uncertainty and conflict.* Washington DC: Island Press.
- Bandura, A. (1977). Social learning theory. New Jersey: Prentice-Hall.
- Barnes, D. (2008). Exploratory Talk for Learning. *In:* Mercer, N. & Hodgkinson, S. (eds.) *Exploring Talk in School.* London: Sage.
- Baxter, J. & Eyles, J. (1997). Evaluating qualitative research in social geography: establishing 'rigour' in interview analysis. *Transactions of the Institute of British Geographers*, **22**, 505-525.
- Bennett, E. M., Solan, M., Biggs, R., McPhearson, T., Norström, A. V., Olsson, P., Pereira, L.,
 Peterson, G. D., Raudsepp-Hearne, C., Biermann, F., Carpenter, S. R., Ellis, E. C., Hichert,
 T., Galaz, V., Lahsen, M., Milkoreit, M., Martin López, B., Nicholas, K. A., Preiser, R., Vince,
 G., Vervoort, J. M. & Xu, J. (2016a). Bright spots: seeds of a good Anthropocene. *Frontiers in Ecology and the Environment*, 14, 441-448.
- Bennett, N. J., Kadfak, A. & Dearden, P. (2016b). Community-based scenario planning: a process for vulnerability analysis and adaptation planning to social–ecological change in coastal communities. *Environment, Development and Sustainability*, **18**, 1771-1799.
- Biggs, R., Raudsepp-Hearne, C., Atkinson-Palombo, C., Bohensky, E., Boyd, E., Cundill, G., Fox, H., Ingram, S., Kok, K., Spehar, S., Tengo, M., Timmer, D. & Zurek, M. (2007). Linking futures across scales: a dialog on multiscale scenarios. *Ecology and Society*, **12**.
- Bohensky, E. B., Butler, J. R. A. & Mitchell, D. (2009). Scenarios as models for knowledge integration: ecotourism futures in Milne Bay, Papua New Guinea. *18th World Imacs Congress and Modsim09 International Congress on Modelling and Simulation: Interfacing Modelling and Simulation with Mathematical and Computational Sciences*, 2826-2832.
- Bohensky, E. L., Reyers, B. & Van Jaarsveld, A. S. (2006). Future ecosystem services in a Southern African river basin: a scenario planning approach to uncertainty. *Conservation Biology*, **20**, 1051-1061.
- Borjeson, L., Hojer, M., Dreborg, K.-H., Ekvall, T. & Finnveden, G. (2006). Scenario types and techniques: Towards a user's guide. *Futures*, **38**, 723-739.
- Boyd, E. & Folke, C. (2012). Conclusions: adapting institutions and resilience. *In:* Boyd, E. & Folke,
 C. (eds.) *Adapting institutions: governance, complexity, and social-ecological resilience.* Cambridge: Cambridge University Press.
- Bradfield, R., Wright, G., Burt, G., Cairns, G. & Van Der Heijden, K. (2005). The origins and evolution of scenario techniques in long range business planning. *Futures*, **37**, 795-812.
- Brand, F. S., Seidl, R., Le, Q. B., Brandle, J. M. & Scholz, R. W. (2013). Constructing Consistent Multiscale Scenarios by Transdisciplinary Processes: the Case of Mountain Regions Facing Global Change. *Ecology and Society*, 18.
- Brookfield, S. (2005). *The power of critical theory for adult learning and teaching*. Berkshire: Open University Press.

- Brown, I., Martin-Ortega, J., Waylen, K. & Blackstock, K. (2016). Participatory scenario planning for developing innovation in community adaptation responses: three contrasting examples from Latin America. *Regional Environmental Change*, **16**, 1685–1700.
- Brown, K., Adger, W. N., Tompkins, E., Bacon, P., Shim, D. & Young, K. (2001). Trade-off analysis for marine protected area management. *Ecological Economics*, **37**, 417-434.
- Brown, K., Eernstman, N., Huke, A. R. & Reding, N. (2017). The drama of resilience: learning, doing, and sharing for sustainability. *Ecology and Society*, **22**.
- Bryman, A. (2008). Social Research Methods. Third Edition ed. Oxford: Oxford University Press.
- Butler, J. R. A., Wise, R. M., Skewes, T. D., Bohensky, E. L., Peterson, N., Suadnya, W., Yanuartati, Y., Handayani, T., Habibi, P., Puspadi, K., Bou, N., Vaghelo, D. & Rochester, W. (2015).
 Integrating Top-Down and Bottom-Up Adaptation Planning to Build Adaptive Capacity: A Structured Learning Approach. *Coastal Management*, 43, 346-364.
- Carpenter, S. R., Bennett, E. M. & Peterson, G. D. (2006). Scenarios for ecosystem services: An overview. *Ecology and Society*, **11**.
- Carpenter, S. R., Pingali, P., Bennett, E. M. & Zurek, M. (2005). Ecosystems and Human Well-being: Scenarios, Volume 2. Millennium Ecosystem Assessment.
- Cash, D. W., Clark, W. C., Alcock, F., Dickson, N. M., Eckley, N., Guston, D. H., Jager, J. & Mitchell, R. B. (2003). Knowledge systems for sustainable development. *Proceedings of the National Academy of Sciences of the United States of America*, **100**, 8086-8091.
- Chaiklin, S. (2003). The zone of proximal development in Vygotsky's analysis of learning and instruction. *In:* Kozulin, A. (ed.) *Vygotsky's Educational Theory in Cultural Context.* New York: Cambridge University Press.
- Chambers, R. (1997). Whose reality counts?: putting the first last.
- Chambers, R. (2002). *Participatory workshops: a sourcebook of 21 sets of ideas and activities.* London; Sterling, VA Earthscan Publications.
- Chaudhury, M., Vervoort, J., Kristjanson, P., Ericksen, P. & Ainslie, A. (2012). Participatory scenarios as a tool to link science and policy on food security under climate change in East Africa. *Regional Environmental Change*, **13**, 389-398.
- Cilliers, P., Biggs, H. C., Blignaut, S., Choles, A. G., Hofmeyr, J.-H. S., Jewitt, G. P. W. & Roux, D. J. (2013). Complexity, Modeling, and Natural Resource Management. *Ecology and Society*, **18**.
- Clegg, S. R. (1989). Frameworks of power. London: Sage.
- Cornwall, A. (2004). Spaces for Transformation? Reflections on Issues of Power and Difference in Participation in Development. *In:* Hickey, S. & Mohan, G. (eds.) *Participation, from tyranny to transformation? : exploring new approaches to participation in development.* London: Zed Books.
- Cornwall, A. (2008). Unpacking 'Participation': models, meanings and practices. *Community Development Journal*, **43**, 269-283.
- Crutzen, P. J. & Stoermer, E. F. (2000). The 'Anthropocene'. *Global change newsletter*, **41**, 17-18.
- De Houwer, J., Barnes-Holmes, D. & Moors, A. (2013). What is learning? On the nature and merits of a functional definition of learning. *Psychonomic Bulletin & Review*, **20**, 631-642.
- Duckett, D. G., McKee, A. J., Sutherland, L.-A., Kyle, C., Boden, L. A., Auty, H., Bessell, P. R. & McKendrick, I. J. (2017). Scenario planning as communicative action: Lessons from participatory exercises conducted for the Scottish livestock industry. *Technological Forecasting and Social Change*, **114**, 138-151.

Engeström, Y. (1987). Learning by Expanding. Helsinki: Orienta-Konsultit Oy.

Fazey, I., Evely, A. C., Reed, M. S., Stringer, L. C., Kruijsen, J., White, P. C. L., Newsham, A., Jin, L. X., Cortazzi, M., Phillipson, J., Blackstock, K., Entwistle, N., Sheate, W., Armstrong, F., Blackmore, C., Fazey, J., Ingram, J., Gregson, J., Lowe, P., Morton, S. & Trevitt, C. (2013). Knowledge exchange: a review and research agenda for environmental management. *Environmental Conservation*, 40, 19-36.

- Fernández, M., Wegerif, R., Mercer, N. & Rojas-Drummond, S. (2001). Re-conceptualizing "scaffolding" and the zone of proximal development in the context of symmetrical collaborative learning. *The journal of classroom interaction*, **50**, 40-54.
- Fisher, B., Turner, R. K., Burgess, N. D., Swetnam, R. D., Green, J., Green, R. E., Kajembe, G.,
 Kulindwa, K., Lewis, S. L., Marchant, R., Marshall, A. R., Madoffe, S., Munishi, P. K. T.,
 Morse-Jones, S., Mwakalila, S., Paavola, J., Naidoo, R., Ricketts, T., Rouget, M., Willcock,
 S., White, S. & Balmford, A. (2011). Measuring, modeling and mapping ecosystem services in the Eastern Arc Mountains of Tanzania. *Progress in Physical Geography*, **35**, 595-611.
- Folke, C., Carpenter, S. R., Walker, B., Scheffer, M., Chapin, T. & Rockstrom, J. (2010). Resilience Thinking: Integrating Resilience, Adaptability and Transformability. *Ecology and Society*, 15.
- Foucault, M. (1975). Discipline and Punish: The birth of the prison. Harmondsworth: Penguin.
- Gidley, J. M., Fien, J., Smith, J. A., Thomsen, D. C. & Smith, T. F. (2009). Participatory futures methods: towards adaptability and resilience in climate-vulnerable communities. *Environmental Policy and Governance*, **19**, 427-440.
- Guston, D. H. (2001). Boundary Organizations in Environmental Policy and Science: An Introduction. *Science, Technology, & Human Values,* **26**, 399-408.
- Haddaway, N. R., Woodcock, P., Macura, B. & Collins, A. (2015). Making literature reviews more reliable through application of lessons from systematic reviews. *Conservation Biology*, **29**, 1596–1605.
- Hanspach, J., Hartel, T., Milcu, A. I., Mikulcak, F., Dorresteijn, I., Loos, J., von Wehrden, H., Kuemmerle, T., Abson, D., Kovacs-Hostyanszki, A., Baldi, A. & Fischer, J. (2014). A holistic approach to studying social-ecological systems and its application to southern Transylvania. *Ecology and Society*, **19**.
- Harland, T. (2003). Vygotsky's Zone of Proximal Development and Problem-based Learning: Linking a theoretical concept with practice through action research. *Teaching in Higher Education*, **8**, 263-272.
- Hegger, D., Lamers, M., Van Zeijl-Rozema, A. & Dieperink, C. (2012). Conceptualising joint knowledge production in regional climate change adaptation projects: success conditions and levers for action. *Environmental Science & Policy*, **18**, 52-65.
- Henly-Shepard, S., Gray, S. A. & Cox, L. J. (2015). The use of participatory modeling to promote social learning and facilitate community disaster planning. *Environmental Science & Policy*, **45**, 109-122.
- Henrichs, T., Zurek, M., Eickhout, B., Kok, K., Raudsepp-Hearne, C., Ribeiro, T., Van Vuuren, D. P. & Volkery, A. (2010). Scenario Development and Analysis for Forward-looking Ecosystem Assessment. *In:* Ash, N., Blanco, H., Garcia, K., Tomich, T., Vira, B., Zurek, M. & Brown, C. (eds.) *Ecosystems and Human Well-Being : A Manual for Assessment Practitioners.* Washington DC: Island Press.
- Hughes, N. (2013). Towards improving the relevance of scenarios for public policy questions: A proposed methodological framework for policy relevant low carbon scenarios. *Technological Forecasting and Social Change*, **80**, 687-698.
- Illeris, K. (2009). A comprehensive understanding of human learning. *In:* Illeris, K. (ed.) *Contemporary Theories of Learning: Learning Theorists... in their own words.* Oxon, New York: Routledge.
- Jarvis, P. (2005). Towards a philosophy of human learning: an existentialist perspective. *In:* Jarvis, P. & Parker, S. (eds.) *Human learning: an holistic approach.* London: Routledge.
- Jarvis, P., Holford, J. & Griffin, C. (2003). *The Theory and Practice of Learning*. 2nd ed. London: Konan Page.
- Jasanoff, S. (1996). Beyond Epistemology: Relativism and Engagement in the Politics of Science. *Social Studies of Science*, **26**, 393-418.

- Jessel, B. & Jacobs, J. (2005). Land use scenario development and stakeholder involvement as tools for watershed management within the Havel River Basin. *Limnologica*, **35**, 220-233.
- Johnson, K. A., Dana, G., Jordan, N. R., Draeger, K. J., Kapuscinski, A., Olabisi, L. K. S. & Reich, P. B. (2012). Using Participatory Scenarios to Stimulate Social Learning for Collaborative Sustainable Development. *Ecology and Society*, **17**.
- Kapoor, I. (2005). Participatory development, complicity and desire. *Third World Quarterly*, **26**, 1203-1220.
- Knowles, M. S., Holton, E. F. & Swanson, R. A. (2005). *The adult learner: the definitive classic in adult education and human resource development.* London: Elsevier.
- Kok, K., Biggs, R. & Zurek, M. (2007). Methods for developing multiscale participatory scenarios: Insights from southern Africa and Europe. *Ecology and Society*, **12**.
- Kok, M. T., Kok, K., Peterson, G. D., Hill, R., Agard, J. & Carpenter, S. R. (2016). Biodiversity and ecosystem services require IPBES to take novel approach to scenarios. *Sustainability Science*, **12**, 177–181.
- Kothari, U. (2001). Power, Knowledge and Social Control in Participatory Development. *In:* Cooke, B. & Kothari, U. (eds.) *Participation: the new tyranny?* London: Zed Books.
- Lang, D. J., Wiek, A., Bergmann, M., Stauffacher, M., Martens, P., Moll, P., Swilling, M. & Thomas, C. J. (2012). Transdisciplinary research in sustainability science: practice, principles, and challenges. *Sustainability science*, 7, 25-43.
- Lefebvre, H. (1991). The Production of Space. Oxford: Blackwell.
- Lewin, K. (1997). *Resolving Social Conflicts and Field Theory in Social Science*. Washington DC: American Psychological Association.
- Lewis, J. (2003). Design Issues. *In:* Ritchie, J. & Lewis, J. (eds.) *Qualitative research practice: A guide for social science students and researchers.* London, Thousand Oaks, New Delhi: Sage.
- Lewis, J., Ritchie, J., Ormston, R. & Morrell, G. (2016). Generalising from Qualitative Research. *Qualitative Research Practice: A guide for social science students and researchers.* 2nd ed. Los Angeles: Sage.
- Lincoln, Y. S. & Guba, E. G. (2016). The Constructivist Credo. Walnut Creek, US: Routledge.
- Lukes, S. (2005). *Power: A Radical View.* 2nd ed. London: Palgrave Macmillan in association with the British Sociological Association.
- Malinga, R., Gordon, L. J., Lindborg, R. & Jewitt, G. (2013). Using Participatory Scenario Planning to Identify Ecosystem Services in Changing Landscapes. *Ecology and Society*, **18**.
- McNaughton Nicholls, C., Mills, L. & Kotecha, M. (2016). Observation. *In:* Ritchie, J., Lewis, J., McNaughton Nicholls, C. & Ormston, R. (eds.) *Qualitative Research Practice: A Guide for Social Science Students and Researchers*. 2nd ed. London: Sage.
- Meadows, D. H. & Club of Rome. (1972). *The limits to growth : a report for the Club of Rome's project on the predicament of mankind.* London: Pan.
- Merriam, S. B. & Bierema, L. L. (2013). *Adult Learning: Linking Theory and Practice.* Somerset, US: Jossey-Bass.
- Miller, T. R. (2013). Constructing sustainability science: emerging perspectives and research trajectories. *Sustainability Science*, **8**, 279-293.
- Mistry, J., Bignante, E. & Berardi, A. (2016). Why are we doing it? Exploring participant motivations within a participatory video project. *Area*, **48**, 412-418.
- Mistry, J., Tschirhart, C., Verwer, C., Glastra, R., Davis, O., Jafferally, D., Haynes, L., Benjamin, R., Albert, G., Xavier, R., Bovolo, I. & Berardi, A. (2014). Our common future? Cross-scalar scenario analysis for social-ecological sustainability of the Guiana Shield, South America. *Environmental Science & Policy*, **44**, 126-148.
- Mora, O., Banos, V., Regolini, M. & Carnus, J. M. (2014). Using scenarios for forest adaptation to climate change: a foresight study of the Landes de Gascogne Forest 2050. *Annals of Forest Science*, **71**, 313-324.

- Mosse, D. (2005). *Cultivating Development: An Ethnography of Aid Policy and Practice* (*Anthropology, Culture and Society Series*). London, United Kingdom: Pluto Press.
- Mukherjee, N., Hugé, J., Sutherland, W. J., McNeill, J., Van Opstal, M., Dahdouh-Guebas, F. & Koedam, N. (2015). The Delphi technique in ecology and biological conservation: applications and guidelines. *Methods in Ecology and Evolution*, **6**, 1097-1109.
- Nieto-Romero, M., Milcu, A., Leventon, J., Mikulcak, F. & Fischer, J. (2016). The role of scenarios in fostering collective action for sustainable development: Lessons from central Romania. *Land Use Policy*, **50**, 156-168.
- Olsson, P., Galaz, V. & Boonstra, W. J. (2014). Sustainability transformations: a resilience perspective. *Ecology and Society*, **19**.
- Ormston, R., Spencer, L., Barnard, M. & Snape, D. (2016). The Foundations of Qualitative Research. *In:* Ritchie, J., Lewis, J., McNaughton Nicholls & Ormston, R. (eds.) *Qualitative Research Practice: A guide for social science students and researchers.* 2nd ed. London: Sage.
- Oteros-Rozas, E., Martín-López, B., Daw, T. M., Bohensky, E. L., Butler, J. R. A., Hill, R., Martin-Ortega, J., Quinlan, A., Ravera, F., Ruiz-Mallén, I., Thyresson, M., Mistry, J., Palomo, I., Peterson, G. D., Plieninger, T., Waylen, K. A., Beach, D. M., Bohnet, I. C., Hamann, M., Hanspach, J., Hubacek, K., Lavorel, S. & Vilardy, S. P. (2015). Participatory scenario planning in place-based social-ecological research: insights and experiences from 23 case studies. *Ecology and Society*, 20.
- Palacios-Agundez, I., Casado-Arzuaga, I., Madariaga, I. & Onaindia, M. (2013). The Relevance of Local Participatory Scenario Planning for Ecosystem Management Policies in the Basque Country, Northern Spain. *Ecology and Society*, **18**.
- Palomo, I., Martin-Lopez, B., Lopez-Santiago, C. & Montes, C. (2011). Participatory Scenario Planning for Protected Areas Management under the Ecosystem Services Framework: the Donana Social-Ecological System in Southwestern Spain. *Ecology and Society*, **16**.
- Parker, S. (2005). Human learning: the themes. *In:* Jarvis, P. & Parker, S. (eds.) *Human learning: an holistic approach.* London: Routledge.
- Parsons, T. (1967). On the concept of political power: sociological theory and modern society. London: Free Press.
- Patterson, J., Schulz, K., Vervoort, J., van der Hel, S., Widerberg, O., Adler, C., Hurlbert, M., Anderton, K., Sethi, M. & Barau, A. (2017). Exploring the governance and politics of transformations towards sustainability. *Environmental Innovation and Societal Transitions*, 24, 1-16.
- Pearson, L. J., Park, S., Harman, B. & Heyenga, S. (2010). Sustainable land use scenario framework: Framework and outcomes from pen-urban South-East Queensland, Australia. *Landscape* and Urban Planning, **96**, 88-97.
- Pert, P. L., Hill, R., Williams, K. J., Harding, E. K., O'Malley, T., Grace, R. A., Dale, A. P., Bohnet, I. & Butler, J. (2010). Scenarios for Community-based Approaches to Biodiversity Conservation: a case study from the Wet Tropics, Queensland, Australia. *Australian Geographer*, **41**, 285-306.
- Peterson, G. D., Cumming, G. S. & Carpenter, S. R. (2003). Scenario planning: a tool for conservation in an uncertain world. *Conservation Biology*, **17**, 358-366.
- Piaget, J. (1929). *The Child's Conception of the World*. London: Routledge and Kegan Paul.
- Plieninger, T., Bieling, C., Ohnesorge, B., Schaich, H., Schleyer, C. & Wolff, F. (2013). Exploring Futures of Ecosystem Services in Cultural Landscapes through Participatory Scenario Development in the Swabian Alb, Germany. *Ecology and Society*, 18.
- Pohl, C., Rist, S., Zimmermann, A., Fry, P., Gurung, G. S., Schneider, F., Speranza, C. I., Kiteme, B., Boillat, S. & Serrano, E. (2010). Researchers' roles in knowledge co-production: experience from sustainability research in Kenya, Switzerland, Bolivia and Nepal. *Science and Public Policy*, **37**, 267-281.

QSR, I. P. L. (2015). NVivo qualitative data analysis Software. Version 11 ed.

- Ramirez, R. & Wilkinson, A. (2016). *Strategic Reframing: The Oxford Scenario Planning Approach.* Oxford: Oxford University Press.
- Ravera, F., Hubacek, K., Reed, M. & Tarrason, D. (2011a). Learning from Experiences in Adaptive Action Research: a Critical Comparison of two Case Studies Applying Participatory Scenario Development and Modelling Approaches. *Environmental Policy and Governance*, 21, 433-453.
- Ravera, F., Tarrason, D. & Simelton, E. (2011b). Envisioning Adaptive Strategies to Change: Participatory Scenarios for Agropastoral Semiarid Systems in Nicaragua. *Ecology and Society*, 16.
- Reed, M. S. (2008). Stakeholder participation for environmental management: a literature review. *Biological conservation*, **141**, 2417-2431.
- Reed, M. S., Kenter, J., Bonn, A., Broad, K., Burt, T. P., Fazey, I. R., Fraser, E. D. G., Hubacek, K., Nainggolan, D., Quinn, C. H., Stringer, L. C. & Ravera, F. (2013). Participatory scenario development for environmental management: A methodological framework illustrated with experience from the UK uplands. *Journal of Environmental Management*, **128**, 345-362.
- Rickards, L., Wiseman, J., Edwards, T. & Biggs, C. (2014). The problem of fit: scenario planning and climate change adaptation in the public sector. *Environment and Planning C-Government and Policy*, **32**, 641-662.
- Rittel, H. W. J. & Webber, M. M. (1973). DILEMMAS IN A GENERAL THEORY OF PLANNING. *Policy Sciences*, **4**, 155-169.
- Rivard, B. & Reay, D. S. (2012). Future scenarios of Malawi's energy mix and implications for forest resources. *Carbon Management*, **3**, 369-381.
- Robinson, J., Burch, S., Talwar, S., O'Shea, M. & Walsh, M. (2011). Envisioning sustainability: Recent progress in the use of participatory backcasting approaches for sustainability research. *Technological Forecasting and Social Change*, **78**, 756-768.
- Robinson, J. B. (1982). Energy backcasting a proposed method of policy analysis. *Energy Policy*, **10**, 337-344.
- Rockstrom, J., Steffen, W., Noone, K., Persson, A., Chapin, F. S., Lambin, E. F., Lenton, T. M., Scheffer, M., Folke, C., Schellnhuber, H. J., Nykvist, B., de Wit, C. A., Hughes, T., van der Leeuw, S., Rodhe, H., Sorlin, S., Snyder, P. K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R. W., Fabry, V. J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P. & Foley, J. A. (2009). A safe operating space for humanity. *Nature*, 461, 472-475.
- Roessger, K. M. (2012). Re-conceptualizing adult education's monolithic behaviourist interpretation: toward a new understanding of radical behaviourism. *International Journal of Lifelong Education*, **31**, 569-589.
- Rotmans, J. & van Asselt, M. (1999). Integrated assessment modelling. *Climate change: An integrated perspective*, 239-275.
- Rounsevell, M. D. A. & Metzger, M. J. (2010). Developing qualitative scenario storylines for environmental change assessment. Wiley Interdisciplinary Reviews-Climate Change, 1, 606-619.
- Sandker, M., Suwarno, A. & Campbell, B. M. (2007). Will forests remain in the face of oil palm expansion? Simulating change in Malinau, Indonesia. *Ecology and Society*, **12**, 37.
- Schoemaker, P. J. H. (1991). When and how to use scenario planning a heuristic approach with illustration. *Journal of Forecasting*, **10**, 549-564.
- Schoemaker, P. J. H. (1993). Multiple scenario development its conceptual and behavioural foundation. *Strategic Management Journal*, **14**, 193-213.

- Schulz, C., Ioris, A. A. R., Martin-Ortega, J. & Glenk, K. (2015). Prospects for Payments for Ecosystem Services in the Brazilian Pantanal: A Scenario Analysis. *Journal of Environment* & Development, 24, 26-53.
- Schwartz, P. (1998). *The Art of the Long View: Planning for the Future in an Uncertain World.* 3 ed. Chichester: John Wiley and Sons.
- Shaw, A., Sheppard, S., Burch, S., Flanders, D., Wiek, A., Carmichael, J., Robinson, J. & Cohen, S. (2009). Making local futures tangible: Synthesizing, downscaling, and visualizing climate change scenarios for participatory capacity building. *Global Environmental Change-Human and Policy Dimensions*, **19**, 447-463.
- Sheppard, S. R. J., Shaw, A., Flanders, D., Burch, S., Wiek, A., Carmichael, J., Robinson, J. & Cohen,
 S. (2011). Future visioning of local climate change: A framework for community
 engagement and planning with scenarios and visualisation. *Futures*, 43, 400-412.
- Silverman, D. (2014). *Interpreting Qualitative Data*. 5th ed. London, Thousand Oaks, New Delhi, Singapore: Sage.
- Skinner, B. F. (1972). Beyond Freedom and Dignity. London: Jonathan Cape.
- Spencer, L., Ritchie, J., Ormston, R., O'Connor, W. & Barnard, M. (2016). Analysis: Principles and Processes. In: Ritchie, J., Lewis, J., McNaughton Nicholls, C. & Ormston, R. (eds.) Qualitative Research Practice: A Guide for Social Science Students and Researchers. 2nd ed. London: Sage.
- Stanley, M. C., Beggs, J. R., Bassett, I. E., Burns, B. R., Dirks, K. N., Jones, D. N., Linklater, W. L., Macinnis-Ng, C., Simcock, R., Souter-Brown, G., Trowsdale, S. A. & Gaston, K. J. (2015). Emerging threats in urban ecosystems: a horizon scanning exercise. *Frontiers in Ecology* and the Environment, **13**, 553-560.
- Star, J., Rowland, E. L., Black, M. E., Enquist, C. A. F., Garfin, G., Hoffman, C. H., Hartmann, H., Jacobs, K. L., Moss, R. H. & Waple, A. M. (2016). Supporting adaptation decisions through scenario planning: Enabling the effective use of multiple methods. *Climate Risk Management*, **13**, 88-94.
- Star, S. L. (2010). This is Not a Boundary Object: Reflections on the Origin of a Concept. *Science, Technology, & Human Values,* **35**, 601-617.
- Star, S. L. & Griesemer, J. R. (1989). Institutional Ecology, translations and boundary objects amateurs and professionals in Berkeleys-Museum-of-Vertebrate-Zoology, 1907-39. Social Studies of Science, 19, 387-420.
- Steffen, W., Broadgate, W., Deutsch, L., Gaffney, O. & Ludwig, C. (2015). The trajectory of the Anthropocene: The Great Acceleration. *The Anthropocene Review*, **2**, 81-98.
- Stirling, A. (2010). Keep it complex. Nature, 468, 1029-1031.
- Stringer, L., Dougill, A., Fraser, E., Hubacek, K., Prell, C. & Reed, M. (2006). Unpacking "participation" in the adaptive management of social–ecological systems: a critical review. *Ecology and Society*, **11**.
- Sutherland, W. J. & Woodroof, H. J. (2009). The need for environmental horizon scanning. *Trends in Ecology & Evolution*, **24**, 523-527.
- Swetnam, R. D., Fisher, B., Mbilinyi, B. P., Munishi, P. K. T., Willcock, S., Ricketts, T., Mwakalila, S., Balmford, A., Burgess, N. D., Marshall, A. R. & Lewis, S. L. (2011). Mapping socio-economic scenarios of land cover change: A GIS method to enable ecosystem service modelling. *Journal of Environmental Management*, 92, 563-574.
- Tett, L. (2016). Community-based learning and research: partnerships, power and learning. *Widening Participation and Lifelong Learning*, **18**, 6-16.
- Tschakert, P., Dietrich, K., Tamminga, K., Prins, E., Shaffer, J., Liwenga, E. & Asiedu, A. (2014). Learning and envisioning under climatic uncertainty: an African experience. *Environment and Planning A*, **46**, 1049-1068.

- Twyman, C., Fraser, E. D. G., Stringer, L. C., Quinn, C., Dougill, A. J., Ravera, F., Crane, T. A. & Sallu,
 S. M. (2011). Climate Science, Development Practice, and Policy Interactions in Dryland
 Agroecological Systems. *Ecology and Society*, 16.
- Van Berkel, D. B., Carvalho-Ribeiro, S., Verburg, P. H. & Lovett, A. (2011). Identifying assets and constraints for rural development with qualitative scenarios: A case study of Castro Laboreiro, Portugal. *Landscape and Urban Planning*, **102**, 127-141.
- Van der Heijden, K. (1996). Scenarios: The Art of Strategic Conversation. Chichester: John Wiley & Sons.
- van der Helm, R. (2009). The vision phenomenon: Towards a theoretical underpinning of visions of the future and the process of envisioning. *Futures*, **41**, 96-104.
- Van der Pol, J., Volman, M. & Beishuizen, J. (2010). Scaffolding in Teacher—Student Interaction: A Decade of Research. *Educational Psychology Review*, **22**, 271–296.
- van Notten, P. W. F., Rotmans, J., van Asselt, M. B. A. & Rothman, D. S. (2003). An updated scenario typology. *Futures*, **35**, 423-443.
- Varum, C. A. & Melo, C. (2010). Directions in scenario planning literature A review of the past decades. *Futures*, **42**, 355-369.
- Verma, R. (2001). *Gender, land and livelihoods in East Africa*. Ottawa: International Development Research Centre.
- Vermeulen, S. J., Challinor, A. J., Thornton, P. K., Campbell, B. M., Eriyagama, N., Vervoort, J. M., Kinyangi, J., Jarvis, A., Laderach, P., Ramirez-Villegas, J., Nicklin, K. J., Hawkins, E. & Smith, D. R. (2013). Addressing uncertainty in adaptation planning for agriculture. *Proceedings of the National Academy of Sciences of the United States of America*, **110**, 8357-8362.
- Vervoort, J., Palazzo, A., Mason-D'Croz, D., Ericksen, P., Thornton, P., Kristjanson, P., Forch, W., Herrero, M., Havlik, P. & Jost, C. (2013). The future of food security, environments and livelihoods in Eastern Africa: four socio-economic scenarios.
- Vervoort, J. M., Thornton, P. K., Kristjanson, P., Forch, W., Ericksen, P. J., Kok, K., Ingram, J. S. I., Herrero, M., Palazzo, A., Helfgott, A. E. S., Wilkinson, A., Havliik, P., Mason-D'Croz, D. & Jost, C. (2014). Challenges to scenario-guided adaptive action on food security under climate change. *Global Environmental Change-Human and Policy Dimensions*, 28, 383-394.
- Volkery, A. & Ribeiro, T. (2009). Scenario planning in public policy: Understanding use, impacts and the role of institutional context factors. *Technological Forecasting and Social Change*, 76, 1198-1207.
- Vygotsky, L. (1978). *Mind in society : the development of higher psychological processes.* Cambridge, Massachussetts, London: Harvard University Press.
- Wack, P. (1985). Scenarios Shooting the Rapids. Harvard Business Review, 63, 139-150.
- Wals, A. E. J. & Dillon, J. (2013). Conventional and emerging learning theories. In: Stevenson, R. B., Brody, M., Dillon, J. & Wals, A. E. J. (eds.) International Handbook of Research on Environmental Education. New York; London: Routledge.
- Watson, C. W. (2011). Ethical Issues in Research. *In:* Newing, H. (ed.) *Conducting Research in Conservation: A Social Science Perspective.* London: Routledge.
- Waylen, K. A., Martin-Ortega, J., Blackstock, K. L., Brown, I., Avendaño Uribe, B. E., Basurto Hernández, S., Bertoni, M. B., Bustos, M. L., Cruz Bayer, A. X., Escalante Semerena, R. I., Farah Quijano, M. A., Ferrelli, F., Fidalgo, G. L., Hernández López, I., Huamantinco Cisneros, M. A., London, S., Maya Vélez, D. L., Ocampo-Díaz, N., Ortiz-Guerrero, C. E., Pascale, J. C., Perillo, G. M. E., Piccolo, M. C., Pinzón Martínez, L. N., Rojas, M. L., Scordo, F., Vitale, V. & Zilio, M. I. (2015). Can scenario-planning support community-based natural resource management? Experiences from three countries in Latin America. *Ecology and Society*, 20.

- Wesche, S. D. & Armitage, D. R. (2014). Using qualitative scenarios to understand regional environmental change in the Canadian North. *Regional Environmental Change*, **14**, 1095-1108.
- White, D. D., Wutich, A., Larson, K. L., Gober, P., Lant, T. & Senneville, C. (2010). Credibility, salience, and legitimacy of boundary objects: water managers' assessment of a simulation model in an immersive decision theater. *Science and Public Policy*, **37**, 219-232.
- Wiek, A., Talwar, S., O'Shea, M. & Robinson, J. (2014). Toward a methodological scheme for capturing societal effects of participatory sustainability research. *Research Evaluation*, 23, 117-132.
- Wiles, R., Charles, V., Crow, G. & Heath, S. (2006). Researching researchers: lessons for research ethics. *Qualitative Research*, **6**, 283-299.
- Wilkinson, A. & Eidinow, E. (2008). Evolving practices in environmental scenarios: a new scenario typology. *Environmental Research Letters*, **3**.
- Wilkinson, A. & Kupers, R. (2014). *The Essence of Scenarios: Learning from the Shell experience*. Amsterdam: Amsterdam University Press.
- Winchester, H. P. M. & Rofe, M. W. (2010). Qualitative Research and Its Place in Human Geography. In: Hay, I. (ed.) Qualitative Research Methods in Human Geography. Oxford: Oxford University Press.
- Wittmayer, J. M. & Schäpke, N. (2014). Action, research and participation: roles of researchers in sustainability transitions. *Sustainability Science*, **9**, 483-496.
- Wollenberg, E., Edmunds, D. & Buck, L. (2000). Using scenarios to make decisions about the future: anticipatory learning for the adaptive co-management of community forests. *Landscape and Urban Planning*, **47**, 65-77.
- Wood, D., Bruner, J. S. & Ross, G. (1976). The role of tutoring in problem solving. *Journal of Child Psychology and Psychiatry*, **17**, 89-100.
- Yeo, A., Legard, R., Keegan, J., Ward, K., McNaughton Nicholls, C. & Lewis, J. (2016). In-depth interviews. *In:* Ritchie, J., Lewis, J., McNaughton Nicholls, C. & Ormston, R. (eds.) *Qualitative Research Practice: A Guide for Social Science Students and Researchers.* 2nd ed. London: Sage.

Appendix 1 – Interview guide for practitioner interviews

Discussion topics

- 1. Experiences of using scenario planning.
- 2. Disciplinary background
- 3. The approach taken to using scenario planning.
- 4. Motivations/rationales for using scenario planning. The intended benefits of using

scenario planning for the management of wicked problems.

- a. The theory underlying these intentions and rationales.
- b. The objectives for using scenario planning.
- The reported outcomes of using scenario planning for the management of wicked problems.
 - a. The learning outcomes of scenario planning processes.
 - b. The relationship (if any) between the learning outcomes and the management of wicked problems in practice?
- 6. The evidence they used to justify the reported outcomes of SPPs.
- 7. The challenges they experienced in using scenario planning.
- Discussion on the results of my case review, including the specific projects they were involved with, where appropriate.

Questions to discuss

Informal, factual, ice-breakers

- 1. Approximately how many scenario planning processes have you been involved in?
 - a. What role(s) did you take in each of them?
- Can you tell me any interesting anecdotes? For example, are there any particular scenario narratives, or scenario planning processes that have really stuck in your mind?
 ©University of Reading 2018 Tuesday, 22 May 2018 Page 220

- 3. Tell me a bit about your disciplinary background. How did you end up practicing scenario planning?
- 4. Did you receive any formal or informal training in how to facilitate scenario planning processes?
 - a. Could you describe what this entailed?
 - b. How did this inform the way you facilitated scenario planning?

Lead into more meaty questions on the core research

- 5. How do think your disciplinary background has influenced your role in facilitating scenario planning?
- 6. Can you describe your rationale for using scenario planning?
 - a. What were your expectations regarding the benefits scenario planning could achieve?
 - b. Why was scenario planning selected over other methods?
 - c. How did you come to form these expectations? Where did you get the idea that scenario planning might be beneficial?
- 7. Thinking specifically about the processes you have been involved in, what would you say were the benefits achieved by using scenario planning?
 - To what extent did scenario planning processes influence participants' understandings of the wicked problem(s) being addressed? If so, how?
 - b. Was there any variation between different participants in terms of the outcomes scenario planning had for them?
 - i. Could you describe this variation?
 - ii. Why do you think this was?
 - c. How would you describe the roles played by different participants in the scenario planning process?

- i. Could you describe the relationships that developed between different participants over the course of the process?
- ii. What do you think influenced the development of these relationships?
- d. What tangible impacts have scenario planning processes achieved in practice?
 - i. What were the mechanisms by which these impacts resulted from the scenario planning process?
 - ii. What was the relationship between the tangible impacts and the internal dynamics of the scenario planning process itself?
- 8. Can you justify these claims?
 - a. What concrete evidence do you have for these outcomes?
 - b. What methods and criteria have you used to assess the outcomes of scenario planning processes?
- 9. Could you tell me about any challenges you have faced in the use of scenario planning?
 - a. Why do you think these challenges came about?
 - b. How do you think these challenges could have been avoided, and could be avoided in the future?

Appendix 2 - Pre-workshop questionnaire in the case studies

Hello, my name is Sam Poskitt. I'm a PhD student from the University of Reading, UK. I'm studying learning in workshops that use scenario planning techniques to help explore possible future social and environmental conditions. I have been kindly invited to explore learning in the creative visioning workshop: 'Envisioning Good Anthropocenes in southern Africa,' which is being held by the Centre for Complex Systems in Transition in November 2016. I will be observing the workshop as it happens to get a sense of how the workshop is structured, how participants engage with it, and how it might encourage learning. I am very interested in finding out about what you learn in the workshop, and would be very grateful if we could arrange to meet in the weeks afterwards, to chat about what you learned from it.

This short survey is designed to help me, the coordinators of the workshop, and you, yourself to come to the workshop with a clear understanding of your expectations and aspirations for what you want to get out of it. The survey should only take around 15 minutes to complete and consists of 5 open-ended questions about you, your reasons for attending the workshop and what you expect the outcomes will be.

The information you choose to provide in the survey is confidential and will be accessed only by me, my research supervisors and the organisers of the workshop.

Thank you very much for your time and effort in taking part in this survey!

<u>About you</u>

Name:

Organisation(s):

May I contact you after the workshop to chat about what you learned?: Yes/No

If 'Yes' please provide details through which I may get in touch:

Telephone number:

Email address:

1) How would you describe what you do for a living?

2) What are the reasons you are interested in attending this workshop?

3) What do you expect the benefits of the workshop to be?

4) What topics do you think are important to consider in imagining the future of southern Africa?

5) What are the reasons you think these topics are important to consider when imagining the future of southern Africa?

Appendix 3 - Interview guide used in the case studies

- Icebreaker questions to ask me? Did my presence as an observer affect their experience of the workshop? If so, how?
- 2. Participant's expectations of the workshop.
 - a. A little bit about participant's background and reasons for attending the workshop.
 - b. What they expected the benefits of the workshop to be.
- 3. Participants' experiences of the workshop
 - a. How well participants understood the objectives of the workshop.
 - b. How easily participants were able to carry out the tasks set for them by the facilitators
 - c. What participants found interesting, challenging, easy, difficult about the process?
 - d. What participants thought about the location and layout of the workshop space? – How comfortable did they feel, how did the space affect their participation?
 - e. What participants felt they, and others contributed to discussions in the workshop and to what conditions, events and trajectories were eventually included in the storylines.
 - f. Participants' interactions with other participants. Who they spoke with most, the extent to which they felt included in group discussions, who they thought was most vocal and who was more of an active listener, what they thought about the characteristics of their group (argumentative, cooperative, friendly, relaxed, hostile, imaginative, pragmatic, analytical).
 - g. Participants interactions with facilitators how they helped participants carry out the tasks to engage with the process, how well the time was managed, how well they managed the group discussions and the interactions between different participants.
 - h. Observations as prompts "I noticed you seemed to be having an interesting discussion with *x*, could you tell me some more about that?"
- 4. What participant's think were the benefits of imagining future conditions of humanenvironmental systems.
 - a. What do they think were the benefits of imagining alternative futures of socialecological conditions?

- b. Has the way they imagine alternative futures of social-ecological systems changed? If so, how? – What topics, problems, opportunities, relationships has the workshop flagged up for them?
- c. What aspects of the workshop encouraged learning to occur? What sorts of processes do they feel help them to learn?
- d. What participants felt were the most important outcomes for them what they learnt, how they think the workshop will affect them in their everyday activities, any opportunities for new actions, roles and relationships to help encourage more sustainable and socially equitable future conditions in socialecological systems.
- e. Anything they thought could have been better about the process. What else would they have liked to learn about? What else do they think it would have been important for others to learn about? What would have helped further encourage learning?
- f. Would they take part in a participatory scenario planning exercise again in future? What are their reasons for this?